UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549 FORM 10-Q

(Ma	ırk	On	e)
-----	-----	----	----

(Ma	rk One)			
7	QUARTERLY REPORT PURSUANT TO SECTION	13 OR 15(d) OF THE SECUR	RITIES EXCHA	NGE ACT OF 1934
	For the	e quarterly period ended Ma	ay 1, 2021	
		or		
	TRANSITION REPORT PURSUANT TO SECTION	13 OR 15(d) OF THE SECUR	RITIES EXCHA	NGE ACT OF 1934
	For the transiti	on period from	to	
	Con	nmission File Number: 001	-15059	
	NOI	RDSTROM,	INC	
		me of registrant as specified		
	Washington		ŕ	91-0515058
	(State or other jurisdiction of			(I.R.S. Employer
	incorporation or organization)			Identification No.)
		th Avenue, Seattle, Washin dress of principal executive o	-	
		206-628-2111		
	(Registrant	t's telephone number, includi	ng area code)	
Sec	urities registered pursuant to Section 12(b) of the A	ct:		
	Title of each class	Trading Symbol	Name	of each exchange on which registered
	Common stock, without par value	JWN		New York Stock Exchange
Act sub	cate by check mark whether the registrant (1) has fi of 1934 during the preceding 12 months (or for sucject to such filing requirements for the past 90 days \square No \square	h shorter period that the regi		
Rule requ	cate by check mark whether the registrant has subrevented 405 of Regulation S-T (\S 232.405 of this chapter) of the submit such files). \square No \square			
com	cate by check mark whether the registrant is a large npany, or an emerging growth company. See the de "emerging growth company" in Rule 12b-2 of the E	finitions of "large accelerated		
√	Large Accelerated Filer	☐ Accelera	ted filer	
	Non-accelerated filer		reporting com _l g growth com _l	
	n emerging growth company, indicate by check mark any new or revised financial accounting standards			
	cate by check mark whether the registrant is a shell \square No \boxdot	company (as defined in Rul	e 12b-2 of the	Exchange Act).
Cor	nmon stock outstanding as of May 28, 2021: 158,89	98,676 shares		
		1 of 20		

TABLE OF CONTENTS

EODWADD I O	DOMING STATEMENTS	Page
FORWARD-LC	DOKING STATEMENTS	<u>3</u>
DEFINITIONS		<u>5</u>
PART I - FINA	NCIAL INFORMATION	
Item 1.	Financial Statements (Unaudited).	<u>6</u>
	Condensed Consolidated Statements of Earnings	<u>6</u>
	Condensed Consolidated Statements of Comprehensive Earnings	<u>6</u>
	Condensed Consolidated Balance Sheets	<u>7</u>
	Condensed Consolidated Statements of Shareholders' Equity	6 6 7 8 9
	Condensed Consolidated Statements of Cash Flows	<u>9</u>
	Notes to Consolidated Financial Statements	
	Note 1: Basis of Presentation	<u>10</u>
	Note 2: Revenue	<u>11</u>
	Note 3: Debt and Credit Facilities	<u>12</u>
	Note 4: Fair Value Measurements	<u>13</u>
	Note 5: Commitments and Contingencies	<u>13</u>
	Note 6: Shareholders' Equity Note 7: Stock-based Compensation	<u>14</u> 1 <i>1</i>
	Note 8: Earnings Per Share	14 15
	Note 9: Segment Reporting	12 13 13 14 14 15 15
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations.	<u>16</u>
Item 3.	Quantitative and Qualitative Disclosures About Market Risk.	<u>25</u>
Item 4.	Controls and Procedures.	<u>25</u>
PART II – OTH	IER INFORMATION	_
Item 1.	Legal Proceedings.	<u>26</u>
Item 1A.	Risk Factors.	<u>26</u>
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds.	<u>26</u>
Item 6.	Exhibits.	<u>26</u>
Exhibit Index		<u>27</u>
<u>Signatures</u>		<u>28</u>

FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, which are subject to the "safe harbor" created by those sections. Forward-looking statements are based on our management's beliefs and assumptions and on information currently available to our management. In some cases, you can identify forward-looking statements by terms such as "may," "will," "should," "could," "goal," "would," "expect," "plan," "anticipate," "believe," "estimate," "project," "predict," "potential," "pursue," "going forward," and similar expressions intended to identify forward-looking statements. These statements involve known and unknown risks, uncertainties and other factors, which may cause our actual results, performance, time frames or achievements to be materially different from any future results, performance, time frames or achievements expressed or implied by the forward-looking statements. These risks, uncertainties and other factors include, but are not limited to, our anticipated financial outlook for the fiscal year ending January 29, 2022, trends in our operations and the following:

Strategic and Operational

- the novel coronavirus ("COVID-19") global pandemic and civil unrest, each of which may make it necessary to close our physical stores
 and facilities in affected areas and may have a negative impact on our business and results, any of which may exacerbate the risks
 helow.
- successful execution of our customer strategy to provide customers superior service, products and experiences, online, through our fulfillment capabilities and in stores,
- timely and effective implementation and execution of our evolving business model, including:
 - scaling our market strategy by providing a differentiated and seamless experience, which consists of the integration of our digital and physical assets, development of new supply chain capabilities and timely delivery of products,
 - expanding the reach of Nordstrom Rack, including expanding our price range and selection and leveraging our digital and physical assets.
 - increasing our digital velocity by enhancing our platforms and processes to deliver core capabilities to drive customer, employee and partner experiences and service,
- · our ability to effectively manage our merchandise strategy, including our ability to offer compelling assortments,
- our ability to effectively utilize internal and third-party data in strategic planning and decision making,
- our ability to effectively allocate and scale our marketing strategies and resources between The Nordy Club, advertising and promotional campaigns,
- our ability to respond to the evolving retail environment, including new fashion trends, environmental considerations and our customers' changing expectations of service and experience in stores and online, and our development of new market strategies and customer offerings,
- our ability to prevent or mitigate disruptions in the global supply chain and rising freight expenses and control costs through effective inventory management, fulfillment and supply chain processes and systems,
- our ability to realize the expected benefits, anticipate and respond to potential risks and appropriately manage costs associated with our credit card revenue sharing program,
- our ability to acquire, develop and retain qualified talent and offer competitive compensation and benefits, especially in areas with increased market compensation.
- potential goodwill impairment charges, future impairment charges, fluctuations in the fair values of reporting units or of assets in the event projected financial results are not achieved within expected time frames or our strategic direction changes,

Data, Cybersecurity and Information Technology

- successful execution of our information technology strategy, including engagement with third-party service providers,
- the impact of any systems or network failures, cybersecurity and/or security breaches, including any security breach of our systems or those of a third-party provider that results in the theft, transfer or unauthorized disclosure of customer, employee or Company information, and our compliance with information security and privacy laws and regulations in the event of such an incident,

Reputation and Relationships

- our ability to maintain our reputation and relationships with our customers, vendors and third-party partners and landlords,
- our ability to maintain relationships with and motivate our employees and to effectively attract, develop and retain our top talent and future leaders.
- our ability to market our brand on a variety of publisher or platform channels, as well as our access to mobile operating systems and website identifiers for personalized delivery of targeted advertising,

Investment and Capital

- efficient and proper allocation of our capital resources,
- our ability to properly balance our investments in technology, Supply Chain Network facilities and existing and new store locations, including the expansion of our market strategy,
- our ability to maintain or expand our presence, including timely completion of construction associated with Supply Chain Network facilities and new, relocated and remodeled stores, as well as any potential store closures, all of which may be impacted by third parties, consumer demand and other natural or man-made disruptions, and government responses to any such disruptions,

- market fluctuations, increases in operating costs, exit costs and overall liabilities and losses associated with owning and leasing real
 estate.
- compliance with debt and operating covenants, availability and cost of credit, changes in our credit rating and changes in interest rates,
- the actual timing, price, manner and amounts of future share repurchases, dividend payments, or share issuances, if any,

Economic and External

- the length and severity of epidemics or pandemics, such as the COVID-19 pandemic, or other catastrophic events, and the related impact on customer behavior, store and online operations and supply chain functions, as well as our future consolidated financial position, results of operations and cash flows,
- the impact of the seasonal nature of our business and cyclical customer spending,
- the impact of economic and market conditions in the U.S. and Canada, including inflation and the resulting changes to our customer purchasing behavior, unemployment and bankruptcy rates, as well as any fiscal stimulus and the resultant impact on consumer spending and credit patterns.
- the impact of economic, environmental or political conditions in the U.S. and Canada and countries where our third-party vendors operate,
- the impact of changing traffic patterns at shopping centers and malls,
- financial insecurity or potential insolvency experienced by our vendors, suppliers, landlords, peers, or customers as a result of any economic downturn,
- weather conditions, natural disasters, climate change, national security concerns, other market and supply chain disruptions, the effects
 of tariffs, or the prospects of these events and the resulting impact on consumer spending patterns or information technology systems
 and communications,

Legal and Regulatory

- our compliance with applicable domestic and international laws, regulations and ethical standards, including those related to COVID-19, minimum wage, employment and tax, information security and privacy, consumer credit and environmental regulations and the outcome of any claims, litigation and regulatory investigations and resolution of such matters,
- the impact of the current regulatory environment, financial system and tax reforms,
- the impact of changes in accounting rules and regulations, changes in our interpretation of the rules or regulations, or changes in underlying assumptions, estimates or judgments.

These and other factors, including those factors we discussed in Part II, Ltem 1A: Risk Factors, could affect our financial results and cause our actual results to differ materially from any forward-looking information we may provide. Given these risks, uncertainties and other factors, you should not place undue reliance on these forward-looking statements. Also, these forward-looking statements represent our estimates and assumptions only as of the date of this filing. You should read this Quarterly Report on Form 10-Q completely and with the understanding that our actual future results may be materially different from what we expect. We hereby qualify our forward-looking statements by these cautionary statements. Except as required by law, we assume no obligation to update these forward-looking statements publicly, or to update the reasons actual results could differ materially from those anticipated in these forward-looking statements, even if new information becomes available in the future.

All references to "we," "us," "our," or the "Company" mean Nordstrom, Inc. and its subsidiaries.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based upon information available to us as of the filing date of this Quarterly Report on Form 10-Q, and while we believe such information forms a reasonable basis for such statements, such information may be limited or incomplete, and our statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain and investors are cautioned not to unduly rely upon these statements.

DEFINITIONS

The following table includes definitions of our commonly used terms:

	Definition
2019 Plan	2019 Equity Incentive Plan
2020 Annual Report	Annual Report on Form 10-K filed on March 15, 2021
Adjusted EBITDA	Adjusted earnings (loss) before interest, income taxes, depreciation and amortization (a non-GAAP financial measure)
Adjusted EBITDAR	Adjusted earnings (loss) before interest, income taxes, depreciation, amortization and rent, as defined by our Revolver covenant (a non-GAAP financial measure)
Adjusted ROIC	Adjusted return on invested capital (a non-GAAP financial measure)
ASC	Accounting Standards Codification
ASU	Accounting Standards Update
CARES Act	Coronavirus Aid, Relief and Economic Security Act
CODM	Chief operating decision maker
COVID-19	Novel coronavirus
Digital sales	Sales conducted through a digital platform such as our websites or mobile apps. Digital sales may be self-guided by the customer, as in a traditional online order, or facilitated by a salesperson using a virtual styling or selling tool, such as Nordstrom Trunk Club or Style Board. Digital sales may be picked up in our Nordstrom stores, Nordstrom Rack stores or Nordstrom Local service hubs. Digital sales also include a reserve for estimated returns.
EBIT	Earnings (Loss) before interest and income taxes
EPS	Earnings (Loss) per share
ESPP	Employee Stock Purchase Plan
Exchange Act	Securities Exchange Act of 1934, as amended
Express Services	Nordstrom order pickups and returns offered at certain Nordstrom Rack stores
FASB	Financial Accounting Standards Board
First quarter of 2021	13 fiscal weeks ending May 1, 2021
First quarter of 2020	13 fiscal weeks ending May 2, 2020
First quarter of 2019	13 fiscal weeks ending May 4, 2019
Fiscal year 2021	52 fiscal weeks ending January 29, 2022
Fiscal year 2020	52 fiscal weeks ending January 30, 2021
GAAP	Generally accepted accounting principles
Gross profit	Net sales less cost of sales and related buying and occupancy costs
Inventory turnover rate	Trailing 4-quarter cost of sales and related buying and occupancy costs divided by the trailing 4-quarter average inventory
Lease Standard	ASU No. 2016-02, Leases, and all related amendments (ASC 842)
Leverage Ratio	The sum of the preceding twelve months of rent expense under the previous lease guidance multiplied by six and funded debt divided by the preceding twelve months of EBITDAR as defined by our debt covenant
MD&A	Management's Discussion and Analysis of Financial Condition and Results of Operations
Nordstrom	Nordstrom.com, TrunkClub.com, Nordstrom-branded U.S. stores, Canada, which includes Nordstrom.ca, Nordstrom Canadian stores and Nordstrom Rack Canadian stores, and Nordstrom Local
Nordstrom Local	Nordstrom Local service hubs, which offer Nordstrom order pickups, returns, alterations and other services
Nordstrom NYC	Our New York City Nordstrom flagship store, including the Men's location
Nordstrom Rack	Nordstromrack.com, Nordstrom Rack-branded U.S. stores, Last Chance clearance stores and, prior to the first quarter of 2021, HauteLook.com
The Nordy Club	Our customer loyalty program
NYSE	New York Stock Exchange
Operating Lease Cost	Fixed rent expense, including fixed common area maintenance expense, net of developer reimbursement amortization
PCAOB	Public Company Accounting Oversight Board (United States)
Property incentives	Developer and vendor reimbursements
PSU	Performance share unit
Revolver	Senior revolving credit facility
ROU asset	Operating lease right-of-use asset
RSU	Restricted stock unit
SEC	Securities and Exchange Commission
SERP	Unfunded defined benefit Supplemental Executive Retirement Plan
Secured Notes	8.750% senior secured notes due May 2025
CCOA	Selling, general and administrative
SG&A	
Supply Chain Network	Fulfillment centers that primarily process and ship orders to our customers, distribution centers that primarily process and ship merchandise to our stores and other facilities and omni-channel centers that both fulfill customer orders and ship merchandise to our stores

PART I — FINANCIAL INFORMATION

Item 1. Financial Statements (Unaudited).

NORDSTROM, INC. CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS

(Amounts in millions except per share amounts) (Unaudited)

	Quarter Ended	
	May 1, 2021	May 2, 2020
Net sales	\$2,921	\$2,026
Credit card revenues, net	88	93
Total revenues	3,009	2,119
Cost of sales and related buying and occupancy costs	(2,019)	(1,810)
Selling, general and administrative expenses	(1,075)	(1,122)
Loss before interest and income taxes	(85)	(813)
Interest expense, net	(137)	(34)
Loss before income taxes	(222)	(847)
Income tax benefit	56	326
Net loss	(\$166)	(\$521)
Loss per share:		
Basic	(\$1.05)	(\$3.33)
Diluted	(\$1.05)	(\$3.33)
Weighted-average shares outstanding:		
Basic	158.5	156.4
Diluted	158.5	156.4

The accompanying Notes to Condensed Consolidated Financial Statements are an integral part of these financial statements.

NORDSTROM, INC. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE EARNINGS

(Amounts in millions) (Unaudited)

	Quarter E	Quarter Ended	
	May 1, 2021	May 2, 2020	
Net loss	(\$166)	(\$521)	
Foreign currency translation adjustment	10	(24)	
Post retirement plan adjustments, net of tax	1	2	
Comprehensive net loss	(\$155)	(\$543)	

NORDSTROM, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

(Amounts in millions) (Unaudited)

	May 1, 2021	January 30, 2021	May 2, 2020
Assets			
Current assets:			
Cash and cash equivalents	\$377	\$681	\$1,355
Accounts receivable, net	238	245	154
Merchandise inventories	1,961	1,863	1,489
Prepaid expenses and other	923	853	669
Total current assets	3,499	3,642	3,667
Land, property and equipment (net of accumulated depreciation of \$7,322, \$7,159 and \$6,683)	3,642	3,732	3,974
Operating lease right-of-use assets	1,560	1,581	1,722
Goodwill	249	249	249
Other assets	383	334	357
Total assets	\$9,333	\$9,538	\$9,969
Liabilities and Shareholders' Equity			
Current liabilities:			
Borrowings under revolving line of credit	\$200	\$—	\$800
Accounts payable	1,676	1,960	1,125
Accrued salaries, wages and related benefits	330	352	280
Current portion of operating lease liabilities	246	260	243
Other current liabilities	1,056	1,048	1,351
Current portion of long-term debt	500	500	_
Total current liabilities	4,008	4,120	3,799
Long-term debt, net	2,847	2,769	3,264
Non-current operating lease liabilities	1,662	1,687	1,836
Other liabilities	650	657	673
Commitments and contingencies (Note 5)			
Shareholders' equity:			
Common stock, no par value: 1,000 shares authorized; 158.9, 157.8 and 157.0 shares issued and outstanding	3,221	3,205	3,148
Accumulated deficit	(2,996)	(2,830)	(2,661)
Accumulated other comprehensive loss	(59)	(70)	(90)
Total shareholders' equity	166	305	397
Total liabilities and shareholders' equity	\$9,333	\$9,538	\$9,969

NORDSTROM, INC. CONDENSED CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY

(Amounts in millions except per share amounts) (Unaudited)

	Quarter End	Quarter Ended	
	May 1, 2021	May 2, 2020	
Common stock			
Balance, beginning of period	\$3,205	\$3,129	
Issuance of common stock under stock compensation plans	7	11	
Stock-based compensation	9	8	
Balance, end of period	\$3,221	\$3,148	
A commission of the state of th			
Accumulated deficit	(44 444)	(+0.000)	
Balance, beginning of period	(\$2,830)	(\$2,082)	
Net loss	(166)	(521)	
Dividends	_	(58)	
Balance, end of period	(\$2,996)	(\$2,661)	
Accumulated other comprehensive loss			
Balance, beginning of period	(\$70)	(\$68)	
Other comprehensive income (loss)	11	(22)	
Balance, end of period	(\$59)	(\$90)	
Total Shareholders' Equity	\$166	\$397	
Dividends per share	\$—	\$0.37	

NORDSTROM, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(Amounts in millions) (Unaudited)

	Quarter Ended	
	May 1, 2021	May 2, 2020
Operating Activities		
Net loss	(\$166)	(\$521)
Adjustments to reconcile net loss to net cash used in operating activities:		
Depreciation and amortization expenses	162	176
Asset impairment	-	117
Right-of-use asset amortization	43	44
Deferred income taxes, net	8	(54)
Stock-based compensation expense	22	13
Other, net	86	2
Change in operating assets and liabilities:		
Accounts receivable	7	25
Merchandise inventories	(16)	228
Prepaid expenses and other assets	(126)	(393)
Accounts payable	(296)	(292)
Accrued salaries, wages and related benefits	(22)	(227)
Other current liabilities	7	167
Lease liabilities	(81)	(65)
Other liabilities	8	2
Net cash used in operating activities	(364)	(778)
The cash asea in operating activities	(55-7)	(110)
Investing Activities		
Capital expenditures	(126)	(131)
Other, net	16	5
Net cash used in investing activities	(110)	(126)
Financian Astivitics		
Financing Activities	200	800
Proceeds from revolving line of credit		
Proceeds from long-term borrowings	675	600
Principal payments on long-term borrowings	(600)	_
(Decrease) increase in cash book overdrafts	(17)	83
Cash dividends paid		(58)
Proceeds from issuances under stock compensation plans	7	11
Tax withholding on share-based awards	(13)	(8)
Make-whole payment and other, net	(85)	(11)
Net cash provided by financing activities	167	1,417
Effect of exchange rate changes on cash and cash equivalents	3	(11)
Net (decrease) increase in cash and cash equivalents	(304)	502
Cash and cash equivalents at beginning of period	681	853
	\$377	\$1,355
Cash and cash equivalents at end of period	\$311	Φ1,355
Supplemental Cash Flow Information		
Cash paid during the period for:		
Income taxes, net	\$3	\$—
Interest, net of capitalized interest	63	34

(Dollar and share amounts in millions except per share, per option and per unit amounts) (Unaudited)

NOTE 1: BASIS OF PRESENTATION

The accompanying Condensed Consolidated Financial Statements include the balances of Nordstrom, Inc. and its subsidiaries. All intercompany transactions and balances are eliminated in consolidation. The interim Condensed Consolidated Financial Statements have been prepared on a basis consistent in all material respects with the accounting policies described and applied in our 2020 Annual Report and reflect all adjustments of a normal recurring nature that are, in management's opinion, necessary for the fair presentation of the results of operations, financial position and cash flows for the periods presented.

The Condensed Consolidated Financial Statements as of and for the periods ended May 1, 2021 and May 2, 2020 are unaudited. The Condensed Consolidated Balance Sheet as of January 30, 2021 has been derived from the audited Consolidated Financial Statements included in our 2020 Annual Report. The interim Condensed Consolidated Financial Statements should be read together with the Consolidated Financial Statements and related footnote disclosures contained in our 2020 Annual Report.

Seasonality

Our business, like that of other retailers, is subject to seasonal fluctuations and cyclical trends in consumer spending. Our sales are typically higher in our second quarter, which usually includes our Anniversary Sale, and the holidays in the fourth quarter. As a result of COVID-19, the Anniversary Sale was moved to August in 2020, which fell entirely in our third fiscal quarter. Results for any one quarter may not be indicative of the results that may be achieved for a full fiscal year. We plan our merchandise receipts to coincide with expected sales trends and other supply chain factors. For instance, our merchandise purchases and receipts increase prior to the Anniversary Sale, and we purchase and receive a larger amount of merchandise in the fall as we prepare for the holiday shopping season (from late November through December). Consistent with our seasonal fluctuations, our working capital requirements have historically increased during the months leading up to the Anniversary Sale and the holidays as we purchase inventory in anticipation of increased sales.

Use of Estimates

The preparation of financial statements in conformity with GAAP in the U.S. requires that we make estimates, judgments and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses and disclosure of contingent assets and liabilities during the reporting period. Uncertainties regarding such estimates and assumptions are inherent in the preparation of financial statements and actual results may differ from these estimates and assumptions. Our most significant accounting judgments and estimates include revenue recognition, inventory valuation, long-lived asset recoverability and income taxes, all of which involve assumptions about future events. We may be unable to accurately predict the ongoing impact of COVID-19 going forward and as a result our estimates may change in the near term.

Land, Property and Equipment

Our net non-cash investing activities primarily related to technology expenditures resulted in a decrease to accounts payable of \$44 in 2021.

Long-Lived Assets

In the first quarter of 2020, as we optimized our mix of physical and digital assets to align with longer-term customer trends, we closed 16 Nordstrom stores. In conjunction with these closures, we incurred non-cash impairment charges on long-lived tangible and ROU assets to adjust the carrying values to their estimated fair value. The following table provides details related to asset impairment charges as a result of COVID-19:

Quarter Ended	May 2, 2020
Long-lived asset impairment ¹	\$94
Operating lease ROU asset impairment ¹	23
Total asset impairment	\$117

¹ As of May 2, 2020, the carrying value of the applicable long-lived and operating lease ROU assets after impairment was \$15 and \$6.

These charges are primarily included in our Retail segment SG&A expense on the Condensed Consolidated Statement of Earnings.

Income Taxes

We recorded \$576 and \$275 in current taxes receivable as of May 1, 2021 and May 2, 2020, which is classified in prepaid expenses and other on the Condensed Consolidated Balance Sheet. In addition, we recorded \$60 and \$2 in noncurrent taxes receivable as of May 1, 2021 and May 2, 2020, which is classified in other assets on the Condensed Consolidated Balance Sheet. The current and noncurrent income tax receivables are primarily associated with the loss carryback provision of the CARES Act.

Severance

In the first quarter of 2020, we recorded \$88 of restructuring costs in connection with our regional and corporate reorganization, including \$25 recorded in cost of sales and related buying and occupancy costs and \$63 in SG&A on the Condensed Consolidated Statement of Earnings.

(Dollar and share amounts in millions except per share, per option and per unit amounts)
(Unaudited)

Leases

We incurred operating lease liabilities arising from the commencement of lease agreements of \$33 for the quarter ended May 1, 2021 and \$37 for the guarter ended May 2, 2020.

Reclassification

We reclassified our fiscal 2020 Condensed Consolidated Statement of Cash Flows to conform with current period presentation. To adjust our net loss to reconcile to operating activity cash flows, we present depreciation and amortization separate from other, net, which includes the "make-whole" premium in the first quarter of 2021 (see Note 3: Debt and Credit Facilities).

Recent Accounting Pronouncements

In November 2020, the SEC adopted the final rule under SEC Release No. 33-10890, Management's Discussion and Analysis, Selected Financial Data, and Supplementary Financial Information, which eliminates the requirement for selected financial data, streamlines certain disclosures in MD&A and eliminates duplicative disclosures with the intention of simplifying reporting compliance. In the fourth quarter of 2020, we early adopted the amendments to provision 301, Selected financial data and provision 302, Supplementary financial information. In the first quarter of 2021, we adopted all remaining provisions of this final rule. The adoption of this final rule did not have a material effect on our Consolidated Financial Statements.

NOTE 2: REVENUE

Contract Liabilities

Contract liabilities represent our obligation to transfer goods or services to customers and include deferred revenue for The Nordy Club (including points and Nordstrom Notes) and gift cards. Our contract liabilities are classified as current on the Condensed Consolidated Balance Sheets and are as follows:

	Contract Liabilities
Balance as of February 1, 2020	\$576
Balance as of May 2, 2020	489
Balance as of January 30, 2021	478
Balance as of May 1, 2021	436

Revenues recognized from our beginning contract liability balance were \$114 for the quarter ended May 1, 2021 and \$130 for the quarter ended May 2, 2020.

Disaggregation of Revenue

The following table summarizes our disaggregated net sales:

	Quarter Endec	Quarter Ended	
	May 1, 2021	May 2, 2020	
Nordstrom	\$1,854	\$1,357	
Nordstrom Rack	1,067	669	
Total net sales	\$2,921	\$2,026	
Digital sales as a % of total net sales	46 %	54 %	

(Dollar and share amounts in millions except per share, per option and per unit amounts)
(Unaudited)

The following table summarizes the percent of net sales by merchandise category:

	Quarter End	Quarter Ended	
	May 1, 2021	May 2, 2020	
Women's Apparel	31 %	33 %	
Shoes	26 %	24 %	
Women's Accessories	14 %	12 %	
Men's Apparel	12 %	12 %	
Beauty	11 %	12 %	
Kids' Apparel	4 %	4 %	
Other	2 %	3 %	
Total net sales	100 %	100 %	

NOTE 3: DEBT AND CREDIT FACILITIES

Debt

During the first quarter of 2021, we issued \$250 aggregate principal amount of 2.3% senior notes due April 2024 and \$425 aggregate principal amount of 4.25% senior notes due August 2031. These notes are unsecured and may be redeemed at any time in whole or in part. The April 2024 notes can be redeemed at par starting in April 2022. With the net proceeds of these new notes, together with cash on hand, we retired our \$600 Secured Notes. We recorded \$88 related to the redemption in interest expense, net, which primarily consisted of a one-time payment of \$78 for a "make-whole" premium, and the write-off of unamortized balances associated with the debt discount and issuance costs. The make-whole payment was not included in cash paid during the period for interest, net of capitalized interest.

As a result of this redemption, our outstanding long-term debt is unsecured and all real estate will be unencumbered.

Credit Facilities

During the first quarter of 2021, we borrowed \$200 and amended our existing Revolver. Under the Revolver, we are in a "Collateral Period" if our Leverage Ratio is greater than four or our unsecured debt is rated below BBB- with a stable outlook at Standard & Poor's or Baa3 with a stable outlook at Moody's. In the Collateral Period, any borrowings under our Revolver will be secured by substantially all our personal property and we will be subject to asset coverage and minimum liquidity covenants, as well as a fixed charge coverage covenant. If our Leverage Ratio is below four and our unsecured debt is rated at or above BBB- with a stable outlook at Standard & Poor's or Baa3 with a stable outlook at Moody's, any borrowings under our Revolver will be unsecured, we will not be subject to the above covenants and the restrictions on dividend payments and share repurchases will be removed. In May 2021, subsequent to quarter end, we completely repaid \$200 on our Revolver.

Under our Revolver amendment, we created flexibility for dividends and share repurchases during the Collateral Period, provided no default or event of default exists as a result of such payments, the pro-forma Leverage Ratio as of the most recent fiscal quarter is less than 3.75, pro-forma liquidity at the date of such payment is at least \$600, and the amount of such payments do not exceed the amount of the corresponding fiscal quarter of 2019. Additionally, the "make-whole" premium and unamortized deferred bond issuance costs related to the redemption of the \$600 Secured Notes is excluded from the definition of interest expense.

The Revolver expires in September 2023 and is classified in total current liabilities on the Condensed Consolidated Balance Sheet. As of May 1, 2021, we had \$200 outstanding under this facility, our borrowings under the Revolver were classified as secured as our Leverage Ratio exceeded four and we did not meet or exceed our credit rating threshold. We met all other financial covenant measures for the quarter. Provided that we obtain written consent from the lenders, we have the option to increase the Revolver by up to \$200, to a total of \$1,000, and two options to extend the Revolver by one year. The Revolver contains customary representations, warranties, covenants and terms, including paying a variable rate of interest and a commitment fee based on our debt rating. The Revolver is available for working capital, capital expenditures and general corporate purposes.

As a result of our borrowings under the Revolver, the full capacity of our \$800 commercial paper program is not available to us at this time. When available, the program allows us to use the proceeds to fund operating cash requirements. Under the terms of the commercial paper agreement, we pay a rate of interest based on, among other factors, the maturity of the issuance and market conditions. The issuance of commercial paper has the effect of reducing available liquidity under the Revolver by an amount equal to the principal amount of commercial paper outstanding. As of May 1, 2021, we had no issuances outstanding under our commercial paper program.

(Dollar and share amounts in millions except per share, per option and per unit amounts) (Unaudited)

NOTE 4: FAIR VALUE MEASUREMENTS

We disclose our financial assets and liabilities that are measured at fair value in our Condensed Consolidated Balance Sheets by level within the fair value hierarchy as defined by applicable accounting standards:

- Level 1: Quoted market prices in active markets for identical assets or liabilities
- Level 2: Other observable market-based inputs or unobservable inputs that are corroborated by market data
- Level 3: Unobservable inputs that cannot be corroborated by market data that reflect the reporting entity's own assumptions

Financial Instruments Measured at Carrying Value

Financial instruments measured at carrying value on a recurring basis include cash and cash equivalents, accounts receivable, accounts payable and our Revolver, which approximate fair value due to their short-term nature.

Long-term debt is recorded at carrying value. If long-term debt was measured at fair value, we would use quoted market prices of the same or similar issues, which is considered a Level 2 fair value measurement. The following table summarizes the carrying value and fair value estimate of our long-term debt, including current maturities:

	May 1, 2021	January 30, 2021	May 2, 2020
Carrying value of long-term debt	\$3,347	\$3,269	\$3,264
Fair value of long-term debt	3,480	3,430	2,804

Non-financial Assets Measured at Fair Value on a Nonrecurring Basis

We also measure certain non-financial assets at fair value on a nonrecurring basis, primarily goodwill, long-lived tangible and ROU assets, in connection with periodic evaluations for potential impairment. We estimate the fair value of these assets using primarily unobservable inputs and, as such, these are considered Level 3 fair value measurements. There were no material impairment charges for these assets for the quarter ended May 1, 2021. For more information regarding long-lived tangible and ROU asset impairment charges for the quarter ended May 2, 2020, see Note 1: Basis of Presentation.

NOTE 5: COMMITMENTS AND CONTINGENCIES

Our NYC flagship store opened in October 2019 and the related building and equipment assets were placed into service at the end of the third quarter of 2019. While our store has opened, construction continues in the residential condominium units above the store. As of May 1, 2021, we have a fee interest in the retail condominium unit. We are committed to make one remaining installment payment based on the developer meeting final pre-established construction and development milestones. Precautions related to COVID-19 have caused delays in meeting these milestones and the timing of the remaining payment.

(Dollar and share amounts in millions except per share, per option and per unit amounts)
(Unaudited)

NOTE 6: SHAREHOLDERS' EQUITY

In August 2018, our Board of Directors authorized a program to repurchase up to \$1,500 of our outstanding common stock, with no expiration date. On March 23, 2020, in response to uncertainty from the COVID-19 pandemic, we announced the suspension of our quarterly dividend payments beginning in the second quarter of 2020 and the immediate suspension of our share repurchase program. We remain committed to these programs over the long-term and intend to resume dividend payments and share repurchases when appropriate. We had \$707 remaining in share repurchase capacity as of May 1, 2021. The actual timing, price, manner and amounts of future share repurchases, if any, will be subject to the discretion of the Board of Directors, contractual commitments, market and economic conditions, and applicable SEC rules.

Our Revolver contains negative covenants with respect to the payment of dividends and share repurchases. As of May 1, 2021, our Leverage Ratio exceeded four and we did not meet our credit rating covenant, preventing us from paying dividends or repurchasing shares. For more information regarding our debt covenants, see Note 3: Debt and Credit Facilities.

NOTE 7: STOCK-BASED COMPENSATION

The following table summarizes our stock-based compensation expense:

	Quarter End	Quarter Ended	
	May 1, 2021	May 2, 2020	
RSUs	\$14	\$13	
Stock options	7	2	
Other ¹	1	(2)	
Total stock-based compensation expense, before income tax benefit	22	13	
Income tax benefit	(6)	(5)	
Total stock-based compensation expense, net of income tax benefit	\$16	\$8	

¹ Other stock-based compensation expense includes PSUs, ESPP and nonemployee director stock awards.

The following table summarizes our grant allocations:

	Quarter Ended			
	May 1,	2021	May 2	, 2020
	Granted	Weighted-average grant-date fair value per unit	Granted	Weighted-average grant-date fair value per unit
RSUs	1.4	\$33	2.2	\$23
Stock options	1.2	\$13	0.3	\$7
PSUs	_	_	0.4	\$24

Under our deferred and stock-based compensation plan arrangements, we issued 1.1 shares of common stock during the first quarter of 2021 and 1.5 shares during the first quarter of 2020.

(Dollar and share amounts in millions except per share, per option and per unit amounts) (Unaudited)

NOTE 8: EARNINGS PER SHARE

The computation of EPS is as follows:

	Quarter E	Quarter Ended	
	May 1, 2021	May 2, 2020	
Net loss	(\$166)	(\$521)	
Basic shares	158.5	156.4	
Dilutive effect of common stock equivalents	_	_	
Diluted shares	158.5	156.4	
Loss per basic share	(\$1.05)	(\$3.33)	
Loss per diluted share	(\$1.05)	(\$3.33)	
Anti-dilutive common stock equivalents	12.7	13.1	

NOTE 9: SEGMENT REPORTING

The following table sets forth information for our reportable segment:

	Quarter Ended	
	May 1, 2021	May 2, 2020
Retail segment EBIT	(\$55)	(\$711)
Corporate/Other EBIT	(30)	(102)
Interest expense, net	(137)	(34)
Loss before income taxes	(\$222)	(\$847)

For information about disaggregated revenues, see Note 2: Revenue.

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Dollar and share amounts in millions except per share amounts)

OVERVIEW

Our first quarter results reflected improving sales momentum and continued progress in our transformation as we work to unlock the full potential of our digital-first platform. Net loss for the first quarter was \$166, or \$1.05 per diluted share, including an after-tax debt refinancing charge of \$0.41. First quarter net sales increased 44% from the same period in 2020 and decreased 13% from the same period in 2019, representing a sequential improvement of 720 basis points relative to the fourth quarter of 2020.

Our top-line trends increased sequentially for the third quarter in a row, with improvements in both Nordstrom and Nordstrom Rack, supported by recovery in stores as COVID-19 restrictions were lifted, and continued growth in digital. Sales trends reflected broad-based improvement across channels, regions and merchandise categories, both in-store and online. Stores in markets that opened up earlier outperformed other markets by 7 to 10 percentage points, giving us increasing optimism about the pace of recovery as we look to the remainder of the year. Our performance in the quarter reflects solid execution toward the growth priorities we laid out at our investor day in February: win in our most important markets, broaden the reach of Nordstrom Rack and increase our digital velocity.

Market Strategy – Our market strategy helps us engage with customers through better service and greater access to product, no matter how they choose to shop. During the quarter, we successfully expanded this strategy to our top 20 markets, which comprise approximately 75% of sales. We continued to scale the enhanced capabilities we launched in 2020, such as the expansion of order pick-up and ship-to-store to all Nordstrom Rack stores. Almost one-third of next day order pickup volume for Nordstrom.com in our top 20 markets was picked up at Rack stores, as we continue to integrate our capabilities across our two powerful brands.

Nordstrom Rack – First quarter Nordstrom Rack sales declined 13% compared with 2019, a 10-percentage point sequential improvement from the fourth quarter of 2020. Merchandise repositioning across price, hybrid and brand doors is progressing, in spite of challenges managing slower than anticipated inbound inventory flow. We remain in the early stages of these initiatives, and our progress is encouraging. Increased customer choice of price-oriented offerings in Kids, Home and Active supported a 37% increase in sales compared with 2019 in these categories.

Digital Velocity – We maintained strong growth at Nordstrom.com and Nordstromrack.com in the first quarter, even as store traffic and sales rebounded. Digital sales increased 23% over last year and 28% over the first quarter of 2019. With continued growth in digital, our total penetration has increased by 15 percentage points over the past two years, to 46%. One key opportunity we see is to offer our customers more choices, with plans to increase choice count to approximately 1.5 million over the next several years. This quarter, choice counts increased almost 20% compared with 2019, primarily driven by an expanded dropship assortment in both our core categories and in-demand categories like Home, Active and Kids. This allowed us to drive strong sales growth in our digital business without a corresponding increase in our inventory investment.

As we look ahead to the second quarter, we believe that our Anniversary Sale will be well-timed to benefit from customers' increasing confidence and return to pre-pandemic activities. Our goal is to have an event that rewards and engages our best customers with a superior shopping experience. We will also significantly increase selection for Anniversary this year, with total customer choices up double-digits compared with 2019, supported by an expansion of alternative partnership models with our vendors.

Our focus on accelerating our strategic priorities to serve customers in new and differentiated ways is gaining momentum. We are in a stronger position than ever to capitalize on our market share opportunity as customer demand recovers. While there is still considerable uncertainty with respect to COVID-19, we remain confident in our ability to deliver on our targets for 2021 and generate profitable sales growth as demand recovers.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

RESULTS OF OPERATIONS

In our ongoing effort to enhance the customer experience, we are focused on providing a seamless experience across our Company. We invested early in our omni-channel capabilities, integrating our operations, merchandising and technology across our stores and online, and in both Nordstrom and Nordstrom Rack brands. While our customers may engage with us through multiple ways, we know they value the integrated brand experience, which is ultimately how we view our company. We have one Retail reportable segment and analyze our results on a total company basis, using customer, market share, operational and net sales metrics.

Due to the extraordinary impact of COVID-19 on our results in fiscal 2020, we analyzed fiscal year 2021 net sales through EBIT against both fiscal years 2020 and 2019 to provide useful supplemental comparability.

Net Sales

The following table summarizes net sales:

	Quarter Ended		
	May 1, 2021	May 2, 2020	May 4, 2019
Net sales:			
Nordstrom	\$1,854	\$1,357	\$2,127
Nordstrom Rack	1,067	669	1,222
Total net sales	\$2,921	\$2,026	\$3,349
Net sales increase (decrease):			
Nordstrom	36.7 %	(36.2 %)	(5.1 %)
Nordstrom Rack	59.5 %	(45.2 %)	(0.6 %)
Total Company	44.2 %	(39.5 %)	(3.5 %)
Digital sales as a % of total net sales	46 %	54 %	31 %
Digital sales increase	23 %	5 %	7 %

Net Sales (2021 vs. 2020)

Total Company net sales increased 44% for the first quarter of 2021, compared with the same period in 2020, during which stores were temporarily closed for approximately half of the quarter due to COVID-19. Total Company digital sales increased 23% in the first quarter of 2021, compared with the same period in 2020 and represented 46% of total net sales during the first quarter of 2021. During the quarter ended May 1, 2021, we closed one Nordstrom Rack store.

Nordstrom net sales increased 37% for the first quarter of 2021, compared with the same period in 2020. Nordstrom Rack net sales increased 59% for the first quarter of 2021, compared with the same period in 2020. Home, active, designer and beauty were the top-performing merchandise categories.

Net Sales (2021 vs. 2019)

Total Company net sales decreased 13% for the first quarter of 2021, compared with the same period in 2019, and marked sequential improvement of 720 basis points relative to the fourth quarter of 2020. Total Company digital sales increased 28% in the first quarter of 2021, compared with the same period in 2019.

Nordstrom net sales decreased 13% for the first quarter of 2021, compared with the same period in 2019. Nordstrom Rack net sales decreased 13% for the first quarter of 2021, compared with the same period in 2019. Home, active and kids were the top-performing merchandise categories.

Credit Card Revenues, Net

Credit card revenues, net include our portion of the ongoing credit card revenue, net of credit losses, pursuant to our program agreement with TD. TD is the exclusive issuer of our consumer credit cards and we perform the account servicing functions.

Credit Card Revenues, Net (2021 vs. 2020 and 2021 vs. 2019)

Credit card revenues, net were \$88 for the quarter ended May 1, 2021, compared with \$93 and \$94 for the same periods in 2020 and 2019. These decreases were primarily a result of lower finance charges and late fee revenues throughout the first quarter of 2021 due to lower outstanding balances as consumers made higher payments, driven in part by government stimulus packages.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Gross Profit

The following table summarizes gross profit:

		Quarter Ended	
	May 1, 2021	May 2, 2020	May 4, 2019
Gross profit	\$902	\$216	\$1,121
Gross profit as a % of net sales	30.9 %	10.7 %	33.5 %
Inventory turnover rate	4.51	4.81	4.68

Gross Profit (2021 vs. 2020)

Gross profit increased \$686 and 20 percentage points as a rate of net sales during the first quarter of 2021, compared with the same period in 2020, primarily due to lower markdowns and leverage from higher net sales volume.

Gross Profit (2021 vs. 2019)

Gross profit decreased \$219 and 260 basis points during the first quarter of 2021, compared with the same period in 2019, as a result of deleverage on lower sales and lower merchandise margins, partially offset by permanent reductions in buying and occupancy costs.

Ending inventory decreased 2% compared with the same period in 2019, versus a 13% decrease in sales. The change in inventory levels compared with 2019 includes an approximately 700 basis point impact resulting from the acceleration of vendor shipments to support sales trends and mitigate potential supply chain backlogs in the second quarter.

Selling, General and Administrative Expenses

SG&A is summarized in the following table:

		Quarter Ended	
	May 1, 2021	May 2, 2020	May 4, 2019
SG&A expenses	\$1,075	\$1,122	\$1,138
SG&A expenses as a % of net sales	36.8 %	55.4 %	34.0 %

SG&A (2021 vs. 2020)

SG&A decreased \$47 and 19 percentage points as a rate of net sales during the first quarter of 2021, compared with the same period in 2020, as a result of \$250 in charges associated with the impact of COVID-19 in 2020, leverage on higher sales and the continued benefit of permanent reductions in overhead expenses of approximately 15%. This was partially offset by higher variable expenses such as supply chain costs associated with the sales volume increase.

SG&A (2021 vs. 2019)

SG&A decreased \$63 during the first quarter of 2021, compared with the same period in 2019, primarily due to planned cost savings initiatives and lower sales volume, partially offset by COVID-19 related labor and freight cost pressures. SG&A rate increased 280 basis points during the first quarter of 2021, compared with the same period in 2019, as a result of COVID-19 related labor and freight cost pressures, partially offset by our planned savings initiatives.

(Loss) Earnings Before Interest and Income Taxes

EBIT is summarized in the following table:

		Quarter Ended	
	May 1, 2021	May 2, 2020	May 4, 2019
EBIT	(\$85)	(\$813)	\$77
EBIT as a % of net sales	(2.9 %)	(40.1 %)	2.3 %

EBIT (2021 vs. 2020)

EBIT improved \$728 during the first quarter of 2021, compared with the same period in 2020. The increase was due to higher sales volume, lower markdowns and \$280 in charges in the first quarter of 2020 related to the impacts of COVID-19.

EBIT (2021 vs. 2019)

For the first quarter of 2021, EBIT decreased \$162, compared with the same period in 2019 primarily due to lower sales volume as we continue to recover from the pandemic and higher freight cost pressures, partially offset by our planned savings initiatives.

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Interest Expense, Net

Interest expense, net was \$137 for the first quarter of 2021, compared with \$34 for the same period in 2020. The increase was primarily due to the debt refinance charges of \$88 related to the redemption of the Secured Notes.

Income Tax Expense

Income tax expense is summarized in the following table:

	Quarter Ended	
	May 1, 2021	May 2, 2020
Income tax (benefit) expense	(\$56)	(\$326)
Effective tax rate	25.4 %	38.4 %

The effective tax rate decreased in the first quarter of 2021, compared with the same period in 2020, primarily due to additional tax benefits recorded in 2020 associated with losses eligible for carryback under the CARES Act. The decrease was partially offset by non-deductible stock compensation.

Earnings Per Share

EPS is as follows:

	Quarter	Quarter Ended	
	May 1, 2021	May 2, 2020	
Basic	(\$1.05)	(\$3.33)	
Diluted	(\$1.05)	(\$3.33)	

Earnings per diluted share increased \$2.28 for the first quarter of 2021, compared with the same period in 2020, during which stores were temporarily closed for approximately half of the quarter, partially offset by an interest expense charge of \$88, or \$0.41 per diluted share related to the redemption of the Secured Notes in the first quarter of 2021. In the first quarter of 2020, COVID-19 related charges reduced earnings per diluted share by \$1.10.

Fiscal Year 2021 Outlook

The Company has reaffirmed the following financial expectations for fiscal 2021:

- Revenue, including retail sales and credit card revenues, is expected to grow more than 25%,
- EBIT margin is expected to be approximately 3% of sales,
- · Income tax rate is expected to be approximately 27%,
- Leverage ratio is expected to be approximately 3x by year-end,
- For the first half of the year, EBIT is expected to be approximately breakeven, reflecting approximately 45% of total year sales.

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Adjusted ROIC (Non-GAAP financial measure)

We believe that Adjusted ROIC is a useful financial measure for investors in evaluating the efficiency and effectiveness of the capital we have invested in our business to generate returns over time. In addition, we have incorporated it in our executive incentive measures and we believe it is an important indicator of shareholders' return over the long term.

Adjusted ROIC is not a measure of financial performance under GAAP and should be considered in addition to, and not as a substitute for, return on assets, net earnings, total assets or other GAAP financial measures. Our method of calculating non-GAAP financial measures may differ from other companies' methods and therefore may not be comparable to those used by other companies. The financial measure calculated under GAAP which is most directly comparable to Adjusted ROIC is return on assets.

The following is a reconciliation of return on assets to Adjusted ROIC:

	Four Quarters En	Four Quarters Ended	
	May 1, 2021	May 2, 2020	
Net loss	(\$334)	(\$62)	
Less: income tax benefit	(269)	(156)	
Add: interest expense	285	121	
Loss before interest and income tax expense	(318)	(97)	
Add: operating lease interest ¹	93	102	
Adjusted net operating (loss) profit	(225)	5	
Add (Less): estimated income tax benefit (expense)	100	(4)	
Adjusted net operating (loss) profit after tax	(\$125)	\$1	
Average total assets	\$9,637	\$9,811	
Less: average deferred property incentives in excess of ROU assets ²	(265)	(303)	
Less: average non-interest-bearing current liabilities	(3,095)	(3,324)	
Average invested capital	\$6,277	\$6,184	
Return on assets ³	(3.5 %)	(0.6 %)	
Adjusted ROIC ³	(2.0 %)	— %	

¹ We add back the operating lease interest to reflect how we manage our business. Operating lease interest is a component of operating lease cost recorded in occupancy costs.

² For leases with property incentives that exceed the ROU assets, we reclassify the amount from assets to other current liabilities and other liabilities and reduce average total assets, as this better reflects how we manage our business.

³ COVID-19 related charges during fiscal 2020 negatively impacted return on assets by approximately 180 basis points and Adjusted ROIC by approximately 130 basis points for the four quarters ended May 2, 2020.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

LIQUIDITY AND CAPITAL RESOURCES

We strive to maintain a level of liquidity sufficient to allow us to cover our seasonal cash needs and to maintain appropriate levels of short-term borrowings. Our ongoing working capital requirements are generally funded primarily through cash flows generated from operations. In addition, we have access to the commercial paper market and can draw on our revolving credit facilities for working capital, capital expenditures and general corporate purposes. We ended the first quarter of 2021 with \$377 in cash and cash equivalents and \$600 of additional liquidity available on our Revolver. In May 2021, subsequent to quarter end, we completely repaid \$200 on our Revolver. We believe that our operating cash flows are sufficient to meet our cash requirements for the next 12 months and beyond.

Over the long term, we manage our cash and capital structure to maximize shareholder return, maintain our financial position, manage refinancing risk and allow flexibility for strategic initiatives. We regularly assess our debt and leverage levels, capital expenditure requirements, debt service payments, dividend payouts, potential share repurchases and other future investments.

The following is a summary of our cash flows by activity:

	Quarter E	Quarter Ended	
	May 1, 2021	May 2, 2020	
Net cash used in operating activities	(\$364)	(\$778)	
Net cash used in investing activities	(110)	(126)	
Net cash provided by financing activities	167	1,417	

Operating Activities

Net cash used in operating activities improved \$414 for the quarter ended May 1, 2021, compared with the same period in 2020, primarily due to an improvement in net earnings and a decrease in performance-related payments.

Investing Activities

Net cash used in investing activities decreased \$16 for the quarter ended May 1, 2021, compared with the same period in 2020, primarily due to increased proceeds from the sale of assets.

Capital Expenditures

Our capital expenditures, net are summarized as follows:

	Quarter Ende	Quarter Ended	
	May 1, 2021	May 2, 2020	
Capital expenditures	\$126	\$131	
Less: deferred property incentives ¹	(6)	(8)	
Capital expenditures, net	\$120	\$123	
Capital expenditures % of net sales	4.3 %	64%	

¹Deferred property incentives are included in our cash provided by operations in our Consolidated Statements of Cash Flows in Item 1. We operationally view the property incentives we receive from our developers and vendors as an offset to our capital expenditures.

Financing Activities

Net cash provided by financing activities decreased \$1,250 for the quarter ended May 1, 2021, compared with the same period in 2020, primarily due to decreased proceeds on the Revolver and the retirement of the Secured Notes (see Note 3: Debt and Credit Facilities in Item 1).

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Free Cash Flow (Non-GAAP financial measure)

Free Cash Flow is one of our key liquidity measures and, when used in conjunction with GAAP measures, we believe it provides investors with a meaningful analysis of our ability to generate cash from our business.

Free Cash Flow is not a measure of financial performance under GAAP and should be considered in addition to, and not as a substitute for, operating cash flows or other financial measures prepared in accordance with GAAP. Our method of calculating non-GAAP financial measures may differ from other companies' methods and therefore may not be comparable to those used by other companies. The financial measure calculated under GAAP which is most directly comparable to Free Cash Flow is net cash used in operating activities. The following is a reconciliation of net cash used in operating activities to Free Cash Flow:

	Quarter	Quarter Ended	
	May 1, 2021	May 2, 2020	
Net cash used in operating activities	(\$364)	(\$778)	
Less: capital expenditures	(126)	(131)	
(Less) Add: change in cash book overdrafts	(17)	83	
Free Cash Flow	(\$507)	(\$826)	

Adjusted EBITDA (Non-GAAP financial measure)

Adjusted EBITDA is one of our key financial metrics to reflect our view of cash flow from net earnings. Adjusted EBITDA excludes significant items which are non-operating in nature in order to evaluate our core operating performance against prior periods. The financial measure calculated under GAAP which is most directly comparable to Adjusted EBITDA is net earnings.

Adjusted EBITDA is not a measure of financial performance under GAAP and should be considered in addition to, and not as a substitute for net earnings, overall change in cash or liquidity of the business as a whole. Our method of calculating a non-GAAP financial measure may differ from other companies' methods and therefore may not be comparable to those used by other companies. The following is a reconciliation of net loss to Adjusted EBITDA:

	Quarter Ended	
	May 1, 2021	May 2, 2020
Net loss	(\$166)	(\$521)
Less: income tax benefit	(56)	(326)
Add: interest expense, net	137	34
Loss before interest and income taxes	(85)	(813)
Add: depreciation and amortization expenses	162	176
Less: amortization of developer reimbursements	(20)	(19)
Add: asset impairments	_	117
Adjusted EBITDA	\$57	(\$539)

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Credit Capacity and Commitments

During the first quarter of 2021, we borrowed \$200 and amended our existing Revolver. As of May 1, 2021, we had \$200 outstanding under the facility. The Revolver contains customary representations, warranties, covenants and terms, including paying a variable rate of interest and a commitment fee based on our debt rating. The Revolver is available for working capital, capital expenditures and general corporate purposes. Provided that we obtain written consent from our lenders, we have the option to increase the Revolver by up to \$200, to a total of \$1,000, and two options to extend the Revolver by one year. For more information about our credit facilities, see Note 3: Debt and Credit Facilities in Item 1.

Impact of Credit Ratings

Changes in our credit ratings may impact our costs to borrow, whether our personal property secures our Revolver and the debt covenants we follow.

For our Revolver, the interest rate applicable to any borrowings we may enter into depends upon the type of borrowing incurred plus an applicable margin, which is determined based on our credit ratings. At the time of this report, our credit ratings and outlook were as follows:

	Credit Ratings	Outlook
Moody's	Baa3	Negative
Standard & Poor's	BB+	Stable
Fitch	BBB-	Negative

Should the ratings assigned to our long-term debt improve, the applicable margin associated with any borrowings under the Revolver may decrease, resulting in a lower borrowing cost under this facility. Conversely, should the ratings assigned to our long-term debt worsen, the applicable margin associated with any borrowings under the Revolver may increase, resulting in a higher borrowing cost under this facility.

Debt Covenants

As of May 1, 2021, we met all our covenants while our Leverage Ratio exceeded four. Under our current debt covenants, if our Leverage Ratio is greater than four or our unsecured debt is rated below BBB- with a stable outlook at Standard & Poor's or Baa3 with a stable outlook at Moody's, any outstanding borrowings under our Revolver will be secured by substantially all our personal property and we will be prevented from paying dividends and repurchasing shares. For more information about our debt covenants, see Note 3: Debt and Credit Facilities in Item 1.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

(Continued) (Amounts in millions except per share amounts)

Adjusted Debt to EBITDAR (Non-GAAP financial measure)

Adjusted Debt to EBITDAR is one of our key financial metrics and we believe that our debt levels are best analyzed using this measure, as it provides a reflection of our creditworthiness that could impact our credit rating and borrowing costs. This metric is calculated in accordance with our debt covenant and is a key component in assessing whether our revolving credit facility is secured or unsecured, as well as our ability to make dividend payments and share repurchases. Our goal is to manage debt levels to maintain an investment-grade credit rating while operating with an efficient capital structure.

Adjusted Debt to EBITDAR is not a measure of financial performance under GAAP and should be considered in addition to, and not as a substitute for, debt to net earnings, net earnings, debt or other GAAP financial measures. Our method of determining non-GAAP financial measures may differ from other companies' methods and therefore may not be comparable to those used by other companies.

The following is a reconciliation of debt to net loss to Adjusted Debt to EBITDAR:

	May 1, 2021
Debt	\$3,547
Add: estimated capitalized operating lease liability ¹	1,335
Adjusted Debt	\$4,882
	Four Quarters Ended May 1, 2021
Net loss	(\$334)
Less: income tax benefit	(269)
Add: interest expense, net	284
Add: asset impairments	20
Adjusted loss before interest and income taxes	(\$299)
Add: depreciation and amortization expenses	658
Add: rent expense, net ²	223
Add: other Revolver covenant adjustments ³	2
Adjusted EBITDAR	\$584

Adjusted Debt to EBITDAN	
¹ Based upon the estimated lease liability as of the end of the period, calculated as the trailing four quarters of rent expense multiplied by six, a method of estimating the debt we	
would record for our leases that are classified as operating if they had met the criteria for a capital lease or we had purchased the property and is calculated under the previous	
lease standard, consistent with our debt covenant calculation requirements. The estimated lease liability is not calculated in accordance with, nor an alternative for, GAAP and	

(10.6)

CRITICAL ACCOUNTING ESTIMATES

Debt to Net Loss

Adjusted Debt to FRITDAR

The preparation of our financial statements requires that we make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses, and disclosure of contingent assets and liabilities. We base our estimates on historical experience and other assumptions that we believe to be reasonable under the circumstances. Actual results may differ from these estimates. We believe that the estimates, assumptions and judgments involved in the accounting policies referred to in our 2020 Annual Report have the greatest potential effect on our financial statements, so we consider these to be our critical accounting policies and estimates. Our management has discussed the development and selection of these critical accounting estimates with the Audit & Finance Committee of our Board of Directors. There have been no material changes to our significant accounting policies or critical accounting estimates as described in our 2020 Annual Report.

should not be considered in isolation or as a substitution for our results reported under GAAP.

Rent expense, net of amortization of developer reimbursements, is added back for consistency with our debt covenant calculation requirements, and is calculated under the previous lease standard.

³ Other adjusting items to reconcile net loss to Adjusted EBITDAR as defined by our Revolver covenant include interest income and certain non-cash charges where relevant.

Item 3. Quantitative and Qualitative Disclosures About Market Risk.

We discussed our interest rate risk and foreign currency exchange risk in our 2020 Annual Report. There have been no material changes to these risks since that time.

Item 4. Controls and Procedures.

DISCLOSURE CONTROLS AND PROCEDURES

For the purposes of the Exchange Act, our Chief Executive Officer, Erik B. Nordstrom, serves as our principal executive officer and our Chief Financial Officer, Anne L. Bramman, is our principal financial officer.

Under the supervision and with the participation of management, including our principal executive officer and principal financial officer, we have performed an evaluation of the design and effectiveness of our disclosure controls and procedures as of the last day of the period covered by this report.

Based upon that evaluation, our principal executive officer and principal financial officer concluded that our disclosure controls and procedures were effective. Disclosure controls and procedures are defined by Rules 13a-15(e) and 15d-15(e) under the Exchange Act as controls and other procedures that are designed to ensure that information required to be disclosed in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified within the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed in the reports that we file or submit under the Exchange Act is accumulated and communicated to our management, including our principal executive officer and principal financial officer, to allow timely decisions regarding required disclosure.

CHANGES IN INTERNAL CONTROL OVER FINANCIAL REPORTING

There have been no changes in our internal control over financial reporting (as defined in Rules 13a-15(f) or 15d-15(f) of the Exchange Act) during our most recently completed fiscal quarter that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II — OTHER INFORMATION

Item 1. Legal Proceedings.

We are subject from time to time to various claims and lawsuits arising in the ordinary course of business, including lawsuits alleging violations of state and/or federal wage and hour and other employment laws, privacy and other consumer-based claims. Some of these lawsuits may include certified classes of litigants, or purport or may be determined to be class or collective actions and seek substantial damages or injunctive relief, or both, and some may remain unresolved for several years. We believe the recorded accruals in our Condensed Consolidated Financial Statements are adequate in light of the probable and estimable liabilities. As of the date of this report, we do not believe any currently identified claim, proceeding or litigation, either alone or in the aggregate, will have a material impact on our results of operations, financial position or cash flows. Since these matters are subject to inherent uncertainties, our view of them may change in the future.

Item 1A. Risk Factors.

There have been no material changes to the risk factors we discussed in our 2020 Annual Report.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

(c) SHARE REPURCHASES

(Dollar and share amounts in millions, except per share amounts)

In August 2018, our Board of Directors authorized a program to repurchase up to \$1,500 of our outstanding common stock, with no expiration date. As a result of uncertainties from COVID-19 impacts, we repurchased no shares of common stock during the first quarter of 2021 and we had \$707 remaining in share repurchase capacity as of May 1, 2021. The actual timing, price, manner and amounts of future share repurchases, if any, will be subject to the discretion of the Board of Directors, contractual commitments, market and economic conditions, and applicable SEC rules.

Item 6. Exhibits.

(a) The information required under this item is incorporated herein by reference or filed or furnished as part of this report at:

Page

Nordstrom, Inc. and Subsidiaries Exhibit Index

27

All other exhibits are omitted because they are not applicable, not required or because the information required has been given as part of this report.

26 of 28

NORDSTROM, INC. Exhibit Index

	Exhibit	Method of Filing
10.1	Second Amendment to Revolving Credit Agreement dated September 26, 2018	Filed herewith electronically
<u>10.2</u>	Nordstrom 401(k) Plan (2021 Restatement)	Filed herewith electronically
<u>10.3</u>	Nordstrom Deferred Compensation Plan (2019 Restatement)	Filed herewith electronically
<u>10.4</u>	Amendment 2021-1 to the Nordstrom Deferred Compensation Plan	Filed herewith electronically
31.1	<u>Certification of Chief Executive Officer required by Section 302(a) of the Sarbanes-Oxley Act of 2002</u>	Filed herewith electronically
31.2	<u>Certification of Chief Financial Officer required by Section 302(a) of the Sarbanes-Oxley Act of 2002</u>	Filed herewith electronically
<u>32.1</u>	Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002	Furnished herewith electronically
101.INS	Inline XBRL Instance Document	Filed herewith electronically
101.SCH	Inline XBRL Taxonomy Extension Schema Document	Filed herewith electronically
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document	Filed herewith electronically
101.LAB	Inline XBRL Taxonomy Extension Labels Linkbase Document	Filed herewith electronically
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document	Filed herewith electronically
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document	Filed herewith electronically
104	Cover Page Interactive Data File (Inline XBRL)	Filed herewith electronically

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

NORDSTROM, INC. (Registrant)

<u>/s/ Anne L. Bramman</u> Anne L. Bramman Chief Financial Officer (Principal Financial Officer)

Date: June 4, 2021

28 of 28

SECOND AMENDMENT TO REVOLVING CREDIT AGREEMENT

This SECOND AMENDMENT TO REVOLVING CREDIT AGREEMENT (this "<u>Amendment</u>") is entered into and effective as of April 30, 2021 (the "<u>Second Amendment Effective Date</u>") among Nordstrom, Inc., a Washington corporation (the "<u>Borrower</u>"), the Guarantors party hereto, the Lenders party hereto and Bank of America, N.A., as administrative agent (the "<u>Agent</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Credit Agreement (as defined below).

RECITALS

WHEREAS, the Borrower, the Lenders party thereto and the Agent are party to that certain Revolving Credit Agreement dated as of September 26, 2018 (as amended and modified from time to time, including as amended by this Amendment, the "Credit Agreement");

WHEREAS, the Borrower has requested an amendment to the Credit Agreement as described below; and

WHEREAS, the Required Lenders are willing to agree to such amendment, subject to the terms set forth herein as more fully set forth below.

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

AGREEMENT

- 1. Amendments to Credit Agreement.
 - (a) The definition of "Interest Expense" in Section 1.1 of the Credit Agreement is hereby amended to read as follows:
 - "Interest Expense" means the consolidated interest expense (including the amortization of debt discount and premium, the interest component under Capitalized Leases and the implied interest component under synthetic leases, tax retention operating leases, offbalance sheet loans or similar offbalance sheet financing products, but excluding the makewhole payment with respect to the 2020 Senior Notes and the unamortized bond premium with respect to the 2020 Senior Notes) of the Borrower and its Subsidiaries, as determined in accordance with GAAP.
- (b) The following definitions are hereby added to Section 1.1 of the Credit Agreement in the appropriate alphabetical order to read as follows:

"Credit Parties" has the meaning specified in Section 8.13.

"Rescindable Amount" has the meaning specified in Section 2.9(b).

(c) Section 2.9(b) of the Credit Agreement is hereby amended to read as follows:

(b) Unless the Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Agent for the account of the Lenders or any L/C Issuer hereunder that the Borrower will not make such payment, the Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders or the applicable L/C Issuers, as the case may be, the amount due.

With respect to any payment that the Agent makes for the account of the Lenders or any L/C Issuer hereunder as to which the Agent determines (which determination shall be conclusive absent manifest error) that any of the following applies (such payment referred to as the "Rescindable Amount"): (1) the Borrower has not in fact made such payment; (2) the Agent has made a payment in excess of the amount so paid by the Borrower (whether or not then owed); or (3) the Agent has for any reason otherwise erroneously made such payment; then each of the Lenders or the applicable L/C Issuers, as the case may be, severally agrees to repay to the Agent forthwith on demand the Rescindable Amount so distributed to such Lender or such L/C Issuer, in Same Day Funds with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Agent, at the Overnight Rate.

A notice of the Agent to any Lender or the Borrower with respect to any amount owing under this clause (b) shall be conclusive, absent manifest error.

(d) Section 6.2 of the Credit Agreement is hereby amended to read as follows:

6.2 Restricted Payments.

The Borrower shall not, and shall not permit any Subsidiary to, declare, pay or make, or agree to declare, pay or make, any Restricted Payment, except (a) Restricted Payments by any Subsidiary to the Borrower and any other Person that owns capital stock or other equity interests in such Subsidiary, ratably according to their respective holdings of the type of capital stock or other equity interests in respect of which such Restricted Payment is being made, (b) Restricted Payments (other than purchases or other acquisition for value of any Capital Stock of the Borrower or any Subsidiary) so long as no Default or Event of Default then exists or would result therefrom (assuming for this purpose that compliance with Section 6.3 is being measured as of the end of the immediately preceding Fiscal Quarter giving pro forma effect to the Restricted Payment) and/or (c) purchases or other acquisitions for value of any Capital Stock of the Borrower or any Subsidiary. Notwithstanding anything to the contrary contained herein, during any Collateral Period, the Borrower shall not, and shall not permit any Subsidiary to, declare, pay or make, or agree to declare, pay or make, any dividends or repurchases of capital stock or other equity interests; provided, that this provision shall not prohibit (i) any Subsidiary from making Restricted Payments to Persons that own Capital Stock in such Subsidiary, ratably according to their respective holdings of the type of Capital Stock in respect of which such Restricted Payment is being made or (ii) the Borrower and its Subsidiaries from making Restricted Payments so long as (A) no Default or Event of Default then exists or would result therefrom, (B) after

giving effect to such Restricted Payment on a pro forma basis, the Leverage Ratio, computed as of the most recent Fiscal Quarter end of the Borrower for which financial statements were required to be delivered pursuant to Section 5.1(a) or 5.1(b), shall be less than 3.75 to 1.0, (C) after giving effect to such Restricted Payment on a pro forma basis, the Borrower and its Subsidiaries shall have Liquidity, as of the date of such Restricted Payment, of at least \$600,000,000 and (D) the aggregate amount of all Restricted Payments made in any Fiscal Quarter shall not exceed the aggregate amount of Restricted Payments made in the corresponding Fiscal Quarter of 2019.

(e) Article VIII of the Credit Agreement is hereby amended to add a new Section 8.13 to read as follows:

8.13 Recovery of Erroneous Payments.

Without limitation of any other provision in this Agreement, if at any time the Agent makes a payment hereunder in error to any Lender or any L/C Issuer (the "Credit Party"), whether or not in respect of an Obligation due and owing by the Borrower at such time, where such payment is a Rescindable Amount, then in any such event, each Credit Party receiving a Rescindable Amount severally agrees to repay to the Agent forthwith on demand the Rescindable Amount received by such Credit Party in Same Day Funds in the currency so received, with interest thereon, for each day from and including the date such Rescindable Amount is received by it to but excluding the date of payment to the Agent, at the Overnight Rate. Each Credit Party irrevocably waives any and all defenses, including any "discharge for value" (under which a creditor might otherwise claim a right to retain funds mistakenly paid by a third party in respect of a debt owed by another) or similar defense to its obligation to return any Rescindable Amount. The Agent shall inform each Credit Party promptly upon determining that any payment made to such Credit Party comprised, in whole or in part, a Rescindable Amount.

(e) Section 9.18 of the Credit Agreement is hereby amended to read as follows:

9.18 Electronic Execution; Electronic Records.

This Agreement and any document, amendment, approval, consent, information, notice, certificate, request, statement, disclosure or authorization related to this Agreement (each a "Communication"), including Communications required to be in writing, may be in the form of an Electronic Record and may be executed using Electronic Signatures. Each of the Loan Parties and each of the Agent and each Lender agrees that any Electronic Signature on or associated with any Communication shall be valid and binding on such Person to the same extent as a manual, original signature, and that any Communication entered into by Electronic Signature, will constitute the legal, valid and binding obligation of such Person enforceable against such Person in accordance with the terms thereof to the same extent as if a manually executed original signature was delivered. Any Communication may be executed in as many counterparts as necessary or convenient, including both paper and electronic counterparts, but all such counterparts are one and the same Communication. For the avoidance of doubt, the

authorization under this paragraph may include, without limitation, use or acceptance of a manually signed paper Communication which has been converted into electronic form (such as scanned into PDF format), or an electronically signed Communication converted into another format, for transmission, delivery and/or retention. The Agent and each of the Lenders may, at its option, create one or more copies of any Communication in the form of an imaged Electronic Record ("Electronic Copy"), which shall be deemed created in the ordinary course of the such Person's business, and destroy the original paper document. All Communications in the form of an Electronic Record, including an Electronic Copy, shall be considered an original for all purposes, and shall have the same legal effect, validity and enforceability as a paper record. Notwithstanding anything contained herein to the contrary, neither the Agent nor any Lender is under any obligation to accept an Electronic Signature in any form or in any format unless expressly agreed to by such Person pursuant to procedures approved by it; provided, further, without limiting the foregoing, (a) to the extent the Agent, the L/C Issuer and/or the Swing Line Lender has agreed to accept such Electronic Signature, the Agent and each of the Lenders shall be entitled to rely on any such Electronic Signature purportedly given by or on behalf of any Loan Party, the Agent and/or any Lender without further verification and (b) upon the request of the Agent or any Lender, any Electronic Signature shall be promptly followed by such manually executed counterpart. For purposes hereof, "Electronic Record" and "Electronic Signature" shall have the meanings assigned to them, respectively, by 15 USC §7006, as it may be amended from time to time.

- 2. <u>Effectiveness; Conditions Precedent</u>. This Amendment shall be effective upon satisfaction of the following conditions:
- (a) Receipt by the Agent of copies of this Amendment duly executed by the Borrower, the Guarantors, the Agent and the Required Lenders.
- (b) Payment by the Borrower of a non-refundable amendment fee, for the account of each Lender that executes and delivers this Amendment, in the amount of 0.02% of such Lender's Revolving Commitment.
- (c) The Borrower shall have paid all reasonable fees, charges and disbursements of counsel to the Agent (directly to such counsel if requested by the Agent) to the extent invoiced prior to or on the date hereof.
- 3. <u>Ratification of Credit Agreement</u>. On and after the Second Amendment Effective Date, the term "Credit Agreement" as used in each of the Loan Documents shall hereafter mean the Credit Agreement as amended and modified by this Amendment. Except as herein specifically agreed, the Credit Agreement, as amended by this Amendment, is hereby ratified and confirmed and shall remain in full force and effect according to its terms. Each of the Loan Parties acknowledges and consents to the modifications set forth herein and agree that this Amendment does not impair, reduce or limit any of its obligations under the Loan Documents (including, without limitation, the indemnity obligations set forth therein) and that, after the date hereof, this Amendment shall constitute a Loan Document.
 - 4. <u>Authority/Enforceability</u>. Each of the Loan Parties represents and warrants as follows:

- (a) It has taken all necessary action to authorize the execution, delivery and performance of this Amendment.
- (b) This Amendment has been duly executed and delivered by such Person and constitutes such Person's legal, valid and binding obligations, enforceable in accordance with its terms, except as such enforceability may be subject to (i) bankruptcy, insolvency, reorganization, fraudulent conveyance or transfer, moratorium or similar laws affecting creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding at law or in equity).
- (c) No consent, approval, authorization or order of, or filing, registration or qualification with, any court or governmental authority or third party is required in connection with the execution, delivery or performance by such Person of this Amendment.
- (d) The execution and delivery of this Amendment does not (i) violate, contravene or conflict with any provision of its organization or formation documents or (ii) materially violate, contravene or conflict with any law, regulation, order, writ, judgment, injunction, decree or permit applicable to it or any of its Subsidiaries.
- 5. Representations and Warranties of the Loan Parties. The Loan Parties represent and warrant to the Agent and each Lender that (a) the representations and warranties of the Loan Parties set forth (i) in Article VI of the Credit Agreement and (ii) in the other Loan Documents are true and correct in all material respects as of the date hereof (except to the extent a representation and warranty specifically refers to an earlier date and then as of such earlier date); provided that, in each case, such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof and (b) after giving effect to this Amendment, no event has occurred and is continuing which constitutes a Default or an Event of Default.
- 6. <u>Counterparts/Electronic delivery</u>. This Amendment may be executed in any number of counterparts, each of which when so executed and delivered shall be an original, but all of which shall constitute one and the same instrument. Delivery of executed counterparts of this Amendment by telecopy or electronic transmission of a "PDF" copy shall be effective as an original and shall constitute a representation that an original shall be delivered promptly upon request.
- 7. <u>GOVERNING LAW</u>. THIS AMENDMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK. The jurisdiction, service of process, waiver of venue and waiver of jury trial provisions of Sections 1.2(d) and 9.12 of the Credit Agreement are hereby incorporated by reference, mutatis mutandis.

[remainder of page intentionally left blank]

BORROWER: NORDSTROM, INC.	
By: Name: Title:	
NORDSTROM SECOND AMENDMENT TO REVOLVING CREDIT FAC	, IN ILIT

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the date first above written.

ARANTORS:	NIHC, INC.	
		By: Name: Title:
		NORDSTROM CARD SERVICES, INC.
		By: Name: Title:
		TRUNK CLUB, LLC
		(formerly known as Trunk Club, Inc.) By: Name:
		Title:
		NORDSTROM CANADA RETAIL, INC.
		By: Name: Title:

GUARANTORS:

NORDSTROM, INC. SECOND AMENDMENT TO REVOLVING CREDIT FACILITY

GENT:	BANK OF AMERICA, N.A.,	
	as Agent	
	By:	
	Name:	
	Title:	

AGENT:

NORDSTROM, INC. SECOND AMENDMENT TO REVOLVING CREDIT FACILITY

LENDERS:	BANK OF AMERICA	, N.A.,
	· ·	as a Lender
		_
		Ву:
		Name: Title:
		riue:
	,	WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender
		Ву:
]	Name:
	,	Title:
		U.S. BANK NATIONAL ASSOCIATION,
		as a Lender
	,	D ₁₁₁
		By: Name:
		Title:
		THE.
]	Fifth Third Bank, NATIONAL
		ASSOCIATION,
	;	as a Lender
		D.
		By:
		Name: Title:
		Tiue.
	1	MUFG Bank, Ltd.,
		as a Lender
]	Ву:
		Name:
	,	Title:
	,	The Rank of Neva Scotia
		The Bank of Nova Scotia, as a Lender
	•	ao a Denaer

By:___ Name:

Title:
The Toronto-Dominion Bank, New York Branch, as a Lender
By:
Name:
Title:
Goldman Sachs Bank USA, as a Lender
By:
Name:
Title:
JPMorgan Chase Bank, N.A.,
as a Lender
By:
Name:
Title:
Morgan Stanley Bank, N.A.,
as a Lender
By:
Name:
Title:
KeyBank National Association , as a Lender
By:
Name:
Title:
The Bank of New York Mellon, as a Lender
By:

Name:

Bank of Hawaii, as a Lender
By: Name: Title:

Title:

NORDSTROM, INC. SECOND AMENDMENT TO REVOLVING CREDIT FACILITY

NORDSTROM 401(k) PLAN 2021 RESTATEMENT

Includes All Amendments Approved by the Company since the 2019 Restatement, including:

Amendment 2020-1

Jackson Lewis PC
Seattle, WA 98101

NORDSTROM 401(k) PLAN

2021 Restatement

Table of Contents

		Pag
Section	1. NAME AND PURPOSE OF PLAN	<u>1</u>
1.1	Name and Purpose of Plan.	<u>1</u>
1.2	Effective Date.	<u>1</u>
Section	2. DEFINITIONS	<u>2</u>
2.1	Administrator	<u>2</u>
2.2	Anniversary Date	<u>2</u>
2.3	Break in Vesting Service	<u>2</u>
2.4	Code	<u>2</u>
2.5	Company	<u>2</u>
2.6	Compensation	<u>2</u>
2.7	Disability	<u>3</u>
2.8	Eligible Employee	<u>3</u>
2.9	Employee	<u>4</u>
2.10	Employer and Employers	<u>4</u>
2.11	Employment Commencement Date	<u>4</u>
2.12	ERISA	<u>4</u>
2.13	Highly Compensated Employee and Non-Highly Compensated Employee	<u>4</u>
2.14	Hour of Service	<u>5</u>
2.15	Leased Employee	<u>6</u>
2.16	Named Fiduciary	<u>6</u>
2.17	Participant	<u>6</u>
2.18	Payroll Year	<u>6</u>
2.19	Permanent Break in Eligibility Service	<u>6</u>
2.20	Plan	<u>6</u>
2.21	Plan Year	<u>6</u>
2.22	Retirement	<u>6</u>
2.23	Retirement Committee	<u>6</u>
2.24	Severance from Employment Date	<u>6</u>
2.25	Taxable Year	<u>7</u>
2.26	Trustee	<u>7</u>
2.27	Trust Fund	<u>7</u>
2.28	Valuation Date	<u>7</u>
2.29	Year of Service	<u>7</u>
2.30	Non-Adopting Employer	<u>Z</u>
Section	3. ADMINISTRATION OF PLAN	<u>7</u>

3.1	Plan Administrator	<u>7</u>
3.2	Enumerated Administrative Powers	<u>8</u>
3.3	Administrative Records	<u>9</u>
3.4	Employer Records	<u>9</u>
3.5	Duties of Participant	<u>9</u>
3.6	Administrator Expenses	<u>9</u>
3.7	Individuals Indemnified	<u>9</u>
3.8	Administrator Continues Until Trust Exhausted	<u>9</u>
3.9	Plan Expenses.	<u>9</u>
Section	n 4. ELIGIBILITY OF EMPLOYEES TO PARTICIPATE	<u>10</u>
4.1	Initial Participation	<u>10</u>
4.2	Break in Eligibility Service — Reemployment After Break	<u>10</u>
4.3	Information from Employer	<u>11</u>
Section	n 5. CONTRIBUTIONS	<u>11</u>
5.1	Employer Profit Sharing Contribution	<u>11</u>
5.2	Elective Deferral Contributions	<u>11</u>
5.3	Designated Roth Contributions	<u>16</u>
5.4	Employer Matching Contributions	<u>17</u>
5.5	Time of Payment of Contributions	<u>17</u>
5.6	Plan Qualification	<u>18</u>
5.7	Return of Mistaken and Nondeductible Contributions	<u>18</u>
5.8	Military Leave Obligations	<u>18</u>
5.9	Rollover Contributions	<u>19</u>
5.10	Qualified Non-Elective Contributions	<u>20</u>
Section	n 6. ALLOCATION OF CONTRIBUTIONS AND VALUATION OF TRUST FUND	<u>22</u>
6.1	Allocation of Contributions and Forfeitures.	<u>22</u>
6.2	Valuation and Allocation of Trust Fund	<u>22</u>
6.3	Investment of Contributions	<u>22</u>
6.4	Allocation Does Not Vest Rights	<u>22</u>
6.5	Forfeiture Suspense Account	<u>23</u>
6.6	Limitation on Annual Additions	<u>23</u>
6.7	Allocation of Excess Additions	<u>24</u>
6.8	Contribution Limits for Highly Compensated Employees.	<u>25</u>
6.9	Correcting Excess Contributions.	<u>27</u>
Section	n 7. [INTENTIONALLY OMITTED]	<u>28</u>
Section	n 8. VESTING OF BENEFITS	<u>28</u>
8.1	Vesting	<u>28</u>
8.2	Forfeiture of Benefits for Certain Causes	<u>30</u>
8.3	Forfeiture of Nonvested Portion of Account	<u>30</u>
8.4	Reinstatement of Nonvested Portion of Account	<u>31</u>

8.5	Service After Severance from Employment	<u>31</u>
8.6	Forfeiture Reallocation	<u>32</u>
8.7	Maternity/Paternity/Family Absences	<u>32</u>
Section	9. ELIGIBILITY TO RECEIVE BENEFITS	<u>33</u>
9.1	Normal Retirement Benefits	<u>33</u>
9.2	Disability Benefits	<u>33</u>
9.3	Death Benefits	<u>33</u>
9.4	Benefits on Severance from Employment	<u>34</u>
9.5	Accelerated Benefit Option	<u>35</u>
9.6	In-Service Withdrawals	<u>35</u>
9.7	Hardship Withdrawals	<u>35</u>
9.8	Restriction on Distributions of Elective Deferrals	<u>37</u>
9.9	Military Service Distributions	<u>37</u>
Section	10. METHOD OF PAYMENT OF BENEFITS	<u>37</u>
10.1	Distribution of Benefits	<u>37</u>
10.2	Valuation of Account	<u>38</u>
10.3	Time of Distribution	<u>39</u>
10.4	Form of Payment	<u>39</u>
10.5	Qualified Domestic Relations Orders	<u>40</u>
10.6	Partial Withdrawals	<u>40</u>
10.7	Rollovers	<u>40</u>
10.8	Administration of Unclaimed Benefits	<u>41</u>
Section	11. MINIMUM DISTRIBUTION REQUIREMENTS	<u>42</u>
11.1	General Rules	<u>42</u>
11.2	Time and Manner of Distribution	<u>42</u>
11.3	Required Minimum Distributions During Participant's Lifetime	<u>43</u>
11.4	Required Minimum Distributions After Participant's Death	<u>44</u>
11.5	Definitions	<u>45</u>
11.6	Source of Funds for Required Minimum Distributions	<u>47</u>
Section	12. TOP HEAVY PLANS	<u>47</u>
12.	1 Effective Date	<u>47</u>
12.2	2 Effect of Top Heavy Plan Status	<u>47</u>
12.3	1 0	<u>47</u>
12.4	4 Minimum Employer Contributions to Top Heavy Plans	<u>49</u>
Section	13. PARTIES RESPONSIBLE FOR IMPLEMENTING THE PLA	N <u>51</u>
13.1	Plan Sponsor	<u>51</u>
13.2	Plan Fiduciaries	<u>54</u>
13.3	Plan Committees	<u>55</u>
13.4	S	<u>56</u>
	14. SPENDTHRIFT PROVISIONS	<u>56</u>
14.1	Prohibition Against Assignment	<u>56</u>

14.2	Effect of Assignment	<u>57</u>
14.3	QDRO Exception	<u>57</u>
Section	15. AMENDMENT AND TERMINATION OF PLAN	<u>57</u>
15.1	Future of the Plan	<u>57</u>
15.2	Company Right to Amend the Plan	<u>57</u>
15.3	Company Right To Terminate the Plan	<u>58</u>
15.4	Partial Termination	<u>59</u>
15.5	Procedure for Plan Amendment or Termination	<u>59</u>
Section	16. CLAIMS AND REVIEW PROCEDURE	<u>60</u>
16.1	Claims for Benefits and Inquiries	<u>60</u>
16.2	Denial of Claims	<u>60</u>
16.3	Review of Denied Claims	<u>60</u>
16.4	Decision on Review	<u>61</u>
16.5	Rules and Procedures on Review	<u>61</u>
16.6	Exhaustion of Remedies	<u>61</u>
16.7	Time for Initiating Legal Action	<u>62</u>
Section	17. MISCELLANEOUS PROVISIONS	<u>62</u>
17.1	No Right of Continued Employment	<u>62</u>
17.2	Discretion	<u>62</u>
17.3	Separability	<u>62</u>
17.4	Participant and Others Bound by Plan	<u>62</u>
17.5	Applicable Law	<u>62</u>
17.6	Text Controls	<u>62</u>
17.7	Expenses	<u>62</u>
17.8	Plan Document is Controlling	<u>62</u>
17.9	Rules of Construction	<u>63</u>
Section	18. LOANS TO PARTICIPANTS	<u>63</u>
18.1	Loans to Participants	<u>63</u>

Appendices:

- A. Participant Investment Direction Appendix
 B. Store Closure Provisions Appendix
 C. CARES Act Provisions Appendix

- D. Acquisitions Service History Appendix

NORDSTROM 401(k) PLAN

2021 Restatement

WITNESSETH:

WHEREAS, certain of the Employers, and their predecessors, entered into a Profit Sharing Plan ("Plan") and Trust Agreement on December 31, 1952; and

WHEREAS, the parties in 1988 amended and restated the Plan and Trust Agreement to adopt separate provisions regarding Code § 401(k) in a document referred to as the Nordstrom Employee Deferral Retirement Plan, which was subsequently renamed the Nordstrom 401(k) Plan ("401(k) Plan"); and

WHEREAS, the parties in 2003 amended and restated the Plan and Trust Agreement to incorporate the terms of the 401(k) Plan into the Plan to avoid redundancies in maintaining separate plan documents; and

WHEREAS, the parties have amended and restated the Plan in 1993, 1995, 1998, 2000, 2003, 2004, 2008; 2014 and 2019; and

WHEREAS, in 2019, the Company most recently amended and restated the Plan into the 2019 Restatement to incorporate amendments adopted since the 2014 Restatement; and

WHEREAS, the 2019 Restatement of the Plan has subsequently been amended; and

WHEREAS, the Company desires to amend and restate this Plan to incorporate amendments adopted since the 2019 restatement, certain substantive provisions reflecting changes in Plan design, changes in legal rules governing the Plan, and other administrative modifications reflecting changes in the Plan's operation; and

NOW, THEREFORE, the Company does hereby adopt the Nordstrom 401(k) Plan as amended and restated herein in the form of this 2021 Restatement.

Section 1. NAME AND PURPOSE OF PLAN

- 1.1 Name and Purpose of Plan. The Plan shall be known as the Nordstrom 401(k) Plan.
- 1.2 Effective Date.

1.2.1 <u>2021 Restatement</u>. Unless another effective date is specified herein or in a prior Plan amendment, this 2021 Restatement is effective January 1, 2021, and shall govern rights with respect to employment with the Employers on and after January 1, 2021. Rights and benefits with respect to employment prior to 2021 shall be governed by the prior version of the Plan as amended and in effect at the time of reference, unless otherwise specifically provided herein.

Page 1 of 65 NORDSTROM 401(k) PLAN

1.2.2 <u>Retroactive Effective Date</u>. Provisions herein that are needed to comply with any legislation and regulations shall be effective retroactively as of the earliest compliance date required by law. Unless otherwise indicated, such retroactivity shall not change the effective date or amount of any Employer contribution made under Section 5 or other benefit provisions implemented for reasons other than compliance with the law and regulations.

Section 2. DEFINITIONS

When used herein, the following words shall have the following meanings unless the context clearly indicates otherwise:

- 2.1 Administrator means the Company.
- 2.2 <u>Anniversary Date</u> means December 31st of each year.
- 2.3 <u>Break in Vesting Service</u> means a Payroll Year in which the Participant has failed to complete more than five hundred (500) Hours of Service.
 - 2.4 Code means the Internal Revenue Code of 1986, as amended.
 - 2.5 Company means Nordstrom, Inc.
- 2.6 <u>Compensation</u> means that portion of compensation received from an Employer that is described in this Section 2.6 and that appears on an Employee's IRS Form W-2 as taxable wages for the Payroll Year ending with any Plan Year. Compensation includes all monies paid to an Employee for services rendered in the form of salary and wages, including bonuses and commissions, and those amounts which are part of the Employee's basic compensation scheme and paid regularly in accordance with any agreed formula.
- 2.6.1 <u>Items Specifically Included</u>. Except as specifically provided herein, Compensation shall include Employer contributions made pursuant to a salary reduction agreement which are not includible in the gross income of an Employee under Code §§ 125 or 402(g)(3). For Plan Years commencing on and after January 1, 1998, Compensation shall also include Employees' pre-tax contributions for qualified transportation fringe benefits under Code § 132(f)(4).
- 2.6.2 <u>Items Specifically Excluded</u>. Except as specifically provided herein, the term "Compensation" shall not include any amounts paid outside of the regularly occurring payment for services (as described above) including, but not limited to, any reimbursements or other expense allowances, employee awards, taxable fringe benefits (and non-taxable fringe benefits not described in 2.6.1), moving expenses, severance pay, disability pay under the employer's separately written disability program, amounts received as stock or under any stock-based compensation program (such as stock awards, option gains or performance share units) and their equivalent cash value and other deferred compensation and welfare benefits.
- 2.6.3 <u>Timing of Payment</u>. Amounts paid after severance of employment will not be included in Compensation, provided that such amounts shall be included in Compensation

Page 2 of 65 NORDSTROM 401(k) PLAN

if the following conditions are met: (i) the payment is made by the later of: 2 ½ months after severance from employment or the last day of the Limitation Year (the "Limitation Year" is the same as the Plan Year) in which the severance occurs; and (ii) the amounts would have been included in compensation for purposes of Code § 415 if they were paid prior to the employee's severance from employment.

- 2.6.4 <u>Yearly Maximum</u>. The annual Compensation of each Participant to be taken into account under the Plan for any year shall not exceed the maximum compensation limit in effect under Code \S 401(a)(17) as adjusted by the Secretary of the Treasury at the same time and in the same manner as under Code \S 415(d).
- 2.6.5 <u>Compensation for Testing Purposes</u>. For purposes of the nondiscrimination tests under Code §§ 401(a) (4), 401(k) and (m), the Administrator may use any definition of compensation permitted by Code § 414(s). Also, if an Employee is a common-law employee of two or more corporations (or other business entities) that must be treated as a single employer under Code §§ 414(b), (c), (m), or (o), the term Compensation for testing purposes includes compensation from all such corporations and business entities, regardless of whether the corporation or other business entity has a qualified plan.
- 2.7 <u>Disability</u> means that the Participant is determined to be disabled either: (i) under the Company's long-term disability plan (if applicable); or (ii) by the Social Security Administration.
 - 2.8 <u>Eligible Employee</u> means each Employee of an Employer except the following:
- 2.8.1 An Employee covered by a collective bargaining agreement that does not provide for participation in the Plan.
 - 2.8.2 A Leased Employee treated as an employee for pension purposes solely because of Code § 414(n).
- 2.8.3 An individual classified by the Employer as either an independent contractor or employee of a nonaffiliated entity rather than as a common law employee of the Employer, regardless of whether such individual is later determined by a court or a governmental administrative agency to be a common law employee of an Employer.
 - 2.8.4 Non-resident aliens who receive no earned income from sources within the United States.
- 2.8.5 An Employee who, prior to his or her earliest participation date under Section 4.1, makes a one-time irrevocable election not to participate in the Plan, in accordance with procedures established by the Administrator. An Employee is not eligible to receive anything of value from any Employer, from the Administrator, or from any other person associated with the Employer or the Plan in consideration of the Employee's election not to participate in the Plan.

Page 3 of 65 NORDSTROM 401(k) PLAN

2.8.6 An individual who is not paid through Nordstrom payroll and who does not receive a Form W-2 from Nordstrom, regardless of whether such individual is later determined by a court or a governmental administrative agency to be a common law employee of an Employer.

Notwithstanding the above, Sections 2.8.2, 2.8.3, 2.8.5, and 2.8.6 are not intended to exclude such individuals from consideration for the purposes of coverage testing under Code \S 410(b), and, to the extent required, non-discrimination testing under Code \S 401(a), 401(k) and 401(m) even though they are not eligible to participate in the Plan.

- 2.9 <u>Employee</u> means, for purposes of this Plan any person employed as a common-law employee by an Employer or by any other employer required to be aggregated with an Employer under Code §§ 414(b), (c), (m) or (o).
- 2.10 <u>Employer and Employers</u> mean the Company and any other entity required to be aggregated with an Employer under Code §§ 414(b), (c), (m) or (o), provided the Company has authorized and such entity has specifically acted to adopt this Plan.
- 2.11 Employment Commencement Date means the first day on which an Employee performs an Hour of Service for the Employer. In addition, employees of companies acquired by the Company and who become Eligible Employees upon the closing of the acquisition shall be granted credit for Hours of Service performed for the acquired company unless the Company's Chief Human Resources Officer determines that such service credit should not be granted. Any actions taken pursuant to this authority will be documented in Appendix D to the Plan.
 - 2.12 ERISA means the Employee Retirement Income Security Act of 1974, as amended.
 - 2.13 <u>Highly Compensated Employee and Non-Highly Compensated Employee.</u>
- 2.13.1 <u>Highly Compensated Employee</u>. "Highly Compensated Employee" is defined in Code § 414(q) and related Treasury Regulations. In determining which Employees are Highly Compensated Employees, the following shall apply:
- (a) Subject to (b) and (c) below, Highly Compensated Employees for a Plan Year are persons who perform services for an Employer during the Plan Year or the preceding Plan Year and are one or more of the following:
- (1) An owner of greater than 5 percent of an Employer (a "5-percent owner") during either the current or the preceding Plan Year. For this purpose, a 5-percent owner is defined as any person who owns (or is considered as owning by applying the constructive ownership rules of Code § 318) more than 5 percent of the outstanding stock of the corporation or stock possessing more than 5 percent of the total combined voting power of all stock of the corporation.
- (2) An Employee receiving Compensation from the Employer over \$80,000 (as adjusted under Code § 415(d)) for the preceding Plan Year who is among the

Page 4 of 65 NORDSTROM 401(k) PLAN

highest paid twenty percent (20%) of Employees of the Employer during the preceding Plan Year, aggregating Employees of all Employers and excluding Employees to the extent provided by applicable Regulations.

- (b) Former Employees shall be taken into account in accordance with applicable Regulations. A former Employee shall be treated as a Highly Compensated Employee if:
 - (1) such Employee was a Highly Compensated Employee when such employee severed
- (2) such employee was a Highly Compensated Employee at any time after attaining age fifty-five (55).
 - (c) "Compensation" for purposes of this Section 2.13 shall mean Compensation under Section 2.6.5.
- 2.13.2 <u>Non-Highly Compensated Employee</u>. "Non-Highly Compensated Employee" means any Employee who is not a Highly Compensated Employee.

2.14 Hour of Service means:

employment; or

- 2.14.1 <u>Paid for Work</u>. Each hour for which an Employee is paid, or entitled to payment, for the performance of duties for the Employer during the applicable computation period.
- 2.14.2 <u>Paid Nonwork Time</u>. Each hour for which an Employee is paid, or entitled to payment, by the Employer on account of a period of time during which no duties are performed (irrespective of whether the Employee's employment has severed) due to vacation, holiday, illness, incapacity (including disability), layoff, jury duty, military duty or leave of absence. Notwithstanding the preceding sentence,
- (a) No more than five hundred one (501) Hours of Service are to be credited under this paragraph to an Employee on account of any single continuous period during which the Employee performs no duties (whether or not such period occurs in a single computation period);
- (b) An hour for which an Employee is directly or indirectly paid, or entitled to payment, on account of a period during which no duties are performed, is not to be credited to the Employee if such payment is made or due under a plan maintained solely for the purpose of complying with applicable workers' compensation, or unemployment compensation or disability insurance laws; and
- (c) Hours of Service are not to be credited for a payment which solely reimburses an Employee for medical or medically related expenses incurred by the Employee.

Page 5 of 65 NORDSTROM 401(k) PLAN

For purposes of this paragraph, a payment shall be deemed to be made by or due from the Employer regardless of whether such payment is made by or due from the Employer directly, or indirectly through, among others, a trust fund, or insurer, to which the Employer contributes or pays premiums and regardless of whether contributions made or due to the trust fund, insurer or other entity are for the benefit of particular Employees or are on behalf of a group of Employees in the aggregate.

- 2.14.3 <u>Back Pay</u>. Each hour for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the Employer. The same Hours of Service shall not be credited both under Section 2.14.1 or 2.14.2, as the case may be, and under this Section 2.14.3
- 2.14.4 <u>Determination Rules</u>. The determination of Hours of Service for reasons other than the performance of duties, and the crediting of Hours of Service to computation periods, shall be in accordance with Department of Labor regulations 29 CFR § 2530.200b2(b) and (c), which are incorporated by reference.
- 2.15 <u>Leased Employee</u> means any person (other than an Employee of an Employer) who pursuant to an agreement between an Employer and any other person ("leasing organization") has performed services for the Employer (or for the Employer and related persons determined in accordance with Code § 414(n)(6)) on a substantially full-time basis for a period of at least one (1) year, and such services are performed under the primary direction and control of the Employer. The determination of who is a Leased Employee shall be made in accordance with any rules and regulations issued by the Secretary of the Treasury or his delegate.
- 2.16 <u>Named Fiduciary</u> means the appropriate party, parties or entities appointed or delegated such named fiduciary functions pursuant to Section 3 and Section 13.
 - 2.17 Participant means a current or former Eligible Employee who has an account in the Plan.
- 2.18 <u>Payroll Year</u> means the period of fifty-two (52) consecutive weeks for which an Employee's IRS Form W-2 compensation is calculated.
- 2.19 <u>Permanent Break in Eligibility Service</u> means the sixty (60) month period during which an Employee has no Hours of Service, as measured from the date of the Employee's most recent Severance from Employment Date.
 - 2.20 Plan means the Nordstrom 401(k) Plan set forth in this document and all subsequent amendments thereto.
 - 2.21 Plan Year means the 12-month period commencing on January 1 and ending on December 31.
- 2.22 <u>Retirement</u> means a Participant's severance from employment after the Normal Retirement Date as defined in Section 9.1.

Page 6 of 65 NORDSTROM 401(k) PLAN

- 2.23 <u>Retirement Committee</u> means the Nordstrom Retirement Committee, established by the Board of Directors of the Company under Section 13.1.2(e), and charged with those powers and duties under Section 13.1.5.
- 2.24 <u>Severance from Employment Date</u> means the earlier of (a) the date an Employee quits, retires, is discharged, or dies, whichever occurs first, or (b) the first anniversary of the first date that the Employee is continually absent from work for any other reason (except where such absence is attributable to a Company-approved leave of absence).
- 2.25 <u>Taxable Year</u> means the twelve (12) month period adopted by the Company for its tax purposes. Currently, the Company's Taxable Year ends on the Saturday closest to each January 31.
- 2.26 <u>Trustee</u> means the person or persons holding the assets of the Plan pursuant to the terms of one (1) or more Trust Agreements entered into by the Employer.
 - 2.27 Trust Fund means those funds and assets of the Plan held by the Trustee.
 - 2.28 Valuation Date means any day that the New York Stock Exchange is open for business and trading.
- 2.29 <u>Year of Service</u> means a Payroll Year in which an Employee is credited with one thousand (1,000) or more Hours of Service. Unless otherwise specified in this Plan, Eligible Employees who worked for any member of the Controlled Group (as defined in Code § 414(b)) prior to that member being acquired by the Company, and former employees of companies acquired by the Company shall be credited with Years of Service under this Plan with respect to years of service with the other member of the Controlled Group based on the actual hours method or, if actual hours are unavailable, the elapsed time method.
 - 2.30 Non-Adopting Employer means any of the following:
 - (a) Nordstrom Retail Canada, Inc.;
 - (b) Nordstrom Puerto Rico LLC; and
- (c) any other entity that is required to be aggregated with an Employer under Code §§ 414(b), (c), (m) or (o) but is not an Employer because such entity has not adopted this Plan.

Section 3. ADMINISTRATION OF PLAN

3.1 <u>Plan Administrator</u>. The Company as Administrator, in conjunction with the Retirement Committee, has the general powers and authority to administer provided below in Sections 3.1.1 to 3.1.3:

Page 7 of 65 NORDSTROM 401(k) PLAN

- 3.1.1 <u>Complete Administrative Power</u>. The complete power and authority, in its sole discretion, to implement and delegate all functions necessary or desirable for the proper administration of the Plan, including but not limited to powers set forth in this Section 3.
- 3.1.2 <u>Actions Binding</u>. Any action taken in good faith in the exercise of authority conferred by this Plan shall be conclusive and binding upon the Participants and their beneficiaries.
- 3.1.3 <u>Discretion is Absolute</u>. All discretionary powers conferred upon the Administrator and Retirement Committee, as applicable, shall be absolute, provided, however, that no discretionary power shall be exercised in a manner that results in discrimination in favor of Employees who are officers, shareholders or Highly Compensated Employees of an Employer.
- 3.2 <u>Enumerated Administrative Powers.</u> Without limitation of its general powers under the Plan, the Company and Retirement Committee, as applicable, shall have the following enumerated powers:
- 3.2.1 <u>Control Administration</u>. Full power and authority to control and manage the operation and administration of the Plan.
- 3.2.2 <u>Plan Interpretation</u>. To construe and apply all Plan and Trust provisions, including the specific power and authority to interpret the Plan and Trust, to remedy or resolve ambiguities, inconsistencies or omissions and to decide any questions about the rights of Participants and their beneficiaries.
- 3.2.3 <u>Benefit Eligibility</u>. To decide all questions relating to an individual's status as an Employee, the eligibility of Employees to become Participants, the amount of service of any Employee or Participant, and the amount of benefits to which any Participant may be entitled by reason of service prior to or after the effective date hereof.
- 3.2.4 <u>Benefit Payment</u>. To approve the payment of all benefits as they become payable under the Plan and to pursue the recovery of any payment made which exceeds the amount to which an individual is entitled to receive under the terms of the Plan.
- 3.2.5 <u>Service Providers</u>. To engage such professional consultants, assistants and service providers as the Administrator, in its discretion, deems advisable, necessary or appropriate, including (but not limited to) accountants, actuaries, consultants, legal counsel, medical practitioners and clerical assistants to perform services with regard to any of its responsibilities under the Plan, and to rely on opinions and advice given by any such third party.
 - 3.2.6 <u>Records</u>. To ensure that all records necessary for proper operation of the Plan are kept.

Page 8 of 65 NORDSTROM 401(k) PLAN

- 3.2.7 <u>Reports and Disclosures</u>. To ensure compliance with all reporting, filing and disclosure requirements imposed on the Plan "administrator" by ERISA and any other applicable law.
- 3.2.8 <u>Inspection of Records</u>. During business hours to make available to service providers and any Participant or beneficiary any records relating to the Plan as required by law, provided that a Participant or beneficiary shall be entitled to examine only such records as pertain exclusively to him or her, including (but not limited to) the Plan and Trust Agreement and all amendments thereto.
- 3.2.9 <u>Indemnity Bond</u>. To arrange for all bonds required by law, but the amount thereof need not exceed the minimum requirements imposed by law.
- 3.2.10 <u>Legal Process</u>. To designate an agent for service of legal process in any suit or action involving the Plan.
- 3.2.11 <u>Fees and Expenses</u>. To negotiate and fix the compensation or fees, as the case may be, of all officers, agents, counsel, the Trustee, or other person retained or employed by the Administrator or other party designated to carry out administrative duties under the Plan.
- 3.2.12 <u>Other</u>. To perform or cause to be performed such further acts as it may deem necessary, appropriate or convenient for the efficient administration of the Plan.
- 3.3 <u>Administrative Records</u>. Each party having responsibility for any Plan administration function under the Plan shall keep such records as shall be appropriate for the orderly and efficient performance of such functions, and shall permit any other party having Plan administration responsibility to examine any of such records which are appropriate to the latter's functions.
- 3.4 <u>Employer Records</u>. The records of the Employers shall be conclusive evidence as to all matters forming the basis for participation in the Plan and for the calculation of benefits thereunder. Any individual or entity shall be entitled to rely upon a certificate of an officer of the Company as to any Employee's Years of Service, age, Compensation and cause for the severance of service, and as to any other information pertinent to the calculation or determination of the Employee's interest under the Plan.
- 3.5 <u>Duties of Participant</u>. The Administrator may require a Participant to furnish to it such information and instruments or documents as it may deem necessary in the administration of the Plan. Compliance with such requirements shall be a condition of a Participant's receipt of benefits.
- 3.6 <u>Administrator Expenses</u>. No Company employee who performs administrative functions under the Plan shall receive any compensation for such service beyond his or her compensation as an Employee of the Company, but shall be entitled to reimbursement from the

Page **9** of **65** NORDSTROM 401(k) PLAN

Company for any reasonable expenses actually and properly incurred in the performance of such duties.

- 3.7 <u>Individuals Indemnified</u>. The Company hereby indemnifies any Company Employee or Director who carries out any responsibilities under the Plan, and holds them harmless from the effects, consequences, expenses, attorney fees and damages arising from their acts or conduct in such capacity, except to the extent that such consequences are the result of their own willful misconduct or breach of good faith. Such indemnification shall be in addition to any other rights each may have as a matter of law, or by reason of any insurance or other indemnification.
- 3.8 <u>Administrator Continues Until Trust Exhausted</u>. If the Company shall cease to exist and no successor adopts or continues the Plan, the members of the Retirement Committee at that time (and their successors) shall remain in office until final termination of the Trust, and shall assume any and all powers and duties not otherwise previously delegated. The remaining member or members shall fill any vacancies caused by death, resignation, disability or other cause.

3.9 Plan Expenses.

- 3.9.1 Expenses Paid by Trust Fund. The following shall be paid by the Trust Fund:
- (a) <u>Operating Expenses</u>. All expenses of the Administrator and the Trust, as the case may be, attributable to the operation of the Plan and Trust, to the extent they constitute reasonable expenses of administering the Plan and are not paid by the Company under Section 3.9.2.
 - (b) <u>Taxes</u>. Any taxes and related interest and penalties assessed against the Trust Fund.
- 3.9.2 <u>Payment by Company Without Reimbursement</u>. Except for the reimbursement to the Company of direct expenses consistent with Section 17.7, the obligation of the Trust to pay any expenses charged to the Trust shall cease to exist to the extent such charges are paid by the Company.
- 3.9.3 <u>Administrator Protest</u>. Payment under Section 3.9.1 or 3.9.2 may be withheld pending resolution of any objection by the Administrator.

Section 4. ELIGIBILITY OF EMPLOYEES TO PARTICIPATE

4.1 Initial Participation.

4.1.1 <u>Elective Deferral Contributions</u>. An Eligible Employee begins participation for purposes of making Elective Deferrals and designated Roth contributions (including Catch-up Contributions, if applicable) immediately upon his or her Employment Commencement Date.

Page 10 of 65 NORDSTROM 401(k) PLAN

- 4.1.2 <u>Matching Contributions</u>. Effective January 1, 2021, Eligible Employees begin participation for purposes of receiving an allocation of Matching Contributions immediately upon his or her Employment Commencement Date. Notwithstanding the foregoing, Matching Contribution allocation eligibility for Eligible Employees with Employment Commencement Dates before January 1, 2021, shall continue to be determined under the Plan document in effect at that time.
- 4.2 <u>Break in Eligibility Service Reemployment After Break.</u> An Employee who becomes a Participant in this Plan remains a Participant until he or she receives a distribution of his or her entire vested account balance. If the Employee incurs a Permanent Break in Eligibility Service and subsequently is reemployed by the Employer, for Plan eligibility purposes the Participant's Employment Commencement Date shall be his or her most recent reemployment date. If an Employee severs employment and is subsequently reemployed by the Employer prior to incurring a Permanent Break in Eligibility Service, the Severance from Employment Date shall be disregarded for eligibility purposes and he or she will enter the Plan as provided in Section 4.1 following his or her rehire date. However, if the Employee incurs a Permanent Break in Eligibility Service prior to his or her reemployment date, the period of service prior to the Permanent Break in Eligibility Service is disregarded for Plan eligibility purposes.
- 4.3 <u>Information from Employer</u>. As of each Anniversary Date or such other period as the Company deems appropriate, the Employer will provide the Administrator or its designated agent with the appropriate information necessary to ascertain all Eligible Employees, their dates of employment, Hours of Service, Compensation, and dates of termination.

Section 5. CONTRIBUTIONS

- 5.1 Employer Profit Sharing Contribution
- 5.1.1 <u>Generally</u>. Effective January 1, 2021, the Employer will not make any discretionary profit-sharing contributions. For each Plan Year prior to January 1, 2021, the Employer could make a discretionary profit-sharing contribution in an amount to be determined by the Board of Directors of each Employer pursuant to the terms of the Plan document then in effect and shall be allocated in accordance with such Plan document.
- 5.1.2 <u>Forfeitures</u>. To the extent not used to restore amounts previously forfeited under Section 10.8.2, forfeitures under Section 8.3 for the then completed Plan Year shall be used to reduce the Employer contribution obligations or to pay expenses of Plan administration, as determined by the Retirement Committee in its sole discretion.
 - 5.2 <u>Elective Deferral Contributions</u>. The following provisions are effective January 1, 2021:
- 5.2.1 <u>Deferral Amount</u>. Each Participant may elect to defer a portion of his or her Compensation for any Plan Year in a whole percentage between one percent (1%) and fifty percent (50%). However, no Participant shall be permitted to have Elective Deferral Contributions made to this Plan, or any other qualified plan maintained by the Employer during

Page 11 of 65 NORDSTROM 401(k) PLAN

any taxable year, in excess of the dollar limitation contained in Code § 402(g) in effect at the beginning of such taxable year, except to the extent permitted under Section 5.2.3 of the Plan and Code § 414(v) relating to Catch-up Contributions. The Plan Administrator may, at any time, reduce the Elective Deferral Contributions for any Participant if it determines that reduction is necessary in order to avoid exceeding the limits imposed by this subsection or Section 6. The Plan Administrator shall provide for separate elections with respect to bonuses.

5.2.2 Qualified Automatic Contribution Arrangement on and after January 1, 2021. Effective for Plan Years beginning on and after January 1, 2021, the Plan is intended to be a "qualified automatic contribution arrangement" that satisfies the requirements of Code Sections 401(k)(13). To the extent that any other provision of the Plan is inconsistent with the provisions of this Section 5.2.2, the provisions of this Section 5.2.2 shall govern. The definitions in Section 5.2.2(f) below shall apply to capitalized terms used in this Section that are not defined elsewhere in the Plan.

(a) <u>Automatic Enrollment</u>. A Covered Employee will be automatically enrolled in the Plan as soon as administratively practicable following the effective date of this Section 5.2.2(a), or, if later, the date the individual becomes a Covered Employee by reason of hire, rehire or transfer from a nonparticipating affiliate. A Covered Employee's Automatic Salary Deferral Agreement shall take effect no later than the earlier of (i) the pay date for the second payroll period that begins after the date the notice is provided under Section 5.2.2(e) or (ii) the first pay date that occurs at least 30 days after such notice is provided.

(b) <u>Deemed Automatic Salary Deferral Agreement</u>. A Covered Employee who does not affirmatively enter into a Salary Deferral Agreement that takes effect before the Automatic Enrollment Date shall be deemed to have entered into an "Automatic Salary Deferral Agreement" under which the Employer will reduce his or her Safe Harbor Compensation in an amount equal to the percentages shown in Table 1 below.

Table 1:

Automatic Time Period Salary Deferral Contribution Percentage

For all payroll periods ending during the "Initial Period," where the Initial Period begins on the Automatic Enrollment Date and ends on the day immediately preceding the first anniversary of the Automatic Enrollment Date

3%

Page 12 of 65 NORDSTROM 401(k) PLAN

For all payroll periods on or after the first anniversary of the Automatic Enrollment Date to the date immediately preceding the second anniversary of the Automatic Enrollment Date	4%
For all payroll periods on or after the second anniversary of the Automatic Enrollment Date to the date immediately preceding the third anniversary of the Automatic Enrollment Date	5%
For all payroll periods on or after the third anniversary of the Automatic Enrollment Date to the date immediately preceding the fourth anniversary of the Automatic Enrollment Date	6%
For all payroll periods on or after the fourth anniversary of the Automatic Enrollment Date to the date immediately preceding the fifth anniversary of the Automatic Enrollment Date	7%
For all payroll periods on or after the fifth anniversary of the Automatic Enrollment Date to the date immediately preceding the sixth anniversary of the Automatic Enrollment Date	8%
For all payroll periods on or after the sixth anniversary of the Automatic Enrollment Date to the date immediately preceding the seventh anniversary of the Automatic Enrollment Date	9%
For all payroll periods ending on or after the seventh anniversary of the Automatic Enrollment Date	10%

Contributions made pursuant to an Automatic Salary Deferral Agreement shall cease as soon as administratively practicable after a Covered Employee makes an affirmative election to make Salary Deferrals or opt out of making Salary Deferrals. Automatic Salary Deferrals shall be

Page 13 of 65 NORDSTROM 401(k) PLAN

suspended during any approved leave of absence or any other payroll period for which a Participant does not receive Safe Harbor Compensation.

After a leave of absence or other suspension period, Salary Deferrals shall resume in accordance with the Covered Employee's affirmative election previously in effect, if any. If no affirmative election was in effect, the Covered Employee shall be subject to automatic enrollment at a rate determined as if the Covered Employee had been employed during the duration of the leave or suspension period in accordance with Table 1 (unless the Covered Employee makes a new affirmative election to make Salary Deferrals or opt out of making Salary Deferrals).

Upon a Covered Employee's Severance from Employment, any Automatic Salary Deferrals shall cease and any affirmative Salary Deferral election in effect at that time shall be cancelled. On rehire, the Covered Employee shall be subject to automatic enrollment (unless the Covered Employee makes a new affirmative election to make Salary Deferrals or opt out of making Salary Deferrals) as follows:

		If rehired more than 1 Plan Year after Severance from Employment:
_	At the rate set forth in Table 1 as if the Covered	
_	Employee had not experienced a Severance from Employment	Employee was a new employee

(c) <u>Treatment of Catch-up Contributions</u>. If the Covered Employee is eligible to make Catch-Up Contributions and the amount of Automatic Salary Deferrals made under the table set forth above exceeds the limit on 401(k) Contributions for a Plan Year, the excess amount shall be characterized as a Catch-Up Contribution to the maximum extent permitted under Code § 414(v).

(d) <u>Withdrawal Restrictions</u>. Automatic Salary Deferrals shall be subject to the withdrawal restrictions of Section 9.8.

(e) <u>Notice Requirement</u>. At least 30 days, but not more than 90 days, before the beginning of the Plan Year, the Employer will provide each Covered Employee a comprehensive notice of the Covered Employee's rights and obligations under the Automatic Contribution Arrangement, written in a manner calculated to be understood by the average Covered Employee. If an Employee becomes a Covered Employee after the 90th day before the beginning of the Plan Year and does not receive the notice for that reason, the notice will be provided no more than 90 days before the Employee becomes a Covered Employee but not later than the date the Employee becomes a Covered Employee. (If it is not practicable for notice to be provided by the date the Employee becomes a Covered Employee, notice shall be provided as soon as practicable thereafter, and the Covered Employee shall be permitted to elect to defer from Safe Harbor Compensation earned since becoming a Covered Employee.) Notice shall be

provided sufficiently early so that the Covered Employee has a reasonable period of time after receipt to opt out of participation, make an affirmative deferral election, and make an investment election. The notice shall describe (i) the amount of Automatic Salary Deferrals that will be made on the Covered Employee's behalf in the absence of an affirmative election; (ii) the Covered Employee's right to elect to have no Salary Deferrals made on his or her behalf or to have a different amount of Salary Deferrals made; and (iii) how Automatic Salary Deferrals will be invested in the absence of the Covered Employee's investment instructions.

(f) Definitions:

- i. <u>Automatic Salary Deferrals</u>. Automatic Salary Deferrals means pretax contributions made on a Covered Employee's behalf pursuant to an Automatic Salary Deferral Agreement.
- ii. <u>Automatic Salary Deferral Percentage</u>. Automatic Salary Deferral Percentage means the percentage of a Covered Employee's Safe Harbor Compensation contributed to the Plan as an Automatic Salary Deferral for each pay period in a Plan Year.
- iii. <u>Automatic Enrollment Date</u>. Automatic Enrollment Date for purposes of Section 5.2.2(b), the term "Automatic Enrollment Date" means the date a Covered Employee first has Automatic Salary Deferrals made to the Plan, but no earlier than the date the Plan first became a qualified automatic contribution arrangement under Code Section 401(k) (12).
- iv. Covered Employee. Covered Employee means:
 - (A)All Eligible Employees who are hired (or rehired or transferred from a nonparticipating affiliate) by the Employer on or after January 1, 2021, who do not affirmatively enter into a Salary Deferral Agreement or affirmatively opt out of Plan participation; and
 - (B) All Eligible Employees hired by the Employer prior to January 1, 2021, who have not previously made an affirmative election (that remains in effect on January 1, 2021) to make Salary Deferrals or opt out of making Salary Deferrals.
- v. <u>Safe Harbor Compensation</u>. Safe Harbor Compensation means Compensation as set forth in Section 2.6.
 - 5.2.3 Catch-up Contributions. Each Participant who:
 - (a) is eligible to make Elective Deferral Contributions under this Plan; and
- (b) has attained or will attain age fifty (50) before the last day of the Plan Year, is eligible to make Catch-up Contributions in accordance with, and subject to, Code

Page **15** of **65** NORDSTROM 401(k) PLAN

§ 414(v). Catch-up contributions are those Elective Deferral Contributions, up to the applicable dollar limit set forth in Code § 414(v)(2)(B)(i), as adjusted for cost of living, that would exceed a contribution limit under the Code or the Plan if the provisions of Code § 414(v) were not applicable. Each Participant described in this Section 5.2.3 shall have the same right to elect Catch-up Contributions under this Plan. Catch-up Contributions must be allocated to separate Catch-up Contribution Accounts and will not be taken into account for purposes of the provisions of this Plan implementing Code §§ 402(g) regarding limits on elective deferrals, or 416 regarding contributions by Key Employees (except as specifically provided in Section 12.)

5.2.4 <u>Changes to Deferral Election.</u> During employment, a Participant may modify, suspend or resume Elective Deferral Contributions by any telephonic, electronic or written means established by the Administrator. Any such change shall be effective as of the first day of the next payroll cycle following processing of the change notification received by the Administrator; provided, however, that if the Administrator is not able to administratively process the change by such payroll date, the change shall be effective as soon as the administrative processing is complete. Notwithstanding the foregoing, a Participant's Elective Deferral Contributions election will be automatically revoked as soon as administratively possible following the Participant's Severance from Employment Date. A rehired Participant must reinitiate a new Elective Deferral Contribution election in the manner specified by the Administrator.

5.2.5 Excess Deferrals. "Excess Deferral" means, for a given calendar year, that amount by which each Participant's total elective deferrals (as defined in Code § 402(g)(3)) under all plans of all employers exceeds the sum of the dollar limits in effect under Code § 402(g) for the calendar year and under Code § 414(v) for the taxable year, as annually indexed by the Secretary of the Treasury. The Plan Administrator will distribute any Excess Deferral, adjusted for investment gains and losses, to the Participant no later than April 15 of the calendar year immediately following the close of the calendar year for which the Excess Deferral is made. If an Excess Deferral occurs because of deferral amounts under plans maintained by an Employer combined with deferrals under one or more plans not maintained by an Employer, the excess shall be distributed if the following conditions are satisfied:

FIRST, the Participant notifies the Plan Administrator of the Excess Deferral by March 1 following the close of the year, unless the Plan Administrator waives the deadline; and

SECOND, the notice specifies how much of the Excess Deferral is to be distributed from this Plan.

5.2.6 Deferral and Catch-up Contributions Accounts. The amount by which Compensation is reduced, after adjustment for Excess Deferrals under Section 5.2.5, shall be that Participant's Elective Deferral Contribution. The portion of the Elective Deferral Contribution that does not exceed the Plan provisions implementing Code, § 402(g) regarding limits on elective deferrals for all Participants, or § 415 regarding limits on annual additions shall be called the Basic Elective Deferral Contribution and shall be allocated to that Participant's Basic Elective Deferral Account. The remainder of the Elective Deferral Contribution shall be called

Page 16 of 65 NORDSTROM 401(k) PLAN

the Catch-up Contribution and shall be allocated to a Participant's Catch-up Contributions Account.

5.3 <u>Designated Roth Contributions</u>.

- 5.3.1 <u>Designation of Contributions</u>. Beginning September 1, 2007, each Participant may make designated Roth contributions to the Plan. A designated Roth contribution is an Employee Contribution that is (a) designated irrevocably by the Participant at the time of deferral as a designated Roth contribution; (b) made in lieu of all or a portion of the pre-tax Elective Deferral Contributions the Participant is otherwise eligible to make under the Plan; (c) treated by the Employer as includible in the Participant's income at the time the Participant would have received that amount in cash if the Participant had not contributed the amount to the Plan. The Plan provisions set forth in Sections 5.2.1 and 5.2.4 shall apply to designated Roth contributions by substituting "designated Roth contribution" for "Elective Deferral Contribution" each place that the latter term appears. In addition, designated Roth contributions are eligible for treatment as Catch-Up Contributions for Participants who will have attained age 50 by the last day of the calendar year in which the contribution is made.
- 5.3.2 <u>Separate Accounting</u>. Designated Roth contributions, and gains, losses, and other credits or charges will be credited and debited to a separate designated Roth contributions account maintained for each Participant. No contributions other than designated Roth contributions (including designated Roth rollover contributions) and properly attributable earnings will be credited to each Participant's designated Roth contributions account.
- 5.3.3 Correction of Excess Amounts. In case it is necessary to make a distribution to a Participant due to a failure of the Plan to pass the ADP test set forth in Section 6.8, due to an excess deferral under Code § 402(g), or due to an excess annual addition under Code § 415(c), the Plan first will distribute a Participant's pre-tax Elective Deferral Contributions plus earnings for the Plan Year and will distribute designated Roth contributions only to the extent necessary after distribution of the Elective Deferral Contributions. If the Plan re-characterizes and retains excess Elective Deferral Contributions that are treated as Catch-up Contributions in order to satisfy the ADP test, the Plan shall first re-characterize designated Roth contributions as Catch-up Contributions and shall then re-characterize pre-tax Elective Deferral Contributions, up to the limit for Catch-up Contributions in effect for the applicable Plan Year. If it is necessary to make a corrective distribution of designated Roth contributions, earnings attributable to the corrective distribution of designated Roth contributions shall be distributed to the same extent that a distribution of earnings on Elective Deferral Contributions would be required to effect a full corrective distribution of Elective Deferral Contributions.

5.4 Employer Matching Contributions.

5.4.1 <u>Description of Matching Contributions</u>. Effective January 1, 2021, the Employer shall make Matching Contributions with respect to Salary Deferrals, including Automatic Salary Deferrals, but not including Catch-up Contributions made pursuant to Section 5.2.3, according to the terms of Section 5.2.2(b) and the following formula:

Page 17 of 65 NORDSTROM 401(k) PLAN

- i. 100% of the Covered Employee's Salary Deferrals that do not exceed one-percent (1%) of the Covered Employee's Safe Harbor Compensation.
- ii. 50% of the Covered Employee's Salary Deferrals that exceed one-percent (1%) of the Covered Employee's Safe Harbor Compensation but that do not exceed seven percent (7%) of the Covered Employee's Safe Harbor Compensation.

Prior to January 1, 2021, the Company had the discretion to make Matching Contribution as determined by resolution of the Board in accordance with the terms of the Plan document then in effect.

- 5.4.2 <u>Requirements For Match</u>. Effective January 1, 2021, every Participant is eligible to receive an Employer Matching Contribution. Eligibility for Employer Matching Contributions prior to January 1, 2021 will be determined in accordance with the Plan document then in effect.
- 5.5 <u>Time of Payment of Contributions</u>. The Employer shall pay to the Trustee Employer Contributions for each Plan Year within the time prescribed by law, which may extend beyond the end of the Plan Year in accordance with Code § 404(a)(6). On or about the date of the payment, the Administrator shall be advised of the amount of the payment upon which the allocation shall be calculated.
- 5.6 <u>Plan Qualification</u>. Notwithstanding any provisions in this Plan to the contrary, contributions to this Plan are made upon the condition precedent that this amended and restated Plan must be qualified under Code § 401(a). Accordingly, the Employer reserves the right to amend this Plan, retroactively or otherwise, as may be required in order to obtain approval of the Plan from the Internal Revenue Service.

5.7 Return of Mistaken and Nondeductible Contributions.

- $5.7.1 \, \underline{\text{Mistake of Fact}}$. In the event that an Employer shall make an excessive contribution due to a mistake of fact, then pursuant to § 403(c)(2)(A) of ERISA, an Employer may demand repayment of such excessive contribution at any time within one (1) year following the time of payment and Trustee shall return that amount to the Employer within the one (1) year period. Earnings of the Plan attributable to the excess contributions may not be returned to the Employer, but any losses attributable thereto must reduce the amount so returned.
- 5.7.2 <u>Disallowed Deduction</u>. Employer contributions hereunder are made on the condition that such contributions are deductible under Code §§ 401(a) and 404. In the event that a deduction for any contribution hereto is disallowed and found not to be deductible by the Internal Revenue Service, or any other regulatory agency, the Employer may recover all or any portion of such contribution, provided it is returned to the Employer within one (1) year after the denial of the deduction.

Page **18** of **65** NORDSTROM 401(k) PLAN

- 5.7.3 <u>No Participant Interest</u>. No Participant or beneficiary has any vested right or claim to any asset of the Plan or to any benefit under the Plan that may be returned pursuant to Section 5.7 of this Plan.
- 5.8 <u>Military Leave Obligations</u>. Notwithstanding any provision of this Plan to the contrary, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with Code § 414(u). Additionally, effective with respect to periods of qualified military service commencing on and after January 1, 2004, a Participant's obligation to make loan repayments will be suspended under this Plan as permitted under Code § 414(u)(4).
- 5.8.1 <u>Returning Participant with Re-Employment Rights</u>. The following provisions apply to each Participant who, immediately following a period of qualified military service, returns to employment with an Employer with reemployment rights protected by law.
- (a) <u>Employer Profit Sharing Contribution</u>. The Employer shall make Employer Profit Sharing Contributions for the period of the Participant's qualified military service, based on a rate derived from the amount of contribution made to the Plan for each Plan Year in such period, and on the Compensation for the Participant (as determined in Section 5.8.2).
- (b) <u>Employee Contributions</u>. A Participant may make Elective Deferral Contributions or designated Roth contributions to the Plan attributable to a period of qualified military service. Such contributions shall be paid within a period starting on the date of reemployment and continuing for the shorter of (1) three (3) times the length of the qualified military service that resulted in the reemployment rights or (2) five (5) years. Effective January 1, 2009, notwithstanding the preceding sentence, if a Participant receives a distribution under Section 9.9 (which treats certain military service as a severance from employment for distribution purposes), the Participant may not make Elective Deferral Contributions or designated Roth contributions to the Plan during the six-month period beginning on the date of distribution.
- (c) <u>Matching Contributions</u>. The Employer shall make Matching Contributions for a Participant who is reemployed from qualified military service based on the amount of Elective Deferral Contributions made by the Participant under Section 5.8.1(b).
- 5.8.2 <u>Compensation</u>. Compensation for purposes of contributions under Section 5.8.1 shall be the amount described in Section 2.6 that the Participant would have received from the Employer during the period of qualified military service if employment had continued. Such amount shall be based on the rate of pay the Participant would have received in such period or, if such rate was not reasonably certain, the Participant's average pay rate during the 12-month period of employment preceding the period of qualified military service or the entire period of employment if less than 12 months.
- 5.8.3 <u>Limitations</u>. Contributions provided under Section 5.8.1 shall be subject to the limits provided in Section 6 based on the Plan Years within the period of qualified military service to which the contributions relate in accordance with applicable law and regulations.

5.9 Rollover Contributions.

- 5.9.1 <u>General Rule</u>. Subject to the approval of the Administrator, this Plan may accept an eligible rollover distribution on behalf of a Participant who is an Eligible Employee from any of the following:
 - (a) a qualified plan described in Code § 401(a) or § 403(a);
 - (b) an annuity contract described in Code § 403(b);
- (c) an eligible plan under Code § 457(b) that is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state;
- (d) an individual retirement account or retirement annuity described in Code § 408(a) or (b) that is eligible to be rolled over and would otherwise be includible in gross income.

For purposes of this Section 5.9, the plans and arrangements described in (a) through (d) are referred to as "the other plan."

- 5.9.2 <u>Eligible Rollover Distribution</u>. The rollover must be an eligible rollover distribution, as defined in Section 10.7 paid to or on behalf of the Participant either:
 - (a) pursuant to participation of the Participant or the Participant's deceased Spouse in the other plan; or
- (b) pursuant to a qualified domestic relations order to the Spouse or former Spouse of a participant in the other plan.
 - 5.9.3 Qualified Transfer. The rollover must be paid in cash to the Trustee either:
 - (a) by a direct transfer from the trustee(s) of the other plan or IRA; or
- (b) by payment from the Participant on or before the sixtieth (60th) day following the Participant's receipt of the distribution from the other plan or IRA.
- 5.9.4 <u>Rollover Account</u>. The transferred amount accepted by the Plan shall be placed in the Participant's Rollover Account, and shall be at all times fully vested and subject to the investment and distribution provisions of Section 5.2, but shall not be considered a Participant Elective Deferral Contribution for purposes of the Employer Matching Contribution, contribution limits, or nondiscrimination requirements and limitations of this Plan and the Code, or as part of a Participant's total account balance for purposes of the consent requirement under Section 10.1.1 for involuntary distribution of account balances. Rollovers of after-tax contribution amounts described in Section 10.7 will be accounted for separately.

Page 20 of 65 NORDSTROM 401(k) PLAN

5.9.5 <u>Designated Roth Rollover Contributions</u>. The Plan will accept a rollover contribution to a Participant's designated Roth contributions account only if it is a direct rollover from another Roth contributions account under an applicable retirement plan described in Code § 402A(e)(1) and only to the extent the rollover is permitted under the rules of Code § 402(c). The rollover contribution will be accounted for in the Participant's designated Roth contributions account and not as part of the Participant's Rollover Account.

5.10 Qualified Non-Elective Contributions.

5.10.1 <u>Generally</u>. A "Qualified Non-Elective Contribution" (QNEC) means a non-elective contribution which is 100% non-forfeitable at all times, is subject to the distribution restrictions under Section 9.8, is allocated to the Participant's QNEC Account as of a date within the Plan Year being tested, and is actually contributed to the Plan within the 12 month period immediately following such Plan Year. A QNEC under this Section 5.10 shall be considered an Employer contribution for purposes of the Employer's minimum employer contribution obligations under Section 12.4.1.

5.10.2 <u>Allocation of QNECs</u>. QNECs will be allocated to the QNEC Account of each Participant who meets the eligibility requirements under Section 5.10.4 in reverse order of Compensation as provided for herein, subject to the limitations under Section 5.10.3. The QNEC will be allocated to the eligible Participant with the lowest Compensation until all of the QNEC has been allocated. If two or more eligible Participants have the same Compensation, the QNEC will be allocated equally to each eligible Participant until all of the QNEC has been allocated. If any QNEC remains unallocated, this process is repeated for the eligible Participant(s) with the next lowest level of Compensation in accordance with this paragraph until all of the QNEC is allocated, within the limits provided under Section 5.10.3. The portion of any QNEC that cannot be allocated due to the limitations under Section 5.10.3 shall be treated as an additional Employer Profit Sharing Contribution and allocated pursuant to Section 5.1.

5.10.3 QNEC Allocation Limits. The maximum QNEC allocated to any eligible Participant shall not exceed the least of:

- (a) the amount sufficient to satisfy the ADP or ACP test(s) under Section 6.8;
- (b) the Participant's Annual Addition Limitation for the Plan Year under Section 6.6; or
- (c) the amount equal to the Participant's Compensation multiplied by the greater of:
 - (1) five percent (5%); or
 - (2) two times the Plan's Representative Contribution Rate.

Page 21 of 65 NORDSTROM 401(k) PLAN

- 5.10.4 QNEC Eligibility. Eligibility to receive QNEC allocations for a Plan Year is limited to Participants who, as of the last day of such Plan Year, are: (a) Non-Highly Compensated Employees; (b) eligible to receive Employer Matching Contributions pursuant to Section 5.3.2; and (c) not included in the "otherwise excludible" testing group under Section 6.8.2.
- 5.10.5 <u>Representative Contribution Rate</u>. The "Representative Contribution Rate" for purposes of Section 5.10.3(c)(2) is the lowest Applicable Contribution Rate of any eligible Participant among a group of eligible Participants that consists of half (50%) of all eligible Participants for the Plan Year or, if greater, the lowest Applicable Contribution Rate of any eligible Participant in the group of all eligible Participants for the Plan Year and who is employed by the Employer on the last day of the Plan Year.
- 5.10.6 <u>Applicable Contribution Rate</u>. The "Applicable Contribution Rate" under Section 5.10.5 for an eligible Participant equals the Participant's QNEC allocation for a Plan Year divided by the Participant's Compensation for the same period.
- 5.10.7 <u>Compensation for QNEC Purposes</u>. Compensation for purposes of this Section 5.10 is Compensation under Section 6.8.2(d).
- 5.10.8 <u>Investment of QNEC Accounts</u>. A Participant's QNECs will be invested in the same manner as his or her election for future Elective Deferral Contributions.

Section 6. ALLOCATION OF CONTRIBUTIONS AND VALUATION OF TRUST FUND

- 6.1 Allocation of Contributions and Forfeitures.
- 6.1.1 <u>Participant Accounts</u>. The Company, or its delegated third-party administrative service provider, under the supervision of the Retirement Committee, shall keep such separate accounts for each Participant as may be necessary to administer the Plan properly and to accurately reflect the value of the account of each Participant or Beneficiary in the Plan. Such accounts and records may be kept in dollars or units or both, as determined appropriate by the Company so that there may be determined as of any Valuation Date the (a) current value of the Participant's account in the Trust Fund and (b) adjustments from the previous Valuation Date that have produced such value. The Company, or its delegate, shall furnish each Participant a statement showing contributions to date, account balances and vested interests. Such statement shall be furnished no less frequently than annually.
- 6.1.2 <u>Valuation Changes</u>. Each Participant's account shall be adjusted to reflect net income, gain or loss, since the previous Valuation Date, as provided in Section 6.2. For this purpose, Participant accounts are determined on a cash basis, not an accrual basis. Any appreciation or depreciation in the value of a Participant's account will apply only to amounts actually invested under that Participant's account.

Page 22 of 65 NORDSTROM 401(k) PLAN

- 6.2 <u>Valuation and Allocation of Trust Fund</u>. The Trust Fund shall be valued and allocated on each Valuation Date. As of the close of trading on each Valuation Date, the fair market value of each Participant's account shall be determined as follows:
 - FIRST, credit or charge, as appropriate, to the proper accounts all contributions, transfers, payments, fees, forfeitures, withdrawals or other distributions made to or from such accounts since the last Valuation Date and that have not been previously credited or charged.
 - SECOND, credit or charge, as applicable, each account with its share of the appreciation or depreciation in the fair market value of the investments held in each account since the previous Valuation Date. Such appreciation or depreciation will reflect investment income, unrealized gains and losses, other investment transactions and expenses paid from the Plan Assets and other charges properly payable by the Plan in accordance with Section 17.7.
- 6.3 <u>Investment of Contributions</u>. All Contributions and investment earnings, gains or losses thereon, credited to a Participant's account shall be invested and reinvested in one or a combination of investment funds to be established by the Trustee as provided in the Nordstrom 401(k)Plan Participant Investment Direction Appendix attached hereto and incorporated into this Section 6.3 by this reference.
- 6.4 <u>Allocation Does Not Vest Rights</u>. The fact that an allocation is made and credited to the a count of a Participant does not vest in the Participant any right, title or interest in and to any assets except at the time or times and upon the terms and conditions expressly set forth in this Plan.

6.5 Forfeiture Suspense Account.

- 6.5.1 <u>Assets Pending Allocation</u>. Any amounts forfeited pursuant to Sections 8.2, 8.3 or 10.8 shall be held in an account to be known as the "forfeiture suspense account" until allocated pursuant to Section 6.5.3.
- 6.5.2 <u>Investment of the Forfeiture Suspense Account</u>. The forfeiture suspense account referred to in this section shall be invested in a liquid form of investment as determined appropriate by the Company.
- 6.5.3 <u>Allocation of Forfeitures held in the Forfeiture Suspense Account</u>. The forfeiture suspense account will be used first to restore any previously forfeited amounts under Section 10.8.2, and then to reduce Company contributions as provided under Section 5.1.2.

6.6 Limitation on Annual Additions.

6.6.1 <u>Annual Maximum for All DC Plans</u>. Notwithstanding any provisions of this Plan to the contrary, when taking into consideration all defined contribution Plans maintained by Employer, the maximum "annual addition" that may be contributed or allocated to

Page 23 of 65 NORDSTROM 401(k) PLAN

a Participant's account or accounts for any Limitation Year may not exceed the lesser of (1) \$40,000 or (2) one hundred percent (100%) of the Participant's Compensation. The \$40,000 "dollar limitation" shall be adjusted for increases in the cost of living in accordance with regulations prescribed by the Secretary of the Treasury.

- 6.6.2 <u>Annual Addition</u>. With respect to each Participant, "annual addition" means the sum of the following amounts allocated to a Participant for the Limitation Year: (1) Employer contributions (including Elective Deferral Contributions and Designated Roth Contributions, but excluding Catch-up Contributions), (2) the amount of the Participant's contributions determined without regard to any Rollover Contributions or loan repayments, and (3) forfeitures. In addition, contributions to the following types of arrangements are considered contributions to defined contribution plans for purposes of Code § 415: (1) amounts allocated to an individual medical account of a pension or annuity plan established pursuant to Code § 401(h), and (2) amounts attributable to post-retirement medical benefits allocated pursuant to Code § 419(A)(d)(2) to the separate account of individual who is or was a Key Employee during the Plan Year or any preceding Plan Year.
- 6.6.3 <u>Combined Employers</u>. For purposes of applying the limitations under Section 6.6, all members of a controlled group of corporations (as defined by Internal Revenue Code § 414(b) but modified by Code § 415(h)) or of an affiliated service group (as defined by Internal Revenue Code § 414(m)) of which Employer is a member, and all employers which are under common control with Employer (as defined by Internal Revenue Code § 414(c) but modified by Internal Revenue Code § 415(h), and any other entity required to be aggregated with the Employer pursuant to regulations under Code § 414(o), will be considered a single employer.
- 6.6.4 <u>Compensation for 6.6.</u> For the sole purpose of determining the contribution limitation under Section 6.6, an Employee's compensation for a Limitation Year shall be defined to include earned income, wages, salaries and fees for professional services and other amounts paid or includible in gross income for the Limitation Year for personal services actually rendered in the course of employment with the Employer (including, but not limited to, commissions paid for sales, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips and bonuses), excluding the following (a) and (b), but including (c), as applicable:
- (a) <u>Deferred Compensation Excluded</u>. Contributions to a qualified or nonqualified plan of deferred compensation which are not includible in the Employee's gross income for the taxable year in which contributed, or Employer contributions under a simplified employee pension plan to the extent such contributions are deductible by the Employee, or any distributions from a plan of deferred compensation are excluded from compensation for Section 6.6.
- (b) <u>Stock Benefits Excluded</u>. Amounts realized from the exercise of a nonqualified stock option; or when restricted stock (or property) held by the Employee either becomes freely transferable or no longer is subject to a substantial risk of forfeiture; or amounts realized from the sale, exchange or other disposition of stock acquired under a qualified stock option are excluded from compensation for Section 6.6.

Page 24 of 65 NORDSTROM 401(k) PLAN

- (c) <u>Certain Other Non-Taxable Amounts Included</u>. For Plan Years beginning after December 31, 1997, any elective deferral (as defined in Code § 402(g)(3)), and any amounts contributed by the Employer at the election of the Employee which are not includible in the gross taxable income of the Employee by reason of Code §§ 125 or 132(f) are included in compensation for Section 6.6.
- 6.7 <u>Allocation of Excess Additions</u>. If an allocation would have been made to a Participant's account, but for a limitation in Section 6.6, then any such excess shall be disposed of in a manner permitted under the Employee Plans Compliance Resolution System, or similar program, which may include the following:
- 6.7.1 Excess Attributable to Elective Deferrals. If the excess is attributable to amounts contributed by the employee as Elective Deferrals, then any Elective Deferrals, and any income attributable thereto, to the extent they would reduce the excess amount, may be returned to the Participant.
- 6.7.2 <u>Remainder to Suspense Account</u>. Any excess addition that is not attributable to Elective Deferrals and remaining after the application of Section 6.7.1, may be allocated to a suspense account as forfeitures and held therein until the next succeeding date on which forfeitures could be applied under the Plan. In the event of termination of the Plan, the suspense account shall revert to the Employer to the extent that it may not then be allocated to any Participants' accounts.
- 6.7.3 <u>Multiple DC Accounts</u>. In the event that Employer maintains two (2) or more defined contribution plans and the total annual additions to all plans exceeds the limitation contained in Section 6.6 above, the provisions of Section 6.6 shall be applied to all profit sharing plans to the extent necessary to comply with Section 6.6.
- 6.7.4 <u>Code Section 415</u>. The intent of Sections 6.6 and 6.7 is to set forth the basic rule implementing Code § 415 so that, for each Plan Year, the Plan satisfies the contribution limitations of the Code and applicable regulations. The provisions of Sections 6.6 and 6.7 shall be applied in a manner consistent with Code § 415 and applicable guidance.

6.8 Contribution Limits for Highly Compensated Employees.

6.8.1 Non-Discrimination Tests. Effective January 1, 2021, the Plan is intended to qualify as a Qualified Automatic Contribution Arrangement and to satisfy the safe harbor provisions to avoid the Actual Deferral Percentage test and, as a result, the provisions of Section 6.8 shall no longer apply unless the QACA is terminated mid-year or as otherwise required by law. Prior to January 1, 2021, for each Plan Year, the Plan shall satisfy the nondiscrimination tests in Code §§ 401(k)(3) and 401(m) in accordance with Treasury Regulation sections 1.401(k)-2 and 1.401(m)-2. The applicable Code and Regulation sections are incorporated by this reference. The following provisions shall be applied in a manner consistent with such Code and Regulation sections.

Page 25 of 65 NORDSTROM 401(k) PLAN

- 6.8.2 <u>Determining the ADP and ACP</u>. For each Plan Year in which the Plan is required to perform such tests, the Committee shall determine the Actual Deferral Percentage ("ADP") and the Actual Contribution Percentage ("ACP") of the Eligible Employees who are Highly Compensated Employees under Section 2.13 and the ADP and ACP of the remaining Eligible Employees in two separate groups. Employees under age 21 or who have less than one Year of Service as of the end of the Plan Year are one group (the "otherwise excludable group"), and all other Employees are the other group. The "otherwise excludable group" shall not consist of any Highly Compensated Employees. The ADP and ACP shall be determined as follows:
- (a) The ADP (and ACP) for the Highly Compensated Employees and for the remaining Employees is the average of the Actual Deferral Rates (or Actual Contribution Rates) for all eligible Employees within their respective groups. The ADP (and ACP) for a group of eligible Employees shall be calculated to the nearest hundredth of a percentage point.
- (b) An Employee's Actual Deferral Rate ("ADR") is the sum of that individual's Basic Elective Deferral Contributions and QNECs for the Plan Year, divided by such Employee's Compensation under (d). The ADR is calculated to the nearest hundredth of a percentage point. Notwithstanding anything in the foregoing to the contrary:
- (1) Elective Deferral Contributions made pursuant to Section 5.8.1(b) (relating to Employees returning from qualified military service) shall not be taken into account when determining an Employee's ADR for the Plan Year for which the Basic Elective Deferral Contributions are made or for any other Plan Year.
- (2) Excess Deferrals which exceed the limitations under Code § 402(g)(3) shall be taken into account as Basic Elective Deferral Contributions when determining a Highly Compensated Employee's ADR for the Plan Year, even if those Excess Deferrals are distributed pursuant to Section 5.2.5.
- (3) Excess Deferrals which exceed the limitations under Code § 402(g)(3) shall not be taken into account as Basic Elective Deferral Contributions when determining a Non-Highly Compensated Employee's ADR for the Plan Year, to the extent such deferrals are prohibited under Code § 401(a)(30). However, to the extent such amounts are not prohibited under Code § 401(a)(30), they shall be taken into account for ADR purposes, whether or not distributed pursuant to Section 5.2.5.
- (c) An Employee's Actual Contribution Rate ("ACR") is that individual's Employer Matching Contributions for the Plan Year, divided by such Employee's Compensation under (d), subject to (e). The ACR is calculated to the nearest hundredth of a percentage point. Notwithstanding anything in the foregoing to the contrary:
- (1) Employer Matching Contributions made pursuant to Section 5.8.1(c) (relating to Employees returning from qualified military service) shall not be taken into account when determining an Employee's ACR for the Plan Year for which the Employer Matching Contributions are made or for any other Plan Year.

Page 26 of 65 NORDSTROM 401(k) PLAN

- (2) Any Employer Matching Contributions that are forfeited because the Elective Deferral Contributions to which they relate are treated as Excess Contributions or Excess Deferrals shall not be taken into account when determining an Employee's ACR for the Plan Year.
- (d) Compensation for ADR and ACR purposes is Compensation under Section 2.6, or such other definition of compensation permitted by Code § 414(s) in lieu thereof. Only Compensation earned while an Eligible Employee shall be considered for this purpose.
- (e) The Committee may for any Plan Year treat Basic Elective Deferral Contributions or QNECs not needed to pass the ADP test as Employer Matching Contributions for purposes of the ACP test. No single contribution may be used in both tests.
 - (f) The following shall be aggregated to determine the ADR and the ACR:
- (1) All Plans that are aggregated with this Plan under Code §§ 401(a)(4) and 410(b) (other than for purposes of the average benefit percentage test).
- (2) All cash and or deferred arrangements sponsored by the Employer in which the same Highly Compensated Employee is eligible to participate.
- 6.8.3 <u>ADP and ACP Limitations</u>. For Plan Years in which ADP or ACP testing is required, neither the ADP nor the ACP (as applicable) of the Highly Compensated Employees may exceed the greater of the following:
 - (a) 1.25 times the ADP or ACP of the remaining employees for the appropriate Plan Year.
- (b) 2 percentage points higher than the ADP or ACP of the remaining employees, up to 2 times such ADP or ACP for the appropriate Plan Year.

6.8.4 ADP and ACP Testing Methodology.

- (a) <u>Generally</u>. For Plan Years in which ADP or ACP testing is required, the Plan elects to use the current year testing method in computing the ADP and ACP (as applicable) for Non-Highly Compensated Employees under the nondiscrimination rules of Code §§ 401(k) and 401(m).
- (b) <u>Incorporation of Regulations</u>. For purposes of the limitations under this Section 6.8, the provisions of Code $\S\S 401(k)(3)$ and 401(m)(3) together with their specific underlying Treasury Regulations and subsequent Internal Revenue Service guidance are hereby incorporated into this Plan by reference.
 - 6.9 Correcting Excess Contributions.

Page 27 of 65 NORDSTROM 401(k) PLAN

- 6.9.1 <u>Determine the Excess Contribution Amounts</u>. For Plan Years in which ADP or ACP testing is required, if the ADP or ACP (as applicable) of the Highly Compensated Employees exceeds the limits in Section 6.8.3, the Committee shall adjust the contributions for certain Highly Compensated Employees, as follows:
- (a) <u>Correcting for ADP Failures</u>. If the ADP limit is exceeded, Basic Elective Deferral Contributions shall be reduced taking the highest individual dollar amount first. Basic Elective Deferral Contributions reduced under this provision shall not be eligible for Employer Matching Contributions.
- (b) <u>Correcting for ACP Failures</u>. If the ACP limit is exceeded, Employer Matching Contributions shall be reduced taking the highest individual dollar amount first.
- 6.9.2 Excess Contribution Reductions. Amounts reduced under Section 6.9.1 shall be forfeited, withheld or distributed as follows:
- (a) Any amount reduced from Employer Matching Contributions shall be forfeited, with related earnings, as follows:
- (1) Any amount reduced under Section 6.9.1(b) shall be forfeited to the extent of any unvested balance in the Employer Matching Contribution account of the Highly Compensated Employee to whom it applies. The unvested balance shall be determined before the reduction.
 - (2) Amounts forfeited shall be treated in accordance with Section 6.5.
- (b) Any Employer Matching Contribution for which eligibility is lost under Section 6.9.1(a) because a Basic Elective Deferral Contribution was reduced shall not be contributed and thus shall neither be forfeited nor distributed.
- (c) Subject to (d) and (e), any contributed amount not forfeited under (a) shall be distributed to the Highly Compensated Employees to whom it applies. The distribution shall be adjusted for allocable gain or loss, determined under applicable Regulations, for the Plan Year in which the excess arose ("Plan Year income"). Distribution of such amounts generally may be made within two and one-half (2½) months after the end of the Plan Year to which the excess applies (six months if the Plan is determined to include an eligible automatic contribution arrangement as defined in Code § 414(w)(3)) and in any event by the end of the following Plan Year.
- (d) A distribution under (c) because of an ADP limitation shall be reduced by the amount of any Excess Deferral previously withdrawn under Section 5.2.5 for the same Plan Year.

Page 28 of 65 NORDSTROM 401(k) PLAN

(e) Distributions under this Section 6.9 shall be made ratably from all of the investments selected by the Participant except any investments in the Nordstrom Common Stock Fund and the Participant Self-Directed Brokerage Account.

Section 7. [INTENTIONALLY OMITTED]

Section 8. VESTING OF BENEFITS

- 8.1 <u>Vesting.</u> A Participant is always 100 percent vested in Elective Deferral and Rollover Contributions to the Plan. A Participant is 100 percent vested in his or her entire Plan Account if, while an active Employee, the Participant attains the Normal Retirement Date under Section 9.1, suffers a Disability, or dies. In all other cases, the Participant will receive a percentage of the amount in his or her account derived from Employer Profit Sharing Contributions and Employer Matching Contributions based on his or her completed Years of Service. A Participant who dies while on a military leave described in Section 5.8 will be fully vested to the same extent as a Participant who dies while an active Employee.
- 8.1.1 <u>Vesting Service</u>. For purposes of determining a Participant's vested interest, the applicable computation period shall be the Plan Year for years commencing prior to January 31, 1986, and the Payroll Year thereafter. A Participant shall receive a Year of Service for vesting for the Plan Year ending January 31, 1986, and for the Payroll Year ending December 31, 1987.

Vesting Schedules.

(a) <u>Pre-2000 Profit Sharing Contributions</u>. A Participant's vested interest in Employer Profit Sharing Contributions attributable to Years of Service before January 1, 2000 will be determined in accordance with the following schedule:

Years of Service	Vested Interest
Less than 3 years	0
3 years	20
4 years	40
5 years	60
6 years	80
7 or more years	100

(b) <u>Post-1999 Profit-Sharing Contributions</u>. A Participant shall be immediately one hundred percent (100%) vested in Employer Profit Sharing Contributions made after December 31, 1999 unless the Company determines, in its sole discretion, that any Profit Sharing Contributions shall be subject to a vesting schedule.

(c) <u>Pre-2021 Matching Contributions</u>. For Matching Contributions made prior to January 1, 2021, each Participant who has at least one (1) Hour of Service earned prior to January 1, 2000 and for each Participant whose first Hour of Service is with Nordstrom Direct, Inc. after

December 31, 2002 is one hundred percent (100%) vested. For all other Participants, vesting of Matching Contributions is determined under the following schedule:

Years of Service	Vested Interest
Less than 1 year	0%
1 year	33%
2 years	67%
3 or more years	100%

(d) <u>Post-2020 Matching Contributions</u>. Effective January 1, 2021, a Participant's vested interest in his or her Matching Contribution Account shall be determined under the following schedule:

Years of Service	Vested Interest
Less than two (2) Years of Vesting Service	0%
Two (2) or more Years of Vesting Service	100%

(e) <u>Top Heavy Plan</u>. For each Plan Year in which the Plan is considered top heavy under Section 12, the schedule in Section 12.4 will be substituted for the schedules set forth in this section if the Section 12.4 schedule would result in a greater vesting percentage.

8.1.3 <u>Participant Election if Vesting Schedule Amended</u>. If the Plan's vesting schedule is amended, whether by this amended and restated Plan, or by subsequent amendment, or the Plan is amended in any way that directly or indirectly affects the computation of the nonforfeitable percentage of a Participant's account, or if the Plan is deemed amended by an automatic change to or from a top heavy vesting schedule under Section 12, each Participant with at least three (3) Years of Service with the Employer may elect, within the period described below, to have the nonforfeitable percentage computed under the Plan without regard to such amendment or change. Notwithstanding the foregoing, for Participants who do not have at least one (1) Hour of Service in any Plan Year beginning after December 31, 1988, the preceding sentence shall be applied by substituting "five (5) Years of Service" for "three (3) Years of Service" above.

The period during which the election of the prior vesting schedule may be made shall commence with the date the amendment is adopted or deemed to be made and shall end on the latest of:

- (a) Sixty (60) days after the amendment is adopted;
- (b) Sixty (60) days after the amendment becomes effective; or

- (c) Sixty (60) days after the Participant is issued written notice of the amendment by the Employer or Administrator.
- 8.2 <u>Forfeiture of Benefits for Certain Causes</u>. Notwithstanding any other provisions of this Plan to the contrary, the right of any Participant or former Participant to receive or to have paid to any other person and the right of any such other person to receive Employer Profit Sharing or Employer Matching Contributions hereunder shall terminate and shall be forever forfeited if such Participant's employment with the Employer is severed because of his or her fraud, embezzlement or dishonesty or any willful act which injures the Employer or the Employee's fellow workers. This section shall be inapplicable as of the earliest of the following dates:
- 8.2.1 the date the Participant meets the requirements for normal retirement benefits under Section 9.1;the date the Participant completes three (3) Years of Service (five (5) Years of Service with respect to Employer Profit Sharing Contributions made before January 1, 2007);
 - 8.2.2 the date the Plan terminates; or
 - 8.2.3 the date contributions to the Plan have been completely discontinued.

Notwithstanding the provisions of Section 8.2 if the Plan becomes a top heavy plan as defined in Section 12.2, only that portion of a Participant's account which is not vested under the vesting schedule set forth at Section 12.4 of this Plan shall be subject to forfeiture.

- 8.3 <u>Forfeiture of Nonvested Portion of Account</u>. Except as provided herein, if a Participant severs employment with an Employer, the nonvested portion of the severed Participant's account will be forfeited at the earlier of:
 - (a) the date the entire vested portion of the Participant's account is distributed; or
- (b) the date on which the Participant completes five (5) consecutive one-year Breaks in Vesting Service.

The forfeited amount will initially be held in the "forfeiture suspense account" referred to in Section 6.5. For purposes of this section, if the vested value or vested percentage of a Participant's account balance is zero (0) on the date of such Participant's severance of employment, the Participant shall be deemed to have received a distribution of his or her entire account on the date of severance.

8.4 <u>Reinstatement of Nonvested Portion of Account</u>. If a Participant who ceases to be an Employee is subsequently reemployed as an Employee before incurring five (5) consecutive one-year Breaks in Vesting Service, any amount forfeited pursuant to Section 8.3(a) shall be restored to the Participant's account, and Years of Service before the Break in Vesting Service will be taken into account to determine the vesting percentage in the reinstated benefit; provided that the Participant repays the amount previously distributed before the earlier of (1) the fifth (5th) anniversary of his or her

reemployment date or (2) the close of the first period of five (5) consecutive one-year Breaks in Vesting Service following the withdrawal.

- 8.5 Service After Severance from Employment.
 - 8.5.1 Service After a Break in Vesting Service.
- (a) <u>Account Before the Break</u>. Years of Service after five (5) consecutive one-year Breaks in Vesting Service shall not increase the Participant's vested interest in his or her account after the Break in Vesting Service.
- (b) <u>Account After the Break</u>. Service prior to a Break in Vesting Service will be taken into account in determining a Participant's vested interest in his or her account after the Break in Vesting Service.
- (1) <u>General Crediting Rule</u>. Upon reemployment, the Participant shall be credited with all Years of Service, including Years of Service prior to the Break in Vesting Service which have not been forfeited under (b)(2) below, in determining such Participant's vested interest in that portion of the Participant's account balance attributable to contributions, earnings and losses after the Break in Vesting Service. This Section 8.5.1(b)(1) shall apply to any Participant who, at the time of severance of employment, either was vested in his or her Employer Profit Sharing or Employer Matching Contribution Accounts or had an account that was subject to the limitations of Section 9.8 (i.e., an Elective Deferral Account or a QNEC Account).
- (2) Exclusion of Forfeited Service. This provision applies to a Participant who experiences a Break in Vesting Service prior to acquiring a nonforfeitable interest under the Plan, and who subsequently is reemployed by an Employer. This paragraph does not apply to a Participant who at the time of severance of employment had an Elective Deferral Account or a QNEC Account because such a Participant is deemed to have acquired a nonforfeitable interest under the Plan for purposes of this Section 8.5.1(b)(2). If this paragraph applies to a Participant and the Participant's number of consecutive one (1) year Breaks in Vesting Service equals or exceeds the greater of (i) five (5), or (ii) the aggregate number of his or her Years of Service, whether or not consecutive, completed prior to such Break in Vesting Service (other than Years of Service which may be disregarded on account of a prior Break in Vesting Service), Years of Service before the Break in Vesting Service shall not be counted for the purpose of determining the vested percentage of the Participant's account balance derived from Employer contributions to the Plan on the Participant's behalf after such Break in Vesting Service.
- 8.5.2 <u>Return to Service Before a Break in Vesting Service</u>. If a Participant severs employment with an Employer and subsequently returns to the service of an Employer without having incurred a Break in Vesting Service, those Years of Service prior to a Participant's severance from employment shall be credited on behalf of the Participant in determining the Participant's vested interest under the Plan.

Page 32 of 65 NORDSTROM 401(k) PLAN

- 8.5.3 <u>Prior Plan Forfeitures and Exclusions</u>. Notwithstanding anything to the contrary in this Section 8, Years of Service permitted to be disregarded under the terms of a prior version of the Plan while then in effect, shall continue to be disregarded under the terms of this Plan.
- 8.6 <u>Forfeiture Reallocation</u>. Notwithstanding anything herein to the contrary, in the event of a severance for cause pursuant to Section 8.2, the Participant's entire account shall be forfeited immediately and allocated as of the next valuation date as provided in Section 6.5. In the event of severance other than pursuant to Section 8.2, the forfeited portion of a Participant's account shall be allocated as of the Anniversary Date next following the date of severance, as provided in Section 6.5.
- 8.7 <u>Maternity/Paternity/Family Absences</u>. If a Participant is absent from employment due to a "qualified family absence" then the Participant will be credited with certain Hours of Service on account of such absence to the extent necessary, and only to the extent necessary, to avoid a Break in Vesting Service and a Permanent Break in Eligibility Service. The term "qualified family absence" shall mean absence (a) by reason of a Participant's pregnancy, (b) by reason of the birth of a child to the Participant, (c) by reason of the placement of a child for adoption by the Participant, (d) for purposes of caring for a child during a period immediately following the birth or placement by adoption of that child, or (e) by reason of circumstances which qualify as family leave under the Family and Medical Leave Act of 1993 (PL 1033) and the regulations thereunder. All absences on account of a single child shall be aggregated and treated as a single absence. The Participant shall be credited with the number of Hours of Service equal to the Hours of Service that the Participant would have been credited on account of the normal work schedule of such Participant prior to the absence, but in no event will Participant be credited with more than five hundred one (501) hours on account of a single absence. In the event that a Participant's normal working schedule is unknown or cannot be determined, Participant shall be credited with ten (10) Hours of Service for each day of the calendar week, including weekends, but excluding Nordstrom holidays.
- 8.7.1 <u>Hours of Service</u>. Hours of Service on account of a qualified absence shall be credited to the Participant during the Plan Year in which the absence begins if necessary to prevent a Break in Vesting Service in that Plan Year, or if such hours are not necessary to prevent a Break in Vesting Service in that Plan Year, such hours shall be credited to the Participant in the next succeeding Plan Year. Hours of Service on account of a qualified absence shall also be credited if necessary to prevent a Permanent Break in Eligibility Service.
- 8.7.2 <u>Uniform Rules</u>. The Plan Administrator shall adopt uniform and reasonable rules for verification of the purpose of absences as well as determination of the number of days for which there was such an absence. Failure of a Participant to submit appropriate documentation in a timely manner pursuant to such rules will result in no credit being given for the period of the absence.

Section 9. ELIGIBILITY TO RECEIVE BENEFITS

9.1 <u>Normal Retirement Benefits</u>. A Participant shall be eligible for normal retirement benefits upon attaining age sixty (60), which is the Normal Retirement Date under the Plan. Distribution of benefits on severance of employment at or after the Normal Retirement Date shall be made in accordance with the provisions of Section 10. Any Participant who continues to work for an Employer

Page 33 of 65 NORDSTROM 401(k) PLAN

after the Normal Retirement Date may, pursuant to Section 10.3, direct the Administrator to defer distribution of the Participant's account until after the Participant's actual severance of employment.

- 9.2 <u>Disability Benefits</u>. Upon a Participant's Disability, prior to his or her Normal Retirement Date or other severance of employment, the Participant shall be entitled to a distribution of benefits hereunder upon written notification to the Administrator and verification of the Participant's Disability by the Administrator. All amounts credited to a Participant's account shall become fully vested upon the Participant's Disability prior to his or her Normal Retirement Date or other severance of employment. Distribution of benefits on account of Disability shall be made in accordance with the provisions of Section 10.
- 9.3 <u>Death Benefits</u>. Upon a Participant's death before his or her Normal Retirement Date, or other severance of employment, the Participant's designated beneficiary shall be entitled to a distribution of benefits hereunder upon written notification to the Administrator and verification of the Participant's death by the Administrator. All amounts credited to a Participant's account shall become fully vested upon the Participant's death prior to his or her Normal Retirement Date or other severance of employment. Distribution of benefits on account of Participant's death shall be made to the surviving beneficiary or beneficiaries designated by the Participant or determined as provided herein, in accordance with the provisions of Section 10.
- 9.3.1 <u>Designation of Beneficiary</u>. At the time of hire an Employee may designate the beneficiary of any benefits which may become payable to a beneficiary of a deceased Participant in this Plan in accordance with procedures adopted by the Plan Administrator. Any such beneficiary designation may be revoked or changed by a subsequent designation made in accordance with those procedures. If the Participant is married and the designated beneficiary is not the Participant's Spouse, the Spouse must consent to the designation by a signed writing witnessed by a representative of the Plan or notarized by a notary public. No beneficiary designation or revocation or change thereof shall be effective until such writing is furnished to the Administrator or its agent. The revocation of a beneficiary designation shall not require the consent of any beneficiary. Any designation filed on a later date shall be deemed to entirely revoke any designation filed on an earlier date unless otherwise expressly stated in the later designation. The Plan Administrator may adopt a policy that applies to distributions to minor beneficiaries and if such a policy is adopted, distributions to minor beneficiaries will be governed by such policy.
- 9.3.2 Effect of Divorce. If a Participant and his or her named beneficiary are or become married and thereafter their marriage is dissolved by entry of a decree of dissolution or other court order having the effect of dissolving the marriage, then such pre-divorce beneficiary designation shall be deemed automatically revoked as to such beneficiary Spouse as of the date of such dissolution unless the death benefit rights of such former Spouse are subsequently reaffirmed by a qualified domestic relations order or the Participant's subsequent written designation. However, distribution of a deceased Participant's account in accordance with his or her latest beneficiary designation filed with the Administrator shall completely discharge the Employer, the Administrator and the Trustee and they shall have no duty to inquire into, or act on any information concerning, whether a Participant's marriage has been dissolved and his or her beneficiary designation thereby revoked as to his or her Spouse.

Page 34 of 65 NORDSTROM 401(k) PLAN

- 9.3.3 <u>Alternate Payee</u>. For purposes of this Section 9.3, an alternate payee named in a qualified domestic relations order shall be treated as a Participant.
- 9.3.4 <u>Deemed Beneficiary</u>. If no beneficiary designation has been made, or if the designated beneficiary has predeceased the Participant, then the Participant will be deemed to have designated the following as his or her surviving beneficiaries and contingent beneficiaries with priority in the order named below:
- (a) <u>Death Before January 1, 2011</u>. If the Participant died before January 1, 2011, then in accordance with the provisions of the Plan document then in effect;
 - (b) <u>Death On or After January 1, 2011</u>. If the Participant dies on or after January 1, 2011:
 - (A) first, to the Participant's surviving Spouse, as defined under federal law; or
 - (B) if the Participant does not have a surviving Spouse, to his or her estate.
- 9.3.5 <u>Surviving Beneficiary</u>. For purposes of determining the appropriate named or deemed beneficiary or contingent beneficiary, an individual is considered to survive the Participant if that individual is alive seven days after the date of the Participant's death.
- 9.4 <u>Benefits on Severance from Employment</u>. Upon the severance of a Participant's employment with the Employer prior to his or her death, Disability or Retirement, the Participant shall be entitled to distribution of his or her vested account balance. Distribution of benefits on account of a Participant's severance from employment with the Employer as provided herein shall be made to the Participant in accordance with the provisions of Section 10. A change in employment from Employee to Leased Employee status shall not be considered a severance from employment for purposes of this Section 9.4.
- 9.5 Accelerated Benefit Option. A terminally ill Participant, as defined herein, shall be entitled to an early distribution of a portion of his or her benefits upon written notification to the Administrator and verification of the Participant's terminal illness by the Administrator. The Participant entitled to receive a distribution pursuant to this accelerated benefit option may receive a distribution of any Rollover Account, in addition to any vested Employer Profit Sharing Contributions and Employer Matching Contributions, including earnings thereon. Distribution of benefits on account of terminal illness shall be made in accordance with the provisions of Section 10.6 as an in-service withdrawal, but without the requirement that the Participant have attained age fifty-nine and one-half (59½). A Participant shall be deemed to be terminally ill when, by reason of a medically determinable physical condition, the Participant's life expectancy is less than twenty-four (24) months. The Participant's terminally ill condition and probable life expectancy must be certified by a physician acceptable to both the Participant and the Administrator.

Page 35 of 65 NORDSTROM 401(k) PLAN

- 9.6 <u>In-Service Withdrawals</u>. A Participant who continues working after attaining age fifty-nine and one-half (59½) may elect partial in-service withdrawals in accordance with Section 10.6.
- 9.7 <u>Hardship Withdrawals</u>. At the direction of the Administrator and in accordance with uniform rules consistently applied, the Administrator may direct the Trustee to distribute a Participant's Rollover Account, Elective Deferral Contributions and Employer Profit Sharing Contributions to the Participant in the case of "hardship" pursuant to Sections 9.7.1 to 9.7.6 below. Excepting Participants who receive a hardship distribution under Section 9.7.2(g).
 - 9.7.1 Maximum Amount. Withdrawals shall not exceed the lesser of:
- (a) the value of the Participant's Rollover Account, Elective Deferral Account and Employer Profit Sharing Contributions Account on the date the withdrawal is processed; or
- (b) the sum of the Participant's total Elective Deferral Contributions to the Plan, excluding any investment earnings, and the combined value of the Participant's Employer Profit Sharing Contributions Account and Rollover Account on the date the withdrawal is processed.
- 9.7.2 <u>Hardship</u>. The term "hardship" as used herein shall mean an immediate and heavy financial need resulting from any one or more of (a) through (g), below:
- (a) uninsured expenses for (or necessary to obtain) medical or dental care that would be deductible under Code § 213(d) (determined without regard to whether the expenses exceed 7.5% of the Participant's adjusted gross income) incurred or to be incurred by the Participant or the Participant's Spouse or dependents (where a Participant's dependents include Participant's noncustodial children who are treated as dependents pursuant to Code § 213(d)(5), provided however that expenses with respect to any such noncustodial children exclude nonprescription drugs or medicine, other than insulin);
- (b) costs directly related to the purchase (excluding mortgage payments) of a principal residence for the Participant;
- (c) payment of tuition, related educational fees, and room and board expenses for up to the next twelve (12) months of post-secondary education for the Participant or the Participant's Spouse, children or dependents;
- (d) payments necessary to prevent the eviction of Participant from his or her principal residence or to prevent foreclosure on the mortgage of Participant's principal residence;
- (e) payments for burial or funeral expenses for the Participant's deceased parent, Spouse, children or dependents;
- (f) uninsured expenses for the repair of damage to the Participant's principal residence that would qualify for the casualty deduction under Code § 165 (determined without regard to whether the loss exceeds 10% of the Participant's adjusted gross income); or

Page 36 of 65 NORDSTROM 401(k) PLAN

(g) any other financial need determined to be immediate and heavy under rules and regulations issued by the Secretary of the Treasury or his delegate; provided, however, that any such financial need shall constitute an immediate and heavy need under this paragraph (g) no sooner than administratively practicable following the date such rule or regulation is issued.

In addition, a hardship withdrawal may include an amount necessary to satisfy any tax obligation which becomes payable on account of a distribution for any hardship described in (a) through (g), above. For purposes of this Section 9.7.2, the term "dependents" shall have the meaning prescribed under Code § 152, without regard to subsections (b)(1), (b)(2) and (d)(1)(B).

- 9.7.3 <u>Representation that Distribution is Necessary to Satisfy Financial Need</u>. A distribution under Section 9.7 can only be made to the extent it is necessary to satisfy an immediate and heavy financial need.
 - (a) A distribution is necessary to satisfy an immediate and heavy financial need only to the extent that:
- (1) the amount of the distribution is not in excess of the amount required to satisfy the financial need; and
- (2) the financial need cannot be satisfied from other resources reasonably available to the Participant, as determined by the Administrator on the basis of all relevant facts and circumstances.
- (b) Excepting hardship withdrawals made under Section 9.7.2(g), the Administrator shall require the Participant to provide written certification of the facts and circumstances establishing that Participant has met one of the hardship categories and may consider other relevant evidence. Such written certification shall require the Participant to represent that the financial need cannot reasonably be relieved: (1) through reimbursement or compensation by insurance or otherwise; (2) by liquidation of the Participant's assets; (3) by cessation of Elective Deferrals under the Plan; (4) by other currently available distributions and nontaxable loans under the Plan and under any other plan maintained by the Employer or by any other employer; or (5) by borrowing from commercial sources on reasonable commercial terms in an amount sufficient to satisfy the need. A Participant's need cannot reasonably be relieved by taking one of the above actions (1) through (5) if the effect would be to increase the amount of the need.
- (c) For purposes of (a)(2), the Administrator is entitled to rely on the Participant's representation made pursuant to (b), unless the Administrator has actual knowledge to the contrary.
 - 9.7.4 Fee. The Administrator may charge a reasonable fee for processing hardship withdrawals.
- 9.7.5 <u>Valuation</u>. In the event a hardship withdrawal is made by a Participant other than at a regular Valuation Date, the allocation of investment gains and losses to the account shall be made as if such withdrawal had occurred on the preceding Valuation Date, and no gains or losses

Page 37 of 65 NORDSTROM 401(k) PLAN

allocable to the withdrawn funds shall be credited therefor, except that a Participant's Nordstrom Stock Account, if any, shall be valued to the date of withdrawal.

- 9.7.6 Ordering Rule. Excepting hardship withdrawals made under Section 9.7.2(g), hardship withdrawals are not available unless and until the Participant has first exhausted all other sources of funds to satisfy the hardship, including but not limited to Participant loans available from this Plan. Hardship withdrawals will be made from all available sources in accordance with procedures adopted by the Plan Administrator.
- 9.8 <u>Restriction on Distributions of Elective Deferrals</u>. Amounts attributable to Elective Deferral Contributions and QNECs under this Plan may not be distributed prior to the occurrence of one of the following events: severance of employment with all Employers, the Participant's death or Disability, the Participant's attaining age fifty-nine and one-half (59½), or the Participant's establishment of a hardship under Section 9.7.
- 9.9 <u>Military Service Distributions</u>. Effective January 1, 2009, an Employee who is on a leave of absence for active military service for more than 30 days shall be treated as having incurred a severance from employment under Section 9.4 and, consequently, may request and receive a distribution of any amount which is attributable to Employee Contributions under Section 5.2 or 5.3 of the Plan.

Section 10. METHOD OF PAYMENT OF BENEFITS

10.1 <u>Distribution of Benefits</u>.

10.1.1 <u>Lump Sum Payment</u>. Upon the occurrence of any of the events specified in Section 9 requiring or permitting a distribution of benefits to a Participant or his or her beneficiary, the Administrator shall instruct the Trustee to distribute benefits, determined in accordance with Section 10.2, below, in a single lump sum payment unless the Trustee receives a timely election for a different form of benefit. If the present value of a Participant's benefit exceeds \$1,000 and the benefit is Immediately Distributable (see Section 10.1.3), the Administrator must obtain the consent of the Participant for the distribution. For purposes of applying the \$1,000 threshold in the preceding sentence, the Plan shall disregard the balance in a Participant's rollover account for distributions from this Plan made after March 28, 2005 and before April 1, 2010, and shall include the balance in a Participant's rollover account for distributions from this Plan made on or after April 1, 2010. Participant consent may be in writing or in accordance with procedures adopted by the Plan Administrator.

10.1.2 <u>Consent to Distribution</u>. The consent of the Participant shall be obtained in writing within the one hundred eighty (180) day period ending on the annuity starting date. The annuity starting date is the first day of the first period for which an amount is paid as an annuity or any other form. The Administrator shall notify the Participant of the right to defer any distribution until the benefit is no longer Immediately Distributable and shall explain any optional form of benefit under the Plan. The consent of the Participant shall not be required to the extent that a distribution is permitted to be made without consent (under Section 10.1.1) or required to be made to satisfy Code §§ 401(a)(9) or 415. In addition, upon termination of this Plan if the Plan does not offer an annuity option, the Participant's account balance may, without the Participant's consent, be distributed to the Participant or

transferred to another defined contribution plan (other than an employee stock ownership plan as defined in Code § 4975(e)(7)) within the same controlled group.

- 10.1.3 <u>Immediately Distributable</u>. An account balance is immediately distributable upon occurrence of a distribution event under Section 9 prior to the time the Participant attains the later of age 62 or the Normal Retirement Date under Section 9.1.
- 10.1.4 <u>Spousal Consent</u>. Effective for distributions on and after January 1, 2010, spousal consent is not required for distributions from the Plan. For distributions prior to January 1, 2010, the spousal consent provisions of the Plan in effect at the time of distribution apply.
- 10.1.5 <u>Social Security Not Relevant</u>. Notwithstanding any other provisions of this Plan, any benefits payable under this Plan shall not be decreased by reason of any increase in the benefit levels payable under Title II of the Social Security Act or any increase in the wage base under Title II.
- 10.2 <u>Valuation of Account</u>. The benefit payable to a Participant or his or her beneficiary in accordance with Section 9 shall be determined as of the Valuation Date immediately preceding the date of distribution. Contributions allocated to a Participant under Sections Section 5 and Section 6, but which have not yet been deposited to the Participant's account as of the date of distribution, shall not be payable to such Participant until such contributions have actually been deposited.

10.3 Time of Distribution.

- 10.3.1 <u>General Rule</u>. Subject to the consent requirements of Section 10.1, the benefit payable to a Participant or beneficiary shall be made as soon as administratively practicable following the occurrence of a distribution event described in Section 9 and, if applicable, such Participant's request and consent to such distribution.
- 10.1.2 <u>Statutory Deadlines</u>. Unless the Participant otherwise elects in writing, payments hereunder must begin not later than sixty (60) days after (a) or (b):
- (a) The end of the Plan Year in which the Participant (1) attains age sixty (60), (2) reaches the tenth anniversary of the date he or she commenced participation in the Plan, or (3) severs employment, whichever of (1), (2) or (3) is latest; or
- (b) If the Trustee or Administrator requires information which is not available before that latest date under (a), the payments shall begin no later than sixty (60) days after that information is supplied.

10.3.3 Election to Defer Payment.

(a) <u>Written Election</u>. A Participant may elect in writing that a payment to him or her of any benefit under this Plan will commence at a date later than the date specified under Sections 10.3.1 and 10.3.2 above. Any such election shall be signed by the Participant and shall state the date payments are to commence. In any event, a Participant making such election shall be required to commence the receipt of his or her retirement benefit no later than the Participant's required beginning date under Section 11.

(b) <u>Deemed Election</u>. Notwithstanding the foregoing, the failure of a Participant to consent to a distribution while a benefit is Immediately Distributable, within the meaning of Section 10.1.3 of the Plan, shall be deemed to be an election to defer commencement of payment of any benefit sufficient to satisfy this section.

10.4 Form of Payment.

- 10.4.1 <u>Cash Payment</u>. Except as provided in Section 10.4.2, all distributions from the Plan shall be made in the form of cash.
- 10.4.2 <u>In-Kind Distributions</u>. In the following circumstances, the Plan shall make an in-kind distribution of benefits.
- (a) <u>Company Stock</u>. If the Participant's Plan account holds Company stock and the Participant or beneficiary requests an in-kind distribution of the shares, the Administrator shall instruct the Trustee to distribute the shares in lieu of their cash equivalent, in a manner that is consistent with the rules set forth in 1.02.2(f) of the Nordstrom 401(k) Plan Participant Direction Investment Appendix.
- (b) <u>Non-Marketable Security</u>. If the Participant's Plan account holds a security that is not publicly traded on an established securities market (i.e., a non-publicly traded security) at the time that the Participant or beneficiary requests a distribution, the Administrator shall instruct the Trustee to make an in-kind distribution of such non-publicly traded security, in lieu of cash.
- 10.5 <u>Qualified Domestic Relations Orders</u>. Subject to the procedures established by the Administrator under Section 14.3, benefits may be paid from the nonforfeitable balance of a Participant's accounts in accordance with a qualified domestic relations order ("QDRO") as defined in Code § 414(p) without regard to whether the Participant has attained the "earliest retirement age," as defined in Code § 414(p).
- 10.6 <u>Partial Withdrawals</u>. A Participant who is entitled to a distribution under Section 9 may elect partial withdrawals of his or her vested account balance in lieu of a lump sum distribution of his or her entire vested account balance. Partial withdrawals are subject to the consent requirements of Section 10.1 and may be subject to a reasonable administrative fee.

10.7 Rollovers.

- 10.7.1 <u>Direct Rollover Election</u>. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this section, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an eligible rollover distribution paid directly to an eligible retirement plan specified by the distributee in a direct rollover.
 - 10.7.2 <u>Definitions</u>. For purposes of this section, certain terms shall be defined as follows:
 - (a) Eligible Rollover Distribution.

Page **40** of **65** NORDSTROM 401(k) PLAN

(1) General Rule. An eligible rollover distribution is any distribution of all or any portion of the
balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of
a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the
distributee and the distributee's designated beneficiary, or for a specified period of ten (10) years or more; any distribution to the
extent such distribution is required under Code § 401(a)(9); any withdrawal on account of hardship; or, except as provided in (2),
the portion of any distribution that is not includible in the distributee's gross income (disregarding these rollover rules).

- (2) <u>Special Rule for After-tax Amounts</u>. A portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of after-tax employee contributions that are not includible in gross income upon distribution. However, any such portion shall be distributed only to an individual retirement account or annuity described in Code § 408(a) or (b), or to a qualified defined contribution plan described in Code § 401(a) or § 403(a) that agrees to separately account for transferred amounts, including separately accounting for the portion that is includible in gross income and the portion that is not includible in gross income.
- (b) <u>Eligible Retirement Plan</u>. An eligible retirement plan is one of the following that accepts the Participant's eligible rollover distribution:
 - (1) an individual retirement account described in Code § 408(a);
 - (2) an individual retirement annuity described in Code § 408(b);
 - (3) an annuity plan described in Code § 403(a);
 - (4) an annuity contract described in Code § 403(b);
 - (5) a qualified trust described in Code § 401(a);
- (6) an eligible deferred compensation plan described in Code § 457(b) that is maintained by a state, a political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and that agrees to separately account for amounts transferred into such plan from this Plan; or
- (7) for eligible rollover distributions after December 31, 2007, a Roth individual retirement account described in Code § 408A.

This definition of an "eligible retirement plan" also applies in the case of a distribution to a Participant's surviving Spouse, or to a Spouse or former Spouse who is the alternate payee under a qualified domestic relations order, as defined in Section 10.5. For a non-Spouse designated beneficiary, an "eligible retirement plan" means only an individual retirement account under Code § 408(a) or an individual retirement annuity under Code § 408(b) and only if the transfer is a direct rollover.

(c) <u>Distributee</u>: A distributee includes an employee or former employee. In addition, the employee's or former employee's surviving Spouse and the employee's or former

Page **41** of **65** NORDSTROM 401(k) PLAN

employee's Spouse or former Spouse who is the alternate payee under a qualified domestic relations order, as defined in Code § 414(p), are distributees with regard to the interest of the Spouse or former Spouse. A distributee includes the Participant's non-Spouse designated beneficiary.

- (d) <u>Direct Rollover</u>: A direct rollover is a payment by the Plan to the eligible retirement plan specified by the distributee.
- 10.7.3 <u>Rollover of Designated Roth Contributions</u>. A direct rollover of a distribution from a Participant's account attributable to designated Roth contributions under the Plan can be made only to a designated Roth contributions account under an applicable retirement plan described in Code § 402A(e)(1) or to a Roth IRA described in Code § 408A, and only to the extent the rollover is permitted under the rules of Code § 402(c).

10.8 Administration of Unclaimed Benefits.

- 10.8.1 <u>Forfeiture of Unclaimed Benefits</u>. The Plan Administrator shall make reasonable efforts to locate Participants or beneficiaries and notify them of any vested balances. If after making such efforts, the Participant or beneficiary's benefits are unclaimed, the benefit shall be forfeited in accordance with procedures adopted by the Plan Administrator and shall hold the forfeited benefit in the forfeiture suspense account under Section 6.5, subject to restoration under Section 10.8.2, below.
- 10.8.2 Restoration of Unclaimed Benefits. If a Participant or beneficiary whose benefit has been forfeited under Section 10.8.1 above thereafter is located and requests payment of such benefits, and if the Plan has not terminated (or if the Plan has been terminated, all benefits have not yet been distributed), then the benefit of such Participant or beneficiary shall be restored, without any adjustment for investment earnings through the restoration date. The Administrator shall restore the benefit using the forfeiture suspense account pursuant to Section 6.5.3. However, if any such unclaimed benefit has not been restored by the time the Plan terminates and all benefits are distributed, the forfeiture of such unclaimed benefit will be irrevocable.

Section 11. MINIMUM DISTRIBUTION REQUIREMENTS

11.1 General Rules.

- 11.1.1 <u>Effective Date</u>. The provisions of this Section will apply for purposes of determining required minimum distributions for calendar years beginning on and after January 1, 2003.
- 11.1.2 <u>Precedence</u>. The requirements of this Section will take precedence over any inconsistent provisions of the Plan.
- 11.1.3 <u>Requirements of Treasury Regulations Incorporated</u>. All distributions required under this Section will be determined and made in accordance with the Treasury regulations under Code § 401(a)(9).
- 11.1.4 <u>TEFRA Section 242(b)(2) Elections</u>. Notwithstanding the other provisions of this Section, distributions may be made under a designation made before January 1, 1984, in accordance

with Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the plan that relate to Section 242(b)(2) of TEFRA.

11.2 Time and Manner of Distribution.

- 11.2.1 <u>Required Beginning Date</u>. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's required beginning date, as defined in Section 11.5.5.
- 11.2.2 <u>Death of Participant Before Distributions Begin</u>. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
- (a) <u>Surviving Spouse Beneficiary</u>. If the Participant's surviving Spouse is the Participant's sole designated beneficiary, then distributions to the surviving Spouse must begin by December 31 of the calendar year immediately following the calendar year in which the Participant dies, or by December 31 of the calendar year in which the Participant would have attained age 70½, if later.
- (b) <u>Non-Spouse Beneficiary</u>. If the Participant's surviving Spouse is not the Participant's sole designated beneficiary, then distributions to the designated beneficiary must begin by December 31 of the calendar year immediately following the calendar year in which the Participant dies.
- (c) <u>Absence of Beneficiary</u>. If there is no designated beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (d) <u>Death of Surviving Spouse</u>. If the Participant's surviving Spouse is the Participant's sole designated beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this Section 11.2.2, other than Section 11.2.2(a), will apply as if the surviving Spouse were the Participant.

For purposes of this Section 11.2 and Section 11.4, unless Section 11.2.2(d) applies, distributions are considered to begin on the Participant's required beginning date. If Section 11.2.2(d) applies, distributions are considered to begin on the date distributions are required to begin to the surviving Spouse under Section 11.2.2(a). If distributions under an annuity purchased from an insurance company irrevocably commence to the Participant before the Participant's required beginning date (or to the Participant's surviving Spouse before the date distributions are required to begin to the surviving Spouse under Section 11.2.2(a)), the date distributions are considered to begin is the date distributions actually commence.

11.2.3 Forms of Distribution. Unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the required beginning date, as of the first distribution calendar year distributions will be made in accordance with Sections 11.3 and 11.4 of this Section. If the Participant's interest is distributed in the form of an annuity

purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code § 401(a)(9) and the Treasury regulations.

11.3 Required Minimum Distributions During Participant's Lifetime.

- 11.3.1 <u>Amount of Required Minimum Distribution For Each Distribution Calendar Year</u>. During the Participant's lifetime, the minimum amount that will be distributed for each distribution calendar year is the lesser of:
- (a) <u>General Rule</u>. The quotient obtained by dividing the Participant's account balance by the distribution period in the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's age as of the Participant's birthday in the distribution calendar year; or
- (b) <u>Surviving Spouse</u>. If the Participant's sole designated beneficiary for the distribution calendar year is the Participant's Spouse, the quotient obtained by dividing the Participant's account balance by the number in the Joint and Last Survivor Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the distribution calendar year.
- 11.3.2 <u>Lifetime Required Minimum Distributions Continue Through Year of Participant's Death.</u> Required minimum distributions will be determined under this Section 11.3 beginning with the first distribution calendar year and up to and including the distribution calendar year that includes the Participant's date of death.

11.4 Required Minimum Distributions After Participant's Death.

11.4.1 Death On or After Date Distributions Begin.

- (a) <u>Participant Survived by Designated Beneficiary</u>. If the Participant dies on or after the date required minimum distributions begin and there is a designated beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's account balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's designated beneficiary, determined as follows:
- (1) The Participant's remaining life expectancy is calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- (2) If the Participant's surviving Spouse is the Participant's sole designated beneficiary, the remaining life expectancy of the surviving Spouse is calculated for each distribution calendar year after the year of the Participant's death using the surviving Spouse's age as of the Spouse's birthday in that year. For distribution calendar years after the year of the surviving Spouse's death, the remaining life expectancy of the surviving Spouse is calculated using the age of the surviving Spouse as of the Spouse's birthday in the calendar year of the Spouse's death, reduced by one for each subsequent calendar year.

- (3) If the Participant's surviving Spouse is not the Participant's sole designated beneficiary, the designated beneficiary's remaining life expectancy is calculated using the age of the beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
- (b) <u>No Designated Beneficiary</u>. If the Participant dies on or after the date required minimum distributions begin and there is no designated beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's account balance by the Participant's remaining life expectancy calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.

11.4.2 Death Before Date Distributions Begin.

- (a) <u>Participant Survived by Designated Beneficiary</u>. If the Participant dies before the date required minimum distributions begin and there is a designated beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's account balance by the remaining life expectancy of the Participant's designated beneficiary, determined as provided in Section 11.4.1.
- (b) <u>No Designated Beneficiary</u>. If the Participant dies before the date required minimum distributions begin and there is no designated beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (c) <u>Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin</u>. If the Participant dies before the date required minimum distributions begin, the Participant's surviving Spouse is the Participant's sole designated beneficiary, and the surviving Spouse dies before distributions are required to begin to the surviving Spouse under Section 11.2.2(a), this Section 11.4.2 will apply as if the surviving Spouse were the Participant.

11.5 Definitions.

- 11.5.1 <u>Designated beneficiary</u>. The individual who is designated as the beneficiary under Section 9.3 of the Plan is the designated beneficiary under Code § 401(a)(9) and section 1.401(a)(9)-1, Q&A-4 of the Treasury regulations. For purposes of this Section 11, if the Participant has not designated a beneficiary or if a Participant-designated beneficiary does not survive the Participant, then the designated beneficiary shall be determined under the priority rules set forth in Section 9.3.4(a) through (d). If there is more than one individual within the highest priority class under Section 9.3.4(a) through (d), the individual with the shortest life expectancy will be the designated beneficiary for purposes of Section 11.
- 11.5.2 <u>Distribution calendar year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year

Page 45 of 65 NORDSTROM 401(k) PLAN

is the calendar year immediately preceding the calendar year that contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which distributions are required to begin under Section 11.2. The required minimum distribution for the Participant's first distribution calendar year will be made on or before the Participant's required beginning date. The required minimum distribution for other distribution calendar years, including the required minimum distribution for the distribution calendar year in which the Participant's required beginning date occurs, will be made on or before December 31 of that distribution calendar year.

- 11.5.3 <u>Life expectancy</u>. Life expectancy as computed by use of the Single Life Table in section 1.401(a)(9)-9 of the Treasury regulations.
- 11.5.4 Participant's account balance. The account balance as of the last Valuation Date in the calendar year immediately preceding the distribution calendar year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the account balance as of dates in the valuation calendar year after the Valuation Date and decreased by distributions made in the valuation calendar year after the Valuation Date. The account balance for the valuation calendar year includes any amounts rolled over or transferred to the plan either in the valuation calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.

11.5.5 Required beginning date.

(a) <u>Permissive Rule</u>. Participants who remain Employees, and who are not five percent (5%) owners (described in Section 11.5.5(c)), may elect to continue to treat their required beginning date as the first day of April of the calendar year following the calendar year in which the Participant attains age seventy and one-half ($70\frac{1}{2}$).

(b) Mandatory Rule.

- (1) Non Five Percent (5%) Owner. The required beginning date of a Participant who is not a five percent (5%) owner is the first day of April of the calendar year following the later of the calendar year in which the Participant attains age seventy and one-half ($70\frac{1}{2}$), or the calendar year in which the Participant retires.
- (2) <u>Five Percent (5%) Owner</u>. The required beginning date of a Participant who is a five percent (5%) owner during any year beginning after December 31, 1979, is the first day of April following the later of:
 - (A) the calendar year in which the Participant attains age seventy and one-half (70½); or
- (B) the earlier of the calendar year in which the Participant becomes a five percent (5%) owner, or the calendar year in which the Participant retires.

Page 46 of 65 NORDSTROM 401(k) PLAN

(c) Five Percent (5%) Owner.

(1) <u>Defined</u>. A Participant is treated as a five percent (5%) owner for purposes of this section if such Participant is a five percent (5%) owner as defined in Code § 416(i) (determined in accordance with Code § 416 but without regard to whether the plan is top heavy) at any time during the Plan Year ending with or within the calendar year in which such owner attains age sixty-six and one-half (66½) or any subsequent plan year.

(2) <u>Continued Distribution</u>. Once distributions have begun to a five percent (5%) owner under this section, they must continue to be distributed, even if the Participant ceases to be a five percent (5%) owner in a subsequent year.

<u>11.6 Source of Funds for Required Minimum Distributions</u>. The Plan Administrator or its delegate is authorized to adopt such policies, procedures and protocols as it deems necessary for the efficient administration of the Plan regarding the source of funds used to make required minimum distributions under this Section 11.

Section 12. TOP HEAVY PLANS

- 12.1 Effective Date. This Section shall apply for purposes of determining whether the Plan is a top-heavy plan under Code § 416(g) and whether the Plan satisfies the minimum benefits requirements under Code § 416(c) for Plan Years beginning after December 31, 2001.
- 12.2 <u>Effect of Top Heavy Plan Status</u>. In the event that the Plan is determined to be a "top heavy plan" as defined in Section 12.3, the Plan shall comply with the provisions of Section 12.4, in addition to meeting the requirements set forth elsewhere in this Plan.
- 12.3 <u>Determination of Top Heavy Status</u>. The determination of top heavy status will be made with regard to the following defined terms:
- 12.3.1 <u>Determination Date</u>. The last day of the preceding Plan Year, or, in the case of the first Plan Year, the last day of that Plan Year.
- 12.3.2 <u>Key Employee</u>. An Employee (including a deceased Employee or a beneficiary of such Employee) who at any time during the Plan Year which includes the Determination Date is any of the following (a), (b), or (c):
- (a) Officer. An officer of the Employer (as that term is defined within the meaning of the regulations under Code § 416) whose annual compensation is greater than the dollar amount in effect under Code § 416(i)(1)(A)(i) as adjusted by the Secretary of the Treasury. No more than fifty (50) Employees shall be treated as officers.
- (b) <u>Five Percent (5%) Owner</u>. A "five percent (5%) owner" of the Employer. "Five percent (5%) owner" means any person who owns (or is considered as owning within the meaning of Code § 318) more than five percent (5%) of the outstanding stock of the Employer or stock possessing more than five percent (5%) of the total combined voting power of all stock of the Employer.

Page 47 of 65 NORDSTROM 401(k) PLAN

(c) One Percent (1%) Owner. A "one percent (1%) owner" of the Employer having annual compensation from the Employer of more than \$150,000. "One percent (1%) owner" means any person who owns (or is considered as owning within the meaning of Code § 318) more than one percent (1%) of the outstanding stock of the Employer or stock possessing more than one percent (1%) of the total combined voting power of all stock of the Employer. In determining whether an individual has annual compensation of more than \$150,000, annual compensation from each employer required to be aggregated under Code §§ 414(b), (c) and (m) shall be taken into account.

(d) Determining Ownership.

- (1) <u>Employer</u>. In determining percentage ownership hereunder, employers that would otherwise be aggregated under Code §§ 414(b), (c) and (m) shall be treated as separate employers.
- (2) <u>Attribution of Ownership</u>. In the case of a corporation, for purposes of applying the ownership attribution rules of Code § 318 in determining Key Employee status, subparagraph (C) of Code § 318(a)(2) shall be applied by substituting "five percent (5%)" for "fifty percent (50%)." If an entity is not a corporation, ownership attribution rules shall be applied in accordance with regulations promulgated by the Secretary of the Treasury based upon the principles of § 318(a), as herein revised.
- (e) <u>Annual Compensation</u>. For purposes of this Section 12, the term annual compensation means compensation as defined in Code § 415(c)(3), but including amounts contributed by Employer pursuant to a salary reduction agreement which are excludable from Employee's gross income under Code §§ 125, 132(f), 402(a)(8), 402(h) or 403(b).
- 12.3.3 <u>Non-Key Employee</u>. Any Employee who does not meet the definition of a Key Employee pursuant to Section 12.3.2 above, is a Non-Key Employee.
- 12.3.4 <u>Top Heavy Plan</u>. This Plan will be a top heavy plan if, with respect to the applicable Plan Year, as of the determination date for that year the Plan has a top heavy percentage that exceeds sixty percent (60%).
 - (a) Percentage. "Top heavy percentage" shall be that percentage which equals a fraction,
- (1) The numerator of which is the sum of the present value of accrued benefits of all Key Employees as of the determination date, contributions for all Key Employees which are due but unpaid as of the determination date, and distributions made to Key Employees during the one year period ending on the determination date, and
- (2) The denominator of which is the sum of the present value of accrued benefits for all Employees as of the determination date, total contributions for all Participants due but unpaid as of the determination date, and total distributions made to Participants during the one year period ending on the determination date.

Page 48 of 65 NORDSTROM 401(k) PLAN

- (3) However, in the case of a distribution made for a reason other than severance from employment, death or disability, the above provisions shall be applied by substituting "five year period" for "one year period."
 - (b) Related Rules. For purposes of calculating the top heavy percentage under Section 12.3.4(a):
- (1) <u>Accrued Benefit</u>. The present value of a Participant's accrued benefit shall include: (A) in the case of a defined contribution plan, that Participant's account balance (including Catch-up Contributions Accounts); (B) in the case of a defined benefit plan, the present value of the accrued benefits of such individual determined as of the most recent valuation date which is within the twelve (12) month period ending on the determination date; (C) the accrued benefit attributable to nondeductible employee contributions; and (D) the accrued benefit of a participant other than a Key employee shall be determined under (i) the method, if any, that uniformly applies for accrual purposes under all defined benefit plans maintained by the employer, or (ii) if there is not such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional rule of Code § 411(b)(1)(C).
- (2) <u>Distribution</u>. In considering distributions within the one year period (or five year period, as appropriate) ending on the determination date: (A) all distributions from this Plan and distributions from terminated plans which would have been required to be aggregated had they not been terminated, must be considered; and (B) no benefit attributable to deductible contributions, or to amounts rolled over or transferred to this Plan from the Plan of another employer after December 31, 1983, shall be considered in determining a Participant's accrued benefit.
- (3) <u>Exclusions</u>. The following exclusions shall apply: (A) contributions, accrued benefits, and distributions on behalf of a Non-Key Employee who was formerly a Key Employee shall be disregarded in determining the top heavy percentage; and (B) if a Participant or former Participant has not performed services for any Employer maintaining the Plan at any time during the one year period ending on the determination date, the accrued benefit for such Participant shall not be taken into account in determining top heavy plan status.
- (c) <u>Aggregation With Other Plans</u>. If the Employer or an entity affiliated with the Employer pursuant to Code §§ 414(b), (c) or (m) maintains other qualified plans (including simplified employee pension plans), a plan is a top heavy plan only to the extent that the combined top heavy percentage for the plan and all aggregated plans exceeds sixty percent (60%). For the purpose of making this determination:
- (1) Mandatory Aggregation. All qualified plans of the Employer or an entity affiliated with the Employer pursuant to Code §§ 414(b), (c) or (m) which include one or more Key Employees as Participants, and all qualified plans which must be considered in order for a plan including Key Employee Participants to meet the requirements of Code §§ 401(a)(4) or 410, must be aggregated.

Page 49 of 65 NORDSTROM 401(k) PLAN

(2) <u>Permitted Aggregation</u>. Additional qualified plans of the Employer or an entity affiliated with the Employer pursuant to Code §§ 414(b), (c) or (m), if such plans, when aggregated with this Plan, satisfy the requirements of Code §§ 401(a)(4) and 410, may be aggregated.

(3) <u>Determination Date</u>. Where multiple plans with differing determination dates are to be aggregated for the determination of top heavy status, the top heavy percentage shall be calculated by reference to determination dates for all plans falling within the same calendar year.

12.4 Minimum Employer Contributions to Top Heavy Plans.

12.4.1 Minimum Contribution.

(a) <u>General Rule</u>. Except as provided in Section 12.4.1(c) below, for each Plan Year that this Plan is determined to be a top heavy plan, a Participant who is a Non-Key Employee shall have allocated to his or her account (in either this Plan or another defined contribution plan maintained by an Employer) a contribution equal to the product of that Participant's Compensation, as defined in Section 2.6, and the minimum top heavy contribution rate. The minimum top heavy allocation, if any, required shall not be forfeited under §§ 411(a)(3)(B) or 411(a)(3)(D).

(b) Minimum Top Heavy Contribution Rate. Subject to Section 12.4.1(c) below, the minimum top heavy contribution rate for a Participant who is a Non-Key Employee shall equal the lesser of three percent (3%) of such Non-Key Employee's compensation or the highest contribution rate (excluding Catch-up Contributions for the Plan Year containing the determination date) made to the account of a Key Employee, provided that the contribution rate shall not be less than three percent (3%) if this Plan is required to be aggregated with a defined benefit plan in order for that plan to meet the requirements of Code §§ 401(a) and 410(b). The term "contribution rate" shall mean the percentage derived by dividing a numerator, which is the sum of Employer contributions (including amounts deferred at the Employee's election to a Plan described in Code § 401(k) but excluding contributions to Social Security) and forfeitures allocated to a Participant's account, by a denominator equal to the Participant's Compensation. For the purposes of this Section 12.4, the term "Participant who is a Non-Key Employee" shall include all Non-Key Employees who have become Participants but who have failed to complete one thousand (1,000) Hours of Service during the Plan Year and those Non-Key Employees who would be eligible to participate in the Plan except that their compensation does not exceed a specified minimum level or they have failed to make a mandatory employee contribution or an elective contribution to a plan described in Code § 401(k). Employer Matching Contributions to this Plan shall count toward the minimum top heavy contribution rate (and any reduction to the contribution rate that results will not be taken into account in determining whether the Plan impermissibly conditions benefits on the making of elective deferrals under Code § 401(k)(4)(A)).

(c) Exceptions for Defined Benefit Plan. Notwithstanding Sections 12.4.1(a) and 12.4.1(b):

(1) If a defined benefit pension plan providing benefits for one or more Key Employees is maintained by the Employer, and if such defined benefit pension plan depends upon this Plan to satisfy the nondiscrimination rules of Code § 401(a)(4) or the coverage rules of Code § 410

Page 50 of 65 NORDSTROM 401(k) PLAN

(or if another plan benefiting the Key Employee so depends on such defined benefit plan) the guaranteed minimum top heavy contribution for a Non-Key Employee shall be three percent (3%) of his or her compensation regardless of the contribution rate for the Key Employees.

(2) If in addition to this Plan the Employer maintains a qualified defined benefit pension plan which Provides a minimum benefit to Non-Key Employee Participants pursuant to Code § 416(c)(1), no minimum top heavy employer contribution need be made for such Participants under this Plan.

- 12.4.2 <u>Minimum Top Heavy Contributions and/or Benefits in Multiple Plans</u>. In the event that a Non-Key Employee participates in both this Plan and a defined benefit plan, it shall not be necessary to provide such Non-Key Employee with both a minimum top heavy contribution under this Plan (and other defined contribution plans) and a minimum benefit under the defined benefit plan. The minimum top heavy contribution and minimum benefit requirements with respect to all such Plans shall be deemed satisfied if such Non-Key Employee is provided with the minimum benefit under the defined benefit plan.
- 12.4.3 <u>Make-Up Contribution</u>. If the contribution rate for the Plan Year with respect to a Non-Key Employee is less than the minimum top heavy contribution required, the Employer will increase its contribution for such Employee to the extent necessary to cause the Employee's contribution rate for the Plan Year to equal the required minimum top heavy contribution. The Employer will cause this make-up contribution to be made from Employer's net profits.

12.4.4 <u>Vesting Schedule</u>. For each Plan Year in which this Plan is determined to be top heavy, Employer contributions to the Plan on behalf of Non-Key Employees shall vest under the following schedule if it is more favorable to the Non-Key Employees than the schedule set forth in Section 8:

Years of Service	Vested Percentage
1 or less	0
2	20
3	40
4	60
5	80
6 or more	100

Section 13. PARTIES RESPONSIBLE FOR IMPLEMENTING THE PLAN

13.1 <u>Plan Sponsor</u>. The Company is the Plan sponsor for purposes of ERISA and designates in Sections 13.1.1 to 13.1.6 below how Plan powers and duties shall be performed.

13.1.1 <u>Company Powers and Duties</u>. The Company shall have the powers and duties set forth in the following (a)-(e):

- (a) <u>Plan and Trust Documents</u>. To make all Plan and Trust documents needed or desired to establish and operate the Plan and the separate Trust Fund, subject to the direction of the Board, or the Chief Human Resources Officer, as applicable.
- (b) <u>Plan Administration</u>. To perform all duties as Plan Administrator under 13.1.4, Section 3 and elsewhere provided in the Plan and Trust documents.
- (c) <u>Service Providers</u>. To enter into agreements with any third party administrative service providers for the Plan and Trust acting as accountant, actuary, asset custodian, attorney, auditor, contract administrator, recordkeeper or in any other administrative capacity, and to monitor such service providers performance under such agreements.
- (d) Plan Changes. To recommend to the Board or Chief Human Resources Officer any changes in Plan or Trust terms which the Company deems appropriate.
- (e) Other. To take any action deemed necessary or desirable to cause the Plan and Trust to be operated according to the Plan and Trust documents and applicable law.
- 13.1.2 <u>Board Powers and Duties</u>. Subject to the liability limitation in (f) below, the Board of Directors of the Company ("Board") has the exclusive powers set forth in the following (a)-(e):
- (a) <u>Plan and Trust Terms</u>. To establish, amend or terminate the Plan and the related Trust Agreement, subject only to Section 13.1.3 and Section 15.
- (b) <u>Funding Policy</u>. To determine that an appropriate funding policy, consistent with the objectives of the Plan, the Trust Agreement and the requirements of ERISA, is adopted and implemented.
- (c) <u>Contributions</u>. To determine the amount and manner of payment of all Company contributions to the Trust.
- (d) <u>Indemnification</u>. To determine the scope of any indemnification by the Company to any person or entity acting as a fiduciary or otherwise under the Plan or Trust, provide appropriate insurance and bonding coverage of any Employee of the Company acting in such capacity, and determine whether the Company shall furnish such insurance or bonding coverage to any other person or entity, all to the extent permitted by law.
- (e) <u>Committees</u>. To establish any Committee(s) of the Board deemed appropriate for Plan or Trust purposes.
- (f) <u>Liability Limitation</u>. The Board has no administrative or investment authority or functions, and no member of the Board shall be a Plan fiduciary because of such Board membership.
- 13.1.3 <u>Chief Human Resources Officer Powers and Duties</u>. Until such time as the Board shall modify, revoke or rescind such authority, all Employer or Plan sponsor functions and responsibilities vested in the Company shall be exercised pursuant to authorization by the Chief Human

Resources Officer of the Company. Without specific Board approval, the Chief Human Resources Officer has the powers and duties set forth in the following (a)-(d):

- (a) <u>Technical Amendments</u>. To amend the Plan and Trust Agreement to make technical, administrative, editorial and legal compliance changes recommended by Corporate Employee Benefits to comply with applicable law or to simplify or clarify the Plan.
- (b) <u>Substantive Amendments</u>. To take all actions necessary to implement (after approval by the Chairman or the Board) any amendments relating to Plan and Trust benefit or governance provisions.
- (c) <u>Plan Administration</u>. To delegate or terminate the power and authority of any person(s) or entity(ies) responsible for performance and administration of the Plan.
- (d) <u>Committees</u>. To establish, maintain or terminate the existence, membership and powers of any Committee for any Plan purpose, except any Committee established by the Board.
- 13.1.4 <u>Administrator Powers and Duties</u>. The Plan shall be administered by the Company, herein called the "Administrator." The Company shall be the "administrator" for purposes of ERISA § 3(16) and the named fiduciary for purposes of Plan administration. The Administrator shall have all powers necessary to carry out the provisions of the Plan, including those set forth in Section 3, but excluding those relating to the custody, management and control of Trust assets and those allocated or delegated to others.
- 13.1.5 <u>Retirement Committee</u>. Subject to the liability limitation under (g), the Retirement Committee established by the Board shall have the powers and duties set forth in the following (a)-(f):
- (a) <u>Asset Fiduciaries</u>. To make sure that Plan assets are held, safeguarded, invested and distributed by persons or entities that agree to act as the designated 'fiduciary' within the meaning of § 3(21) and other fiduciary provisions of ERISA for purposes of the applicable custodial, trusteeship, investment management or other Plan asset functions.
- (b) <u>Investment Funds</u>. To designate investment funds for the investment of Participants' accounts (the "Funds"), as determined appropriate by the Retirement Committee in consultation with qualified advisors, provided that the Funds shall at all times include the Nordstrom Common Stock Fund. The Nordstrom Common Stock Fund shall be invested primarily in the Company's common stock. The Nordstrom Common Stock Fund may also be invested in such cash or cash equivalent investments and other types of investments, as the Retirement Committee determines is appropriate to provide liquidity for cash benefit payments and transfers, the payments of administrative expenses, and other expected cash requirements. The Retirement Committee in its discretion may increase or decrease the number of Funds and may direct the Trustee to add or terminate specific Funds or modify existing Funds, other than the Nordstrom Common Stock Fund. Different Funds may be made available to different groups of Participants, determined on an Employer-by-Employer basis, in the discretion of the Retirement Committee.

- (c) <u>Investment Policy</u>. To establish the investment policy and guidelines for investment of Plan assets.
- (d) <u>Monitor Plan Asset Fiduciaries</u>. To establish the policies and procedures for periodic reporting by and review of performance by asset fiduciaries, and to implement any changes which such Committee, in its discretion, deems appropriate regarding such policies, procedures or fiduciaries.
- (e) <u>Monitor Plan Administration</u>. To establish the policies and procedures for periodic reporting by and review of performance by the Administrator and service providers involved in Plan administration, and to implement any changes which such Committee, in its discretion, deems appropriate regarding Plan administration.
- (f) <u>Claims Review</u>. To review and decide, as a committee or by its authorized subcommittee, all appeals of denied claims under Section 16.
- (g) <u>Liability Limitation</u>. The Retirement Committee has no administrative or asset responsibility or control beyond the limited oversight functions set forth above, and, subject only to applicable law, no member of such committee shall be liable for errors, omissions or breaches by any fiduciary or service provider having the actual power and authority to act.
- 13.1.6 <u>Investment Committee</u>. Until such time as the Retirement Committee shall modify, revoke or rescind such authority, an Investment Committee shall be established with the following authority to act for the Retirement Committee with respect to the performance of the Plan's investment vehicles and managers:
- (a) <u>Performance Review</u>. To review, monitor and evaluate, at reasonable intervals, the performance of the Trustee(s), the investment managers, investment vehicles and any other appointed or delegated fiduciaries or other service providers to ensure that their performance has been in compliance with the terms of the Plan and Trust documents, the investment policy and applicable law, and satisfies the needs of the Plan, and to report all findings and recommendations to the Retirement Committee.
- (b) <u>Investment Service Providers</u>. Subject to approval by or procedures of the Retirement Committee, to make or terminate the power and authority of any person(s) or entity(ies) responsible to hold, control, manage or invest assets of the Trust, including (but not limited to) any Trustee, custodian, investment manager, investment performance monitor or other provider of services involving Trust assets.
- 13.2 <u>Plan Fiduciaries</u>. The following Sections 13.2.1 to 13.2.6 apply to any individual or entity who is a "fiduciary" under ERISA § 3 (21) with respect to Plan or Trust administration or assets:
- 13.2.1 <u>Authorization</u>. Authority to act as a fiduciary shall be conferred as provided under 13.1 and accepted in writing by the designated fiduciary. Such authorization shall continue until the earliest of (a), (b) or (c), as follows:

Page **54** of **65** NORDSTROM 401(k) PLAN

- (a) if the fiduciary is unable to act, or
- (b) the fiduciary is terminated pursuant to authority under this Plan, or
- (c) upon the effective date of resignation by the fiduciary, which can be no earlier than the 30th day after written notice of resignation.
- 13.2.2 <u>Qualifications of Fiduciary</u>. Any individual, even if an officer, director, Employee or shareholder of the Company, and any corporation, partnership or other entity may serve as a fiduciary hereunder. All fiduciary responsibility may be vested in any single individual, group of individuals, corporation, partnership or other entity, or in any combination thereof, with liability being joint and several; or fiduciary responsibility may be divided among two (2) or more of the foregoing, with such duties and responsibilities as are provided in the authorizing designation and liability being limited solely to breach of the duties so imposed or conduct violating ERISA § 405(a).
- 13.2.3 <u>Other Fiduciaries</u>. The Company shall be the named fiduciary for any other rights or duties imposed by ERISA upon a "named fiduciary" which are not otherwise placed.
- 13.2.4 <u>Performance of Company Duties and Responsibilities</u>. The Company shall carry out its duties and responsibilities under the Plan through its directors, officers and Employees, acting on behalf of and in the name of Company in such respective capacities and not as individual fiduciaries.
- 13.2.5 <u>Scope of Responsibility</u>. No fiduciary or other person or entity responsible for any functions involving administration of the Plan or management of Trust assets shall be obligated to perform any duty or responsibility which has been allocated or delegated to another fiduciary pursuant to the Plan, the Trust Agreement or the procedures established therein.
- 13.2.6 <u>Multiple Fiduciary Capacities</u>. Nothing herein shall prohibit any person or entity, or group of persons or entities, from serving in more than one (1) fiduciary capacity with respect to the Plan.
- 13.3 <u>Plan Committees</u>. Unless otherwise provided in the specific authorization of the Committee, any Committee established under the Plan, having either overall or specifically limited responsibility of a ministerial or discretionary nature, as determined from time to time, shall be established and operated as provided below in Sections 13.3.1 to 13.3.6
- 13.3.1 <u>Procedure for Establishing Committee</u>. The party having authority to establish the Committee shall designate by written instrument the members of the Committee and the nature of the responsibilities the Committee is to carry out under the Plan; provided, however, that if the responsibilities of the Committee are fiduciary in nature, any such members shall consent in writing to serve in such capacity.
- 13.3.2 <u>Committee Composition</u>. The Committee shall be composed of three (3) or more members. Membership in the Committee is limited to individuals who are officers, directors, former directors or Employees of the Company.

- 13.3.3 <u>Committee Governance</u>. The Committee shall appoint from its members a chair and a secretary. The Committee may take any authorized action by a majority vote, and any writing signed by a majority of such members shall have the same effect and may be relied upon to the same extent as if signed by all members.
- 13.3.4 <u>Procedures</u>. To the extent consistent with the provisions of this Plan, the Committee shall have the power to adopt such rules of procedure and regulation as may be necessary for the proper execution of its duties.
- 13.3.5 <u>Vacancies</u>. Any member of a Committee may resign on thirty (30) days' advance written notice. Any member of a Committee may be removed from the Committee by the Board with or without cause. Removal of a Committee member does not require notice to be effective. Any Committee member who is an Employee but is not also an officer, director, or former director of the Company shall automatically cease to be a Committee member effective upon the date such individual ceases to be an Employee of the Company. All Committee vacancies shall be filled as soon as reasonably practicable. Until a new appointment is made, the remaining members of the Committee shall have authority to act although less than a quorum.
- 13.3.6 <u>Committee Compensation</u>. No member of any Committee shall receive any compensation for services as such, except that the Company may pay a reasonable fee to any member who is not a Participant under the Plan, not to exceed the amount paid to a Director to attend a Board meeting, for such person's attendance at any meeting of the Committee. Each member of the Committee shall be reimbursed by the Company for reasonable travel and other expenses actually incurred in attending meetings of the Committee and for any other proper purpose in connection with duties as such member. No bond or other security shall be required of any member of the Committee in such capacity, except to the extent required by law.
- 13.4 <u>Limitation of Individual Liability</u>. Subject to ERISA §§ 404 and 405, any individual acting in the administration of the Plan or as a Committee member shall be protected from personal liability as provided below in Sections 13.4.1 to 13.4.3:
- 13.4.1 <u>Plan Benefits and Expenses</u>. Such individual shall not be liable personally, either individually or jointly, for any debts, obligations, undertakings or benefit payments contracted or authorized in such capacity, but such debts, obligations, undertakings and benefit payments shall be paid solely and exclusively out of assets held in the Trust Fund.
- 13.4.2 <u>Investment</u>. Such individual shall not be obligated to invest or otherwise manage or control any portion of the assets held in the Trust Fund, such obligation having been delegated to third party fiduciaries pursuant to Section 13.1.
- 13.4.3 <u>Other Responsible Party</u>. Such individual shall not be responsible for any duty or function allocated or delegated to another person or entity pursuant to procedures hereunder, except to the extent that such individual is responsible for the selection and supervision of such other person or entity.

Page 56 of 65 NORDSTROM 401(k) PLAN

Section 14. SPENDTHRIFT PROVISIONS

- 14.1 <u>Prohibition Against Assignment</u>. The provisions of this Plan are intended as personal protection for the Participants. A Participant may not assign, anticipate or transfer any assets held for his or her benefit, including amounts credited to his or her account. The benefits under this Plan are not subject to seizure by legal process or in any way subject to the claims of the Participant's creditors, including, without limitation, any liability for contracts, debts, torts, alimony or support of any relative. The Plan's benefits or the Trust assets may not be considered an asset of a Participant in the event of his or her divorce, insolvency or bankruptcy. However, this Section 14.1 shall not apply to preclude the offset of a Participant's benefits if the Participant engages in misconduct with respect to the Plan as described in Code § 401(a)(13)(C).
- 14.2 <u>Effect of Assignment</u>. Any attempt by a Participant to assign, anticipate, or transfer any assets held for his or her benefit under the terms of this Plan shall be null and void.
- 14.3 QDRO Exception. Notwithstanding Sections 14.1 and 14.2, nothing in this Section 14 shall prohibit the distribution of plan assets to a Participant's Spouse or former Spouse pursuant to a "qualified domestic relations order" ("QDRO") as that term is defined in Code § 414(p), including any domestic relations order entered into before January 1, 1985, which Administrator determines to treat as a QDRO. The Administrator shall establish reasonable nondiscriminatory rules for determining the qualification and procedures for handling domestic relations orders, which rules shall be in writing, shall provide for prompt notification of prospective alternate payee under the order of the procedures for designating a representative to receive copies of any notifications.

Section 15. AMENDMENT AND TERMINATION OF PLAN

- 15.1 <u>Future of the Plan</u>. The Company expects to continue the Plan indefinitely. Future conditions, however, cannot be foreseen, and the Company reserves the right to amend or terminate the Plan at any time.
- 15.2 <u>Company Right to Amend the Plan</u>. The Company reserves the right, from time to time, to modify, alter or amend this Plan, as well as the Trust herein provided for, by action of the person or entity having power to amend under Section 13.1, subject to the following Sections 15.2.1 to 15.2.2:
- 15.2.1 <u>Retroactive Effect</u>. Any amendment may have retroactive effect to comply with legal requirements, Plan design, original intent or actual administrative practice, subject only to restrictions under Section 15.2.2.
 - 15.2.2 <u>Restrictions</u>. No amendment shall be made in violation of the following (a)-(d):
- (a) <u>Exclusive Benefit</u>. No amendment shall make it possible, at any time prior to the satisfaction of all liabilities with respect to Employees and their beneficiaries under the Trust, for any part of the corpus or income of the Trust to be used for, or diverted to, purposes other than for the exclusive benefit of the participating Employees of the Company or their beneficiaries.

Page 57 of 65 NORDSTROM 401(k) PLAN

- (b) No Cut Back of Accrued Benefit. No amendment (including a change in the actuarial basis for determining optional or early retirement benefits) shall decrease a Participant's benefit to the date of the amendment, except to the extent permitted under Code § 412(c)(8). A Plan amendment which results in (i) or (ii) with respect to benefits attributable to service before the amendment shall be treated as reducing accrued benefits: (i) eliminating or reducing an early retirement benefit or a retirementtype subsidy, or (ii) eliminating an optional form of benefit. In the case of a retirementtype subsidy, the preceding sentence shall apply only with respect to a Participant who satisfies (either before or after the amendment) the pre-amendment conditions for the subsidy. In general, a retirementtype subsidy is a subsidy that continues after retirement, but does not include a qualified disability benefit, a medical benefit, a Social Security supplement, a death benefit (including life insurance), or a plant shutdown benefit (that does not continue after retirement age).
- (c) <u>No Cut Back of Vested Benefit</u>. No amendment shall decrease a Participant's vested interest determined without regard to such amendment as of the later of the date such amendment is adopted, or becomes effective.
- (d) <u>Director</u>. No amendment shall permit any director who has not been an Employee to derive any benefits under the Plan.

15.3 Company Right To Terminate the Plan.

- 15.3.1 <u>Termination Event</u>. The Company may terminate this Plan at any time, and the Plan shall in any case be considered to have terminated if the Company shall completely discontinue contributions under the Plan or if the Company shall go out of existence, unless prior to such event the Plan shall be adopted and continued by a successor.
- (a) <u>Suspension of Contributions</u>. The Company reserves the right to suspend contributions to this Plan at any time, subject to relevant laws and regulations. A suspension is a temporary cessation of contributions and does not constitute or require a termination of the Plan. Such temporary discontinuance shall not constitute a formal termination of the Plan and shall not preclude later contributions.
- (b) <u>Merger or Consolidation</u>. In the case of any merger or consolidation with, or transfer of assets or liabilities to, any other plan, each Participant of this Plan shall receive a benefit which is equal to the benefit he/she would have been entitled to receive immediately before the merger or consolidation as if the Plan had then terminated. Moreover, prior to any transfer pursuant to this Section 15.3.1(b), the administrator of the transferee plan shall provide adequate assurances and representations to the Administrator that those portions of Participant accounts that are subject to the limitations of Section 9.8 as of the date of transfer shall subsequently remain subject to such limitations under the transferee plan. However, this provision shall not be construed to be a termination or discontinuance of the Plan or to be a guaranty of a specified level of benefit from the Plan.
- (c) <u>Effect of Dissolution, Bankruptcy, General Assignment</u>. The Plan shall be deemed terminated if the Company is dissolved or adjudicated bankrupt, or makes a general assignment its assets (but not Trust assets) for the benefit of creditors, unless a party having proper authority elects to continue the Plan.

- 15.3.2 Termination Benefits and Expenses. In the event of such Plan termination, the rights of each retired Participant and Participant to the benefits accrued or credited to the date of such termination, to the extent then funded, shall become one hundred percent (100%) vested on such termination and shall thenceforth be nonforfeitable, and the assets in the Trust shall be used, so far as they will extend, and subject to the conditions and limitations herein contained:
 - (a) Expenses. To pay all expenses and liabilities (absolute or contingent) of the Trust Fund;
- (b) <u>Benefits</u>. To pay, provide or distribute, pursuant to Section 10, all remaining Trust Fund assets to the Participants in the proportions determined by their respective accounts.
- (c) <u>Source of Payments</u>. To provide for benefit distribution by payment from the Trust Fund or nontransferable annuities purchased from an insurance company, with the right to commute any benefit amount on an actuarial basis, all as determined by the Retirement Committee in the exercise of its discretion.
- (d) <u>Reversion to Company</u>. To pay to the Company any residual assets not allocated under Section 15, to the extent permitted by law.
- 15.4 <u>Partial Termination</u>. In the event of a partial termination of this Plan, Section 15.3 shall be considered as applying, at such time, only to those Participants with respect to whom the Plan has been terminated. All other Participants shall be unaffected by such termination to the fullest extent allowable by then current law and regulations.
- 15.5 <u>Procedure for Plan Amendment or Termination</u>. The amendment and termination powers reserved in Section 13.1 and Section 15 shall be executed as follows:
- 15.5.1 <u>Board Resolution or Chairman Action</u>. Except as provided in Section 15.5.2, the Company may amend or terminate the Plan by execution of the amendment by the Company Chairman, or pursuant to authorization in a resolution adopted by the Board of Directors (or its Executive Committee) and delivered to the Administrator, Retirement Committee and Trustee.
- 15.5.2 <u>Chief Human Resources Officer Action</u>. The Chief Human Resources Officer of the Company may amend the Plan to make such changes as are authorized under Section 13.1.3 by designating such changes in writing to the Administrator, Retirement Committee and Trustee.
- 15.5.3 <u>Proof of Amendment</u>. Any officer of the Company, other than the individual who has the power to create or execute the amendment or termination document, may certify that such document has been adopted by proper authority.

Section 16. CLAIMS AND REVIEW PROCEDURE

16.1 <u>Claims for Benefits and Inquiries</u>. Any Participant or beneficiary may file with the Administrator a written claim for benefits or inquiry concerning the Plan, or concerning present or future rights to benefits under the Plan. Applications for benefits must be made on the forms prescribed by the

Page **59** of **65** NORDSTROM 401(k) PLAN

Administrator, signed by the Participant or beneficiary, as applicable, and submitted to the Administrator's benefit claims office.

- 16.2 <u>Denial of Claims</u>. In the event any claim for benefits is denied, in whole or in part, the Administrator shall notify the applicant of such denial in writing and shall advise the applicant of the right to a review thereof.
- 16.2.1 <u>Content of Notice</u>. Such notice shall be written in a manner calculated to be understood by the applicant and set forth the following:
 - (a) The specific reason for denial.
 - (b) Specific reference to the Plan provisions upon which the denial is based.
- (c) A description of any additional information which is necessary to perfect the claim and why this information is necessary.
 - (d) An explanation of the review procedure described in Section 16.3 below.
- 16.2.2 <u>Timing of Notice</u>. Such written notice shall be given to the applicant within ninety (90) days after the Administrator receives the application, unless special circumstances require an extension of time of up to an additional ninety (90) days for processing the application. If such an extension is required, written notice of the extension shall be furnished to the applicant prior to the termination of the initial ninety (90) day period. This notice of extension shall indicate the special circumstances requiring the extension of time and the date by which the Administrator expects to render its decision on the application for benefits. If written notice of denial of the application for benefits is not furnished within the time specified in this Section 16.2.2, the application shall be deemed denied.
- 16.3 Review of Denied Claims. Any applicant whose claim for benefits is denied (or deemed denied) in whole or in part, or such applicant's authorized representative, may appeal from such denial by submitting to the Retirement Committee a written request for a review of the application within sixty (60) days after receipt of denial of the notice (or, in the case of a deemed denial, sixty (60) days after the application is deemed denied). The Retirement Committee shall give the applicant or such representative an opportunity to review pertinent documents (other than legally privileged documents) in preparing the request for review. The request for review shall be in writing and shall be addressed as follows:

Retirement Committee for the Nordstrom 401(k) Plan c/o Employee Benefits – Retirement 1700 Seventh Avenue, Suite 1000 Seattle, WA 98101

The request for a review shall set forth all grounds on which it is based, all facts and documents in support of the request and any other matters which the applicant deems pertinent. The

Page 60 of 65 NORDSTROM 401(k) PLAN

Retirement Committee may require the applicant to submit such additional facts, documents or other material as it may deem necessary or appropriate in making its decision on review.

- 16.4 <u>Decision on Review</u>. After receiving the application for review, the Retirement Committee, or an authorized review subcommittee thereof ("Review Committee") shall review and decide the final disposition of the claim. Such decision of the Review Committee shall be binding on all parties.
- 16.4.1 <u>Timing of Review</u>. The decision should be reached within sixty (60) days after receipt of the application for review, although special circumstances may delay the review decision up to one hundred twenty (120) days. If such an extension is required, written notice of the extension shall be furnished to the applicant prior to the end of the initial sixty (60) day period.
- 16.4.2 <u>Notice of Decision</u>. If the Review Committee confirms the denial of the application for benefits in whole or in part, such notice shall set forth, in a manner calculated to be understood by the applicant, the specific reasons for such denial and specific references to the Plan provisions on which the decision is based. If the Review Committee determines that the application for benefits should not have been denied in whole or in part, the Review Committee shall direct the Administrator to take appropriate remedial action as soon as reasonably practicable. If written notice of the Review Committee's decision is not given to the applicant within the time period prescribed in Section 16.4.1, the application will be deemed denied on review.
- 16.5 <u>Rules and Procedures on Review</u>. The Review Committee shall establish such rules and procedures, consistent with the Plan and with ERISA, as it may deem necessary or appropriate in carrying out its responsibilities in reviewing a denied claim. The Review Committee may require an applicant who wishes to submit additional information in connection with an appeal to do so at the applicant's own expense, and may convene a hearing if it determines that sufficient cause is shown. To the extent that a claim requires a determination of whether a Participant suffers from a Disability as defined in Section 2.7, the Plan shall adhere to the procedures for administering disability claims under the Nordstrom, Inc. Welfare Benefit Plan, which procedures are incorporated by this reference.
- 16.6 Exhaustion of Remedies. No legal action for benefits under the Plan shall be brought unless and until the applicant has (i) submitted a written claim for benefits in accordance with Section 16.1; (ii) been notified by the Administrator that the application is denied (or the application is deemed denied) as provided in Section 16.2; (iii) filed a written request for a review of the application in accordance with Section 16.3; and (iv) been notified in writing that the Review Committee has affirmed the denial of the application (or the application is deemed denied) on review as provided in Section 16.4.
- 16.7 <u>Time for Initiating Legal Action</u>. Any legal action related to a claim for Plan benefits must be initiated within two (2) years after the Review Committee has issued its final decision on review.

Section 17. MISCELLANEOUS PROVISIONS

17.1 <u>No Right of Continued Employment</u>. The establishment of this Plan, the creation of any fund or account, or the payment of any benefits shall not create in any Employee, Participant or other

Page **61** of **65** NORDSTROM 401(k) PLAN

party a right to continuing employment or create any claim against the Plan or Trust Fund for any payment except as set forth in this Plan.

- 17.2 <u>Discretion</u>. Whenever, under the provisions of this Plan, discretion is granted to the Employer or Administrator which affects the benefits, rights and privileges of Participants, such discretion shall be exercised uniformly so that all Participants similarly situated shall be similarly treated.
- 17.3 <u>Separability</u>. If any provision of this Agreement is declared invalid or unenforceable, the remaining provisions shall be effective.
- 17.4 <u>Participant and Others Bound by Plan</u>. Each Participant, by executing the beneficiary designation, agrees for himself or herself and his or her heirs, beneficiaries, successors, and assigns to be bound by all of the provisions of this Plan.
- 17.5 <u>Applicable Law</u>. This Plan is to be construed according to the laws of the State of Washington, to the extent not preempted by federal law.
- 17.6 <u>Text Controls</u>. The paragraph numbers and headings herein are solely for convenience. In the event of conflict between them and the text, provisions of the text control.
- 17.7 Expenses. All reasonable expenses incurred in operating and administering the Plan, including expenses of the Company, the Committee, and the Trust, may be paid from the Trust Fund or, at the election of the Company, may be paid by the Company, provided, however, that the Trust may reimburse the Company for such expenses only to the extent such amounts constitute "direct expenses" in accordance with U.S. Department of Labor Regulation § 2550.408c-2(b)(3). This provision shall be deemed to be a part of any contract to provide for expenses of Plan administration, whether or not the signatory to such contract is, as a matter of convenience, the Company.
- 17.8 <u>Plan Document is Controlling</u>. All rights and benefits of Participants and beneficiaries are controlled and determined by the provisions of this Plan document. To this end:
- 17.8.1 <u>Authorized Summaries</u>. The only authorized summaries of the Plan are the summary plan description and any employee communications prepared or approved by the Company. Any such authorized summaries will not modify or override the Plan document.
- 17.8.2 <u>Authorized Representatives</u>. The only individuals authorized to explain or interpret the Plan are the Committee members and the Plan administrative personnel who are charged with such responsibility. No other individual or entity has authority to explain or interpret the Plan. No authorized individual has authority to override or modify what is provided in the Plan document.
- 17.8.3 <u>Resolution of Conflicts</u>. In the event of any conflict between this Plan and (1) any authorized summary of the Plan, or (2) other written, oral or electronic statement, or (3) any assumption, inference or reliance by any Participant or beneficiary, this Plan document shall be dispositive.

17.9 <u>Rules of Construction</u>. In construing this Plan, the masculine and neuter genders include the feminine and each other and the singular includes the plural.

Section 18. LOANS TO PARTICIPANTS

18.1 Loans to Participants.

- 18.1.1 <u>Participant's Right to Borrow</u>. Participants and Beneficiaries who are parties in interest under Section 3(14) of ERISA shall have the right to borrow from their Elective Deferral Contribution accounts, Employer Matching Contributions accounts, and Employer Profit Sharing Contributions accounts on a reasonably equivalent basis and subject to prior approval by the Administrator. Application for a loan must be submitted to the Administrator on such form(s) and in such manner as the Administrator may require. Approval shall be granted or denied as specified in Section 18.1.2 on the terms specified in Section 18.1.3.
- 18.1.2 <u>Limits on Borrowed Amount</u>. The Administrator shall grant any loan which meets each of the requirements of paragraphs (a), (b) and (c) below:
- (a) <u>Maximum Loan</u>. The amount of the loan, when added to the outstanding balance of all other loans to the Participant from the Plan or any other qualified plan of the Company or any related Company shall not exceed the lesser of:
- (1) \$50,000, reduced by the excess, if any, of a Participant's highest outstanding balance of all loans from the Plan or any other qualified plan maintained by the Company or any related Company during the preceding twelve (12) months over the outstanding balance of such loans on the loan date, or
- (2) Fifty percent (50%) of the value of the vested balance of the Participant's accounts as of the Valuation Date preceding the date upon which the loan is made.
 - (b) Minimum Loan. The loan shall be for at least \$1,000; and
- (c) <u>Outstanding Loan Limitations</u>. No more than two (2) loans may be outstanding to a Participant at any time. Notwithstanding the foregoing, no more than one (1) loan used to purchase the principal residence of a Participant may be outstanding to a Participant at any time.
- 18.1.3 <u>Repayment and Collateral</u>. Each loan granted shall, by its terms, satisfy each of the following additional requirements:
- (a) <u>Term</u>. Each loan, by its terms, must be repaid within sixty (60) months (except that if the Administrator is satisfied that the loan proceeds are being used to purchase the principal residence of a Participant, the Administrator may, in its discretion, establish a term of up to two hundred and forty (240) months for repayment).
- (b) <u>Interest</u>. Each loan shall bear a reasonable rate of interest, which rate shall be established by the Administrator from time to time and shall provide the Plan with a return commensurate with the interest rates charged by persons in the business of lending money for loans

which would be made under similar circumstances and shall in no event be less than one percent (1%) over the prime rate published in the Wall Street Journal as of the 15th business day of the month preceding the date on which the loan is funded.

- (c) <u>Repayment Amount</u>. Each loan must require substantially level amortization over the term of the loan, with payments not less frequently than semi-monthly (twice each calendar month).
- (d) <u>Collateral</u>. Each loan must be adequately secured, with the security to consist of the balance of the Participant's accounts.
- (e) <u>Means of Payment</u>. Automatic payroll deductions shall be required as additional security and the loan shall become immediately due and payable if the Participant ceases the payroll deduction. Notwithstanding the foregoing, to avoid default, an active Participant who has insufficient payroll from which to deduct the loan payment must, in a manner prescribed by the Plan Administrator, make timely loan payments equal to the amount of the loan payment not deducted by payroll deduction. Other than for Participants on qualified military service (the loan payment suspension rules for which are provided under 5.7), loan payments for a Participant who is on an approved leave of absence shall be suspended during the leave of absence, unless the Participant elects to continue loan repayments during the leave of absence, but the suspension period shall not be longer than 12 months. Upon return from the leave of absence, the Participant may make a single sum make-up payment equal to the amount of the suspended payments during the leave, or may increase the periodic loan payment so that the loan term is not extended beyond the term established when the loan was originated. A Participant who severs employment with an outstanding loan balance may elect to continue monthly loan repayments.
- (f) <u>Value Only in Borrower's Account</u>. To the extent a Participant's loan is secured by the Participant's account, the investment gain or loss attributable to the loan shall not be included in the calculation or allocation of the increase or decrease in fair market value of the general assets of the Plan pursuant to Section 6.2. Instead, the entire gain or loss (including any gain or loss attributable to interest payments or default) shall be allocated to the accounts of the Participant.
- 18.1.4 <u>Payments Credited to Account</u>. All loan payments shall be transmitted by the Company to the Trustee as soon as practicable but not later than the end of the month during which such amounts were received or withheld. Each loan may be prepaid in full at any time. Any prepayment shall be paid directly to the Trustee in accordance with procedures adopted by the Administrator.
- 18.1.5 <u>Promissory Note</u>. Each loan shall be evidenced by a promissory note executed by the Participant and payable in full to the Trustee, not later than the earliest of (a) a fixed maturity date meeting the requirements of Section 18.1.3(a) above, (b) the Participant's death, or (c) the termination of the Plan. Such promissory note shall evidence such terms as are required by this section.
- 18.1.6 <u>Administrator Powers</u>. The Administrator shall have the power to modify the above rules or establish any additional rules with respect to loans extended pursuant to this section. Such additional rules shall include establishment of a reasonable loan fee to reimburse the Plan for the administrative costs of making such loans and establishment of rules for default. The rules may be

included in a separate document or documents and shall be considered a part of this Plan; provided, each rule and each loan shall be made only in accordance with the regulations and rulings of the Internal Revenue Service and Department of Labor and other applicable state or federal law. The Administrator shall act in its sole discretion to ascertain whether the requirements of such regulations and rulings and this section have been met. The Administrator may delegate any of its powers under this Section in accordance with the provisions of Section 3.

IN WITNESS WHEREOF pursuant to Section 13.1.3, this 2021 Restatement has been executed on behalf of the Company by its Chief Human Resources Officer pursuant to authorization of the Company's Board of Directors this 29th day of April 2021.

NORDSTROM, INC.

By:	
	Christine Deputy

Title: Chief Human Resources Officer

Page **65** of **65** NORDSTROM 401(k) PLAN

APPENDIX A Participant Investment Direction Appendix to the Nordstrom 401(k) Plan

2019 Restatement

This Participant Investment Direction Appendix ("Appendix") to the Nordstrom 401(k) Plan (the "Plan") is effective on and after January 1, 2019, for those amounts in Participant's Plan accounts subject to the Participant Investment Control provisions of Section 6.3 of the Plan.

All Plan assets attributable to Salary Deferral, Rollover, Employer Matching Contributions and Profit Sharing Contributions, and investment earnings and gains or losses thereon ("Eligible Accounts"), shall be held by one (1) or more of the Plan Trustees under a Trust Agreement, to be managed, invested, reinvested and distributed in accordance with the Plan, the Trust Agreement and any agreement with an insurance company or other financial institution constituting a part of the Plan and Trust Fund. The Trustee shall cause to be established and maintained the multiple investment Funds approved by the Retirement Committee from time to time.

1.01 Establishment and Operation of Funds. The Trustee shall establish and maintain at least three (3) investment funds ("Funds"), and establish and implement procedures to enable each account to be invested in one (1) or more of such Funds pursuant to directions made from time to time by each Participant on whose behalf the account is maintained in accordance with the provisions of this Appendix. The Trustee shall establish such other Funds or investment options as the Committee may select and designate in writing.

<u>1.02 Fund Selection</u>. As of the effective date of this Appendix, the Funds shall consist of the Basic Investment Funds, the Nordstrom Common Stock Fund, and the Participant Self-Directed Brokerage Account ("SDBA"), pursuant to 1.02.1, 1.02.2 and 1.02.3.

- 1.02.1 Basic Investment Funds. Each Basic Investment Fund shall meet the following criteria 1.02.1(a) (c):
- (a) <u>Individual Criteria</u>. Each Basic Investment Fund shall meet the following specific criteria 1.02.1(a)(1) and (2):
 - (1) <u>Prudent Investment Management</u>. Each Basic Investment Fund shall be under the fiduciary investment management of an Investment Manager or registered investment adviser within the meaning of ERISA §§ 3(21) and 3(38), shall consist of one (1) or more investments which are prudent for Participants who direct how their Plan account balances shall be invested, and which, within the type of investments used for such Basic Investment Fund, meet the diversification and locus of ownership requirements of ERISA § 404 and the liquidity needs of the Plan.

Page 1 of 9
Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

- (2) <u>Types of Funds</u>. Without limitation, any Basic Investment Fund may be a diversified investment portfolio of any of the following types 1.02.1(a)(2)(i) (v) (or any combination thereof):
 - (i) any registered mutual fund or funds;
 - (ii) direct investments made by an Investment Manager designated from time to time by the Retirement Committee;
 - (iii) any common, collective or commingled Trust Fund maintained by a bank or other financial institution which is qualified under Code §§ 401(a) and 501(a) and constitutes a part of the Plan and Trust Fund;
 - (iv) stocks, common or preferred, bonds and other evidences of indebtedness or ownership of Trustee's choice;
 - (v) a group deposit administration annuity contract or other type of contract issued by an insurance company or companies which utilizes commingled, separate or individual separate investment accounts and constitutes a part of the Plan and Trust Fund; or
 - (vi) Retirement Committee-designed funds (such as the "Target Retirement Date Funds"), each of which shall in its entirety be a diversified portfolio composed of one or more portfolios of direct investments made and allocated among any of (i) (v).
- (b) <u>Aggregate Criteria</u>. In the aggregate, the Basic Investment Funds shall meet the following criteria 1.02.1(b)(1) and (2):
 - (1) <u>Broad Range of Investment Alternatives</u>. At any time, there shall be at least three (3) such other diversified Funds which individually have materially different risk and return characteristics from each other, and in the aggregate will enable the Participant to achieve an investment mix with risk and return characteristics that are normally appropriate for a participant in a plan of this type. Without limitation, such grouping of Funds shall include at least one (1) Fund from each of the following three (3) general types of investment characteristics:
 - (i) An income producing fund with low risk and ready liquidity.
 - (ii) A balanced fund designed to achieve higher return than (i) over longer term market cycles by investing both in stocks and other equities, and bonds and other fixed income investments, and which may be expected to have higher risk than (i).
 - (iii) An equity fund designed to achieve higher return than (i) or (ii) over longer term market cycles from investment in stocks and other equity investments, and which may be expected to have higher risk than (i) or (ii).
 - (2) <u>Multiple Funds</u>. At any time there may be more than one (1) Fund of any type specified in 1.02.1(b)(1) above, and one (1) or more additional Funds which

Page 2 of 9

Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

might not be any of types 1.02.1(b)(1)(i) - (iii), and the risk and return characteristics of each Fund within a type may vary from Fund to Fund.

- (c) <u>Participant Choice, Generally</u>. Except as otherwise provided for the Nordstrom Common Stock Fund under 1.02.2 below, Participant choice of investment among Funds shall be administered by the Company under the following rules 1.02.1(c) (1) (7):
 - (1) <u>Frequency</u>. Each Participant shall be entitled to make or change any investment direction or mix for the Participant's account once during each consecutive three (3) month period. The Retirement Committee may allow more frequent direction if appropriate for any particular Fund or if it decides to do so for all Funds.
 - (2) Form. The investment direction by the Participant may be made by any written, telephonic or electronic means approved by the Retirement Committee and shall be in any multiple of one percent (1%) of the amount or account to be invested at the time of direction. The Company shall be responsible for establishing safeguards to assure that a specific direction for a Participant's account is by the Participant or the Participant's legal representative, and for reporting such directions to the Trustee, and to any administrative office of the Plan or any Fund as needed to effect any such direction. Trustee shall make sure that all investments are made pursuant to directions communicated to it. Trustee shall have the right to rely on the direction provided by the Retirement Committee and not be liable for failure to execute a Participant's election unless such failure is due to Trustee's negligence, willful misconduct or failure to act in good faith.
 - (3) <u>Disclosures</u>. The Company shall provide the following information in (A) (I) to all Participants in such form and at such times as the Company shall deem appropriate to provide sufficient information to enable each Participant to make informed decisions with regard to investment alternatives and the incidents of ownership of any such investment:
 - (A) <u>ERISA</u> § 404(c). An explanation that the Plan is intended to constitute a plan described in ERISA § 404(c) and related regulations, and that the fiduciaries of the Plan may be relieved of liability for any losses which are the direct and necessary result of investment instructions given by any Participant.
 - (B) <u>Investment Alternatives</u>. A description of the investment alternatives available under the Plan and, with respect to each designated investment alternative, a general description of the investment objectives and risk and return characteristics of each such alternative, including information relating to the type and diversification of assets comprising the portfolio of the designed investment alternative.
 - (C) Investment Managers. Identification of any designated Investment Managers.

Page **3** of **9** Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

- (D)<u>Choice by Participant</u>. An explanation of the circumstances under which Participant may give investment instructions and of any specified limitations on such instructions under the terms of the Plan, including any restrictions on transfer to or from a designated investment alternative, and any restrictions on the exercise of voting, tender and similar rights appurtenant to a Participant's investment in an investment alternative.
- (E) <u>Fees and Expenses</u>. A description of any transaction fees and expenses which affect the Participant's account balance in connection with purchases or sales of interests in investment alternatives (e.g., commissions, sales load, deferred sales charges, redemption or exchange fees).
- (F) <u>Information Source</u>. The name, address, and phone number of the Company representative responsible for providing relevant information upon request and a description of the information which may be obtained on request.
- (G)<u>Confidentiality</u>. A description of the procedures established to provide for the confidentiality of information relating to the purchase, holding and sale of Company securities in the Nordstrom Common Stock Fund and the exercise of voting, tender and similar rights in Nordstrom Common Stock Fund shares, and the name, address and phone number of the Trustee representative responsible for monitoring compliance with the procedures.
- (H)<u>Prospectus Information</u>. A copy of the most recent prospectus or similar disclosure document furnished by any Fund to the Company or Trustee shall be provided to a Participant immediately before or after such Participant makes the initial investment in such Fund.
- (I) <u>Information Upon Request</u>. Upon request, a Participant shall be given the following most current information available to the Plan: (i) a description of the annual operating expenses of each Fund, including the total expenses as a percentage of average net assets; (ii) the latest prospectus, financial statements and reports furnished to the Plan; (iii) a list of assets held by each Fund and the value or percentage each asset represents in the Fund, including with respect to any fixed rate non-governmental investment contract the identity of the issuer and the term and return rate of the contract; (iv) share or unit value in each Fund and past and current investment performance on a reasonable and consistent basis; (v) the value of shares or units held in the Participant's account.
- (4) <u>Voting Rights</u>. Except as provided below in (d) regarding the Nordstrom Common Stock Fund, the Trustee or Investment Manager of the Fund shall have complete discretion whether or not to exercise all voting, tender and other ownership rights incidental to ownership of Plan investments in any Fund and such rights shall not be passed through to or exercised by any Participant.

Page 4 of 9
Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

- (5) <u>Plan Expenses</u>. No account balance shall be reduced by any expense charge other than (i) expenses charged by the Fund to derive net share or unit value of the Participant's investment in that Fund, and (ii) expenses charged by the Plan attributable to the operation and administration of the Plan under Section 3.9, or to make or process any account transactions with respect to the Participant.
- (6) <u>Default Fund</u>. The Retirement Committee shall designate the Fund or Funds that shall be used in the event a Participant fails to make a timely designation of any particular Fund(s), and assets in Participant's account shall be so invested until changed by the Participant's later timely designation.
- (7) <u>Participant Free Choice</u>. Each Participant shall have full power and discretion to control the investment of all assets in such Participant's account without restriction and without regard to the resulting risk or return or what any other Participant or a prudent fiduciary might do in like circumstances. Every designation of any investment Fund by a Participant shall be an exercise of Participant control for purposes of the limitation of fiduciary liability under 1.04.
- <u>1.02.2 Nordstrom Common Stock Fund</u>. The following special rules 1.02.2(a)-(f) apply to Participant directed investment in Nordstrom common stock under the Nordstrom Common Stock Fund to the extent they modify the general rules in 1.02.1(a), above:
- (a) <u>Participant Directed Investment</u>. The Trustee shall establish the Nordstrom Common Stock Fund to receive, invest and reinvest Eligible Accounts primarily (or exclusively) in Company stock, in accordance with a Participant's direction pursuant to this Appendix and the Plan. The Trustee may place Trust assets awaiting investment in Company stock in any interest bearing accounts or certificates of deposit offered by any bank (including the Trustee) or savings and loan association, in governmental obligations, or in stocks, shares and obligations of corporations or of unincorporated associations or trusts or investment companies or in any kind of investment fund (including any pooled investment fund maintained by the Trustee), without regard to whether or not such investment is an authorized or appropriate investment for Trustees under any state laws, or hold them in cash.
- (b) <u>Voting</u>. Voting, tender and similar ownership rights incidental to ownership of the shares of Company stock held in or for the Nordstrom Common Stock Fund account of each Participant ("shareholder Participant") shall be passed through to the shareholder Participant in accordance with 1.02.2(b)(1) and 1.02.2(b)(2), and 1.02.2(c) and 1.02.2(d) next following:
 - (1) <u>Shareholder Information</u>. Each shareholder Participant shall receive all proxy statements, proxies and other information furnished to Company shareholders in connection with ownership of Company stock and any matters submitted to them for shareholder vote, tender or other shareholder action.
 - (2) Exercise of Shareholder Rights. Each shareholder Participant shall have the right to determine relevant information and exercise or not exercise all shareholder

Page **5** of **9**

Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

rights, in person or by proxy, and to do so on a confidential basis, without potential for undue employer influence in connection therewith.

- (c) <u>Company Role</u>. In addition to other powers and duties under this Appendix involving Nordstrom Common Stock Fund shares, the Company shall comply with the following 1.02.2(c)(A) and (B) with regard to all Nordstrom Common Stock Fund shares and Participant rights:
 - (A)<u>Confidentiality</u>. Each Company employee involved in Plan administration shall keep in confidence all confidential information regarding any individual Participant's action or inaction regarding any Nordstrom Common Stock Fund participation or nonparticipation, unless the Participant gives voluntary consent to disclosure or the information must be disclosed to comply with applicable law.
 - (B) <u>Notice to Trustee</u>. The Company shall notify the independent fiduciary responsible for voting the Nordstrom Common Stock Fund shares (the "Voting Trustee") of any facts or circumstances that do or might involve the potential for undue employer influence.
- (d) <u>Voting Trustee Role</u>. In connection with any Participant's purchase, holding, voting, tender, sale or exercise of other shareholder rights involving Company stock in the Nordstrom Common Stock Fund, the Voting Trustee, as a Plan fiduciary that is independent of the Company or other Plan sponsor, shall establish procedures, make determinations and take other actions the Voting Trustee deems appropriate to comply with the following 1.02.2(d)(1) (3):
 - (1) <u>Confidentiality</u>. Voting Trustee shall ensure that appropriate safeguards are in place to maintain the confidentiality of information relating to such Participant actions, unless the Participant gives voluntary consent to disclosure or the information must be disclosed to comply with applicable law.
 - (2) <u>Avoid Undue Employer Influence</u>. The Voting Trustee shall take appropriate action to enable each shareholder Participant to exercise Nordstrom Common Stock Fund shareholder rights without undue employer influence in any circumstance that Voting Trustee, or another Plan fiduciary, determines to involve a potential for undue employer influence upon Participants with regard to direct or indirect exercise of shareholder rights.
 - (3) <u>Voting or Tender of Shares</u>. With regard to voting or tender of any Nordstrom shares held in the Nordstrom Common Stock Fund, the Voting Trustee (i) shall ensure that Participant directions to vote for or against, or in the discretion of the Voting Trustee, or to tender or not tender, with respect to Participant's Nordstrom Common Stock Fund shares is effected as directed by the Participant; (ii) as legal owner, shall exercise its sole discretion as to any shareholder rights regarding all shares for which the Voting Trustee does not receive the Participant's direction; and (iii) shall take such action as Voting Trustee, in its sole discretion, deems

Page **6** of **9**Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

appropriate in any circumstance that does or might involve the potential for undue employer influence.

- (e) <u>Registration</u>. In the event that the Retirement Committee directs the Trustee to dispose of any Company stock held as a Nordstrom Common Stock Fund asset, under circumstances which require registration and/or qualification of the securities under applicable federal or state securities laws, the Company, at its own expense, will take, or cause to be taken, any and all such actions as may be necessary or appropriate to effect such registration and/or qualification.
- (f) <u>Distributions</u>. The Trustee shall make distributions from the Trust, at such times and in such number of shares of Company stock and/or amounts of cash, to the person entitled thereto under the Plan, as the Retirement Committee directs in writing. Any undistributed portion of a Participant's Company Stock Account under the Plan shall be retained in the Trust until the Retirement Committee directs its distribution. Where distribution is directed in Company stock, the Retirement Committee or the Trustee shall cause the Company to issue an appropriate stock certificate to the person entitled thereto, to be delivered to such person by the Retirement Committee. Any portion of a Participant's Company Stock Account to be distributed in cash shall be paid by the Trustee's furnishing its check to the Retirement Committee for delivery to the Participant (or beneficiary).

<u>1.02.3 Self-Directed Brokerage</u>. In addition to the Funds provided for in 1.02.1 and the Nordstrom Common Stock Fund under 1.02.2, such Eligible Accounts as may be invested in the Funds may also be invested in a Self-Directed Broker Account "SDBA." Participants directing the investment of their Eligible Accounts shall pay all expenses relating to such SDBA, and the SDBA shall be subject to certain restrictions. In particular, the following restrictions apply:

- (a) <u>SDBA Provider</u>. The Company shall select an SDBA provider who, in the Company's judgment, will adequately administer and monitor the operation of each Participant's SDBA account to ensure that it operates as provided for under the Plan.
- (b) <u>Investment Restrictions</u>. The following restrictions apply to investments available through an SDBA:
 - (1) In accordance with ERISA § 404(b), no investment shall be permitted that would cause the Plan to hold assets with "indicia of ownership" outside the jurisdiction of the federal district courts of the United States.
 - (2) No investment shall be permitted that could result in a loss in excess of such investment's principal. This class of investment includes, but is not limited to:
 - (A)margin arrangements;
 - (B) commodities contracts;
 - (C) short selling; and

Page 7 of 9
Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

- (D) any other investment with a loss potential in excess of the investment's principal.
- (3) No investment shall be permitted that would result in a direct or indirect:
 - (A)Loan to the Employer or any of its Affiliates;
 - (B) Acquisition or sale of real property that is leased to an Employer of Employees covered by the Plan (or an Affiliate of such Employer).
 - (C) Acquisition of Company or Affiliate stock; or
 - (D)Sale, exchange or lease of property between the Plan and the Employer or its Affiliates (other than the purchase or sale of a qualifying employer security or the acquisition or disposition of an interest in an investment fund managed by the Employer or any of its Affiliates).
- (4) No investment shall be permitted that would violate the prohibited transaction rules under Department of Labor Regulation § 2510.3-101.
- (5) Investment shall be permitted only in publicly-offered securities registered under the Securities Exchange Act of 1934 or securities issued by an investment company registered under the Investment Company Act of 1940; investments not-so-registered, including but not limited to life insurance contracts and collectibles, shall in no way be permitted as an investment in an SDBA.
- (6) An investment shall be permitted only to the extent that such investment is priced no less frequently than any day the New York Stock Exchange is open for business, so as to ensure that distributions and Participant directed transfers can occur on a timely basis.
- 1.03 Transaction Restrictions. Investments under this Appendix may generally be made on any Valuation Date provided in the Plan; however, notwithstanding any other provision to the contrary, Participant direction and control over the account balances shall be subject to the administrative limitations and restrictions imposed, and by the procedures established by the Company, Committee, Trustee, or the SDBA provider, as applicable. In that regard, the following restrictions may apply to a particular transaction:
 - <u>1.03.1 "Daily" Valuation and Transactions</u>. Although the Plan is currently intended to allow Participants daily access to their account balances, account balance transactions are typically processed after the close of the applicable trading market on a Valuation Date (which typically is 4:30 p.m. eastern time) where trading occurs. Pricing of investments available through the Plan will reflect a weighted average of the transactions in that investment on each Valuation Date.
 - <u>1.03.2 Account Settlement</u>. Amounts subject to a transfer, reallocation or other transaction involving a participant's account will not be available for further direction, transfer, reallocation or other transaction until the transaction actually takes place, is accounted for

Page 8 of 9

Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

and actually settles. In the usual case, this occurs on the next Valuation Date for the Investment Funds and on the third following Valuation Date for the Nordstrom Common Stock Fund. Funds invested and transactions occurring in a Participant's SDBA account will settle as provided for in the SDBA account agreement and as provided for by the SDBA provider. In addition, the establishment of a SDBA will occur as soon as reasonably practicable upon the Participant's completion of the appropriate forms and delivery to the SDBA provider; transfers between the SDBA and the Investment Funds also may be subject to the settlement delays described above in addition to the time for processing the transaction.

- <u>1.03.3 Equity Wash Provisions</u>. In some cases, transfers between Investment Funds and/or the SDBA may require that funds relating to the transaction not be deposited into a competing fund within a certain period of time, typically 90 days. Participant investment direction, transfers and other transactions shall be subject to such so-called equity wash provisions and limited according to the Plan's agreements with the investment fund provider(s) and the Trustee.
- <u>1.04 Legal Limitations</u>. The following limitations 1.04.1 to 1.04.3 on duties and liabilities apply to the Trustee, Retirement Committee, Company and any other individual or entity having investment duties or powers under this Appendix or the Plan, and their respective shareholders, officers, directors, employees, agents and counsel ("protected parties"):
 - <u>1.04.1 No Liability</u>. No protected party shall be liable for any loss or any breach which results from the Participant's exercise of control, by action or inaction, over Participant's Plan account under this Appendix, unless such loss or breach results from such party's own negligence, willful misconduct or failure to act in good faith pursuant to directions of the Participant hereunder received by such party.
 - <u>1.04.2 ERISA</u> § <u>404(c)</u>. The intent of this Appendix is to give the protected parties full protection from liability under ERISA § 404(c), whether the action or inaction by the Participant or protected party is or is not addressed in regulations or other government agency pronouncements.
 - <u>1.04.3 State Law</u>. Investments authorized hereunder shall be made without regard to whether or not such investment is an authorized or appropriate investment for Trustees under any state laws.

Page **9** of **9** Participant Investment Direction Appendix NORDSTROM 401(k) PLAN

APPENDIX B <u>Special Vesting and Matching Contribution Rules for Store Closures</u> to the Nordstrom 401(k) Plan

2021 Restatement

- 1. Plan Provision. The Company has the authority to amend the Plan under Section 15.2 and such authority includes the right to adopt special vesting and contribution rules for Plan participants affected by a store or facility closure. Prior to the adoption of the 2021 Restatement, the Plan provided that the Company's Chief Human Resources Officer had the discretion to adopt such special vesting and contribution rules and the following provisions of this Appendix B memorialize such decisions.
- 2. Triggering Event. Permanent store or facility closures, as listed in the chart on page 2 of this Appendix B.
- 3. Additional Benefits. Nordstrom elected to provide one or both of the additional Plan benefits below to employees affected by the listed closures, as outlined in the chart on page 2 of this Appendix B:
 - (a) Vesting: Participants credited with at least one Year of Service as of their separation date are 100% vested in Company matching contributions allocated to their Plan account prior to the separation date, notwithstanding the vesting schedule under the Plan.
 - (b) Matching Contributions: Participants credited with at least 1,000 Hours of Service during 2011 (as of their separation date) will receive an allocation of matching contributions for 2011, if Nordstrom decides to make a discretionary matching contribution for 2011, notwithstanding that they are not employed on the last day of the Plan Year.
- 4. Conditions. To receive these additional benefits, the participant must have been eligible for, and must have elected to participate in, the Nordstrom Job Elimination or Leadership Separation Program or another severance plan sponsored by the Company.

Store Number additional benefits provided		nal benefits provided	
and Location	Closure Date	Vesting	Matching Contributions
Store 030 Salt Lake City, UT	January 30, 2007	Yes	No
Store 226 Circle Center, IN	July 31, 2011	Yes	Yes
Store 038 Orem, UT	February 24, 2012	Yes	No
Store 318 Long Beach, CA	January 22, 2014	Yes	No

Page 1 of 2
CARES ACT Provisions Appendix NORDSTROM 401(k) PLAN

Store 767 Orlando, FL	August 16, 2014	Yes	Yes
Store 021 Portland, OR	January 10, 2015	Yes	No
Store 026 Vancouver, WA	January 10, 2015	Yes	No
Store 761 Buford, GA	February 28, 2015	Ye	No
Store 362 San Diego, CA	August 26, 2016	Yes	Yes
Store 324 Santa Ana, CA	March 17, 2017	Yes	No
Store 632 Dulles, VA	September 8, 2017	Yes	Yes
Nordstrom Bank	November 1, 2017	Yes	Yes
Store 701 Puerto Rico (temporary closure)	November 4, 2017	Yes	Yes
Store 23 Salem, Or	April 6, 2018	Yes	No
Store 528 Providence Place, RI	January 4, 2019	Yes	No
Store 628 Norfolk, VA	April 5, 2019	Yes	No
Store 766 Wellington, FL	April 5, 2019	Yes	No
Store 2 Northgate, WA	August 9, 2019	Yes	Yes
Store 12 Anchorage, AK	September 13, 2019	Yes	Yes
Store 236 Clinton, MI	September 13, 2019	Yes	Yes
Store 426 San Francisco, CA	September 13, 2019	Yes	Yes
Store 24 Happy Valley, OR	June 21, 2020	Yes	No
Store 35 Broomfield, CO	June 21, 2020	Yes	No
Store 323 Montclair, CA	June 21, 2020	Yes	No
Store 344 Santa Barbara, CA	June 21, 2020	Yes	No
Store 381 Chandler, AZ	June 21, 2020	Yes	No
Store 325 Riverside, CA	June 7, 2020	Yes	No
Store 364 Escondido, CA	June 7, 2020	Yes	No

Store 429			
Sacramento, CA	June 7, 2020	Yes	No
Store 430			N.
Pleasanton, CA	June 7, 2020	Yes	No
Store 722 Hurst, TX	June 7, 2020	Yes	No
Store 512 Queens, NY	January 31, 2020	Yes	No
Store 522 Freehold, NJ	July 12, 2020	Yes	No
Store 626 Annapolis, MD	July 12, 2020	Yes	No
Store 635 Richmond, VA	July 12, 2020	Yes	No
Store 701 Puerto Rico	June 1, 2020	Yes	No
Store 768 Miami, FL	July 12, 2020	Yes	No
Store 775 Naples, FL	July 12, 2020	Yes	No
Just Jeffrey CA, GA, NY	June 1, 2020	Yes	No

Page **3 of 2**CARES ACT Provisions Appendix NORDSTROM 401(k) PLAN

APPENDIX C CARES ACT PROVISIONS

Minimum Required Distributions in 2020.

- 1. Suspension of Required Minimum Distributions for 2020. Notwithstanding section 11.5-2 of the Plan, effective May 8, 2020, a Participant or Beneficiary who would have been required to receive required minimum distributions in 2020 (or paid in 2021 for the 2020 calendar year for a participant with a required beginning date of April 1, 2021) but for the enactment of Code § 401(a)(9)(I) ("2020 RMDs"), and who would have satisfied that requirement by receiving distributions that are either: (1) equal to the 2020 RMDs; or (2) one or more payments (that include the 2020 RMDs) in a series of substantially equal periodic payments made at least annually and expected to last for the life (or life expectancy) of the Participant, the joint lives (or joint life expectancies) of the Participant and the Participant's designated beneficiary, or for a period of at least 10 years ("Extended 2020 RMDs"), the Participant will not receive such distributions unless the Participant elects to receive such distributions in accordance with procedures adopted by the Plan.
- 2. <u>Direct Rollovers</u>. For purposes of the direct rollover provisions of the Plan, the following will also be treated as eligible rollover distributions in 2020:

2020 RMDs and Extended 2020 RMDs.

Coronavirus-Related Distributions.

Notwithstanding any provisions of the Plan to the contrary, the following provisions shall apply effective May 8, 2020 unless otherwise specified herein:

- (a) Coronavirus-Affected Participants may designate all or a portion of a qualifying distribution as a Coronavirus-Related Distribution.
- A "Coronavirus-Related Distribution" means any distribution made from January 1, 2020 to December 30, 2020, to a Coronavirus-Affected Participant, to the extent that such distribution, when aggregated with all other Coronavirus-Related Distributions to the Coronavirus-Affected Participant (including the aggregate amount of such distributions from all plans maintained by the employer and any member of any controlled group which includes the employer), does not exceed \$100,000. A Coronavirus-Related Distribution must be made in accordance with the distribution provisions of the Plan, except that:
 - (1) A Coronavirus-Related Distribution shall be deemed to be made after the occurrence of any distributable events otherwise applicable under Code § 401(k)(2)(B)(i).
 - (2) The requirements of Code §§ 401(a)(31), 402(f), and 3405 shall not apply.
- (d) A Participant who received a Coronavirus-Related Distribution may repay to the Plan in one or more contributions, provided such Coronavirus-Related Distribution is eligible for tax-free rollover treatment. Any such re-contribution:
 - (1) Will be treated as having been made in a direct rollover to the Plan.
 - (2) Must be made during the three-year period beginning on the day after the date on which such distribution was received.

Page 1 of 2
CARES ACT Provisions Appendix NORDSTROM 401(k) PLAN

(3) Cannot exceed the amount of such distribution.

Loans to Coronavirus-Affected Participants.

Notwithstanding any provisions of the Plan to the contrary, the loan repayments that are due for a Coronavirus-Affected Participant during the period beginning on March 27, 2020 and ending not later than December 31, 2020 ("eligible repayment period") shall be suspended until January 2021 ("suspension period"). The loan repayments must resume after the end of the eligible repayment period, and the term of the loan will be extended by the duration of the suspension period. Interest accruing during the suspension period will be added to the remaining principal of the loan. The Coronavirus-Affected Participant shall repay the loan thereafter by amortization in substantially level installments over the remaining period of the loan.

<u>Coronavirus-Affected Participant</u>. A "Coronavirus-Affected Participant" means a participant that meets one of the following requirements:

Who is diagnosed with SARS-CoV-2 or coronavirus disease 2019 (COVID-19) by a test approved by the Centers for Disease Control and Prevention.

- (1) Whose spouse or dependent is diagnosed with the virus or disease.
- (2) Who experiences adverse financial consequences due to the virus or disease as a result of being quarantined; being furloughed, laid off, or having work hours reduced; being unable to work due to the lack of childcare; or the closing or reduction in hours of a business that the individual owns or operates.
- (3) Who experiences an adverse financial consequence as a result of the individual having a job offer rescinded or start date for a job delayed due to COVID-19.
- (4) Who experiences an adverse financial consequence as a result of the individual's spouse or member of the individual's household being quarantined, being furloughed or laid off, having work hours reduced due to COVID-19, being unable to work due to lack of childcare due to COVID-19, having a reduction in pay (or self-employment income) due to COVID-19, or having a job offer rescinded or start date for a job delayed due to COVID-19.
- (5) Who experiences an adverse financial consequence as a result of closing or reducing hours of a business owned or operated by the individual's spouse or a member of the individual's household due to COVID-19.

Page 2 of 2 CARES ACT Provisions Appendix NORDSTROM 401(k) PLAN

NORDSTROM

DEFERRED COMPENSATION PLAN

(2019 Restatement)

Includes All Amendments Approved by the Company since the 2017 Restatement, including:

Amendment 2018-1

Except as otherwise stated herein, the effective date of this 2017 Restatement is January 1, 2019. Amounts deferred and vested prior to January 1, 2005 (and investment gains and losses attributable to such amounts) are governed by the 2003 Restatement and any amendments to the 2003 Restatement. Amounts initially deferred and vested after December 31, 2004 and before January 1, 2008 are subject to the provisions of the 2007 Restatement, except to the extent modified by transition rules separately documented by the Company. Amounts deferred and vested between January 1, 2008 and December 31, 2013 are similarly subject to the provisions of the 2007 Restatement, except as otherwise provided in the 2014 Restatement. Amounts deferred and vested between January 1, 2014 and October 31, 2017 are subject to the provisions of the 2014 Restatement, except as otherwise provided in the 2017 Restatement. Amounts deferred and vested between November 1, 2017 and December 31, 2018 are subject to the provisions of the 2017 Restatement, except as otherwise provided in the 2019 Restatement.

Lane Powell PC 1420 5th Avenue, Suite 4200 Seattle, WA 98101 Telephone: (206) 223-7000 Facsimile: (206) 223-7101

NORDSTROM DEFERRED COMPENSATION PLAN (2019 Restatement)

TABLE OF CONTENTS

ARTIC		<u>1</u>
1.1	Title	<u>1</u>
1.2	Purpose	<u>1</u>
1.3	Effective Date	<u>1</u>
ARTIC	CLE II	<u>1</u>
2.1	Eligible Employee	<u>1</u>
2.2	Entry Date	<u>2</u>
2.3	When Participation Begins	<u>2</u>
2.4	Suspension of Participation	<u>2</u>
2.5	When Participation Ends	<u>2</u>
ARTIC	CLE III	<u>3</u>
3.1	Deferral Elections	<u>3</u>
3.2	Amount of Deferral	<u>4</u>
3.3	Minimum Deferral	<u>6</u>
3.4	Company Contribution Allocations	<u>6</u>
3.5	Deferral of Signing Bonus, Retention Bonus or Separation Payments Prohibited	<u>9</u>
3.6	Requirement for Deferral Agreement	<u>10</u>
3.7	Applicability of Deferral Agreement	<u>10</u>
ARTIC	CLE IV	<u>10</u>
4.1	Account	<u>10</u>
4.2	Time of Crediting Accounts	<u>11</u>
4.3	Participant Deemed Investments	<u>11</u>
4.4	Investments by the Company	<u>12</u>
4.5	Limited Effect of Allocation	<u>12</u>
4.6	Report of Account	<u>12</u>
ARTIC	CLE V	<u>13</u>
5.1	Ownership Rights in Account	<u>13</u>
5.2	Rights in Plan are Unfunded and Unsecured	<u>13</u>
5.3	No Transfer of Interest in Plan Allowed	<u>13</u>
5.4	Plan Binding Upon Parties	<u>13</u>
5.5	Application of Clawback Policy	<u>13</u>
ARTIC	CLE VI	<u>14</u>
6.1	Separation Distributions	<u>14</u>
6.2	In-Service Distributions	<u>14</u>
6.3	Pre-Retirement Separation	<u>16</u>
6.4	Payment Commencement Date	<u>16</u>
6.5	Delayed Payment Date	<u> </u>
		-

6.6	Changing the Time or Form of Distribution	<u>17</u>
6.7	Cash and Stock Distributions	<u>17</u>
6.8	Postponement of Non-Deductible Distributions	<u>17</u>
6.9	Acceleration of Payment	<u>19</u>
6.10	Post-Distribution Allocations	<u>19</u>
ARTIC	CLE VII	<u>19</u>
7.1	Designation of Beneficiary	<u>19</u>
7.2	Married Participants	<u>19</u>
7.3	Deemed Beneficiary	<u>19</u>
7.4	Surviving Beneficiary	<u>20</u>
7.5	Determination of Account Balance at Death	<u>20</u>
7.6	Distribution of Account Balance at Death	<u>20</u>
7.7	Determination of Beneficiary	<u>20</u>
7.8	Payments to Minor or Incapacitated Beneficiaries	<u>20</u>
7.9	Effect of Divorce	<u>21</u>
ARTIC	LE VIII	<u>21</u>
8.1	Plan Sponsor and Administrator	<u>21</u>
8.2	Powers and Authority of the Company	<u>21</u>
8.3	Administrative Committee	<u>21</u>
8.4	Powers and Authority of Leadership Benefits	<u>22</u>
8.5	Reliance on Opinions	<u>23</u>
8.6	Information	<u>23</u>
8.7	Indemnification	<u>23</u>
ARTIC	LE IX	<u>24</u>
9.1	Submission of Claim	<u>24</u>
9.2	Denial of Claim	<u>24</u>
9.3	Review of Denied Claim	<u>24</u>
9.4	Decision upon Review of Denied Claim	<u>24</u>
ARTIC	CLE X	<u>24</u>
ARTIC	CLE XI	<u>25</u>
11.1	No Employment Contract	<u>25</u>
11.2	Employee Cooperation	<u>25</u>
11.3	Illegality and Invalidity	<u>25</u>
11.4	Required Notice	<u>25</u>
11.5	Interest of Participant's Spouse	<u>25</u>
11.6	Tax Liabilities from Plan	<u>25</u>
11.7	Benefits Nonexclusive	<u>26</u>
11.8	Discharge of Company Obligation	<u>26</u>
11.9	Costs of Enforcement	<u>26</u>
11.1	O Gender and Case	<u>26</u>
11.1	1 Titles and Headings	<u>26</u>

11.12	Applicable Law	<u>26</u>
11.13	Counterparts	<u>26</u>
11.14	Additional Definitions	<u>26</u>

Index of Defined Terms

Term	Defined in Section	Page Number
Account	4.1	9
Annual Election Period	2.3(a)	2
Base Compensation	3.1(b)(1)	3
Bonus Compensation	3.1(b)(2)	3
Board	11.14(a)	25
Change in Control	6.8(c)	17
Claiming Party	9.1	22
Code	11.14(b)	25
Common Stock Unit Sub-Account	4.1(b)	10
Company	1.2	1
Deemed Investment Sub-Account	4.1(a)	9
Deferral Agreement	3.1(a)	3
Deferred Retirement Date	11.14(c)	25
Disabled	3.7(b)(2)	9
Early Retirement Date	11.14(d)	25
Election Period	2.3	2
Eligible Compensation	3.1(b)	3
Eligible Employee	2.1	1
Equity Incentive Plan	11.14(e)	25
ERISA	1.2	1
401(k) Plan	11.14(f)	26
Initial Election Period	2.3(b)	2
Leadership Benefits	2.1(c)	2
Make-up Contribution	3.4(a)	6
Normal Retirement Date	11.14(h)	26
Participant	2.3	2
Payment Commencement Date	6.4	15
Performance Share Unit	3.1(b)(3)	4
Plan Administrator	8.1	20
Plan Year	11.14(k)	26
Restoration Contribution	3.4(c)	7
Restricted Stock Unit	3.1(b)(4)	4
Specified Employee	6.5	15
Supplemental Executive Retirement Plan (SERP)	11.14(d)	25
Termination Date	11.14(l)	26
Unforeseeable Financial Emergency	6.2(a)(i)	14
Years of Service	11.14(m)	27

ARTICLE I TITLE, PURPOSE AND EFFECTIVE DATE

- 1.1 <u>Title</u>. This plan shall be known as the Nordstrom Deferred Compensation Plan, and any reference in this instrument to the "Plan" shall include the plan as described herein and as amended from time to time.
- 1.2 <u>Purpose</u>. The Plan is intended to constitute an unfunded plan maintained primarily for the purpose of providing an opportunity for deferred compensation for a select group of management or highly compensated employees, within the meaning of Section 201(2), 301(a)(3) and 401(a)(4) of the Employee Retirement Income Security Act of 1974 ("ERISA"), of Nordstrom, Inc., a Washington corporation, and its Participating Subsidiaries and Affiliates ("Company").
- 1.3 Effective Date. The Plan was originally effective as of January 1, 1994. The Plan was subsequently amended on a number of occasions and most recently was restated effective January 1, 2014. Except as otherwise stated herein, the effective date of this 2017 Restatement is November 1, 2017. Amounts deferred and vested prior to January 1, 2005 (and investment gains and losses attributable to such amounts) are governed by the 2003 Restatement and any amendments to the 2003 Restatement. Amounts initially deferred and vested after December 31, 2004 and before January 1, 2008 are subject to the provisions of the 2007 Restatement, except to the extent modified by transition rules separately documented by the Company. Amounts deferred and vested between January 1, 2008 and December 31, 2013 are similarly subject to the provisions of the 2007 Restatement, except as otherwise provided in the 2014 Restatement, except as otherwise provided in the 2014 Restatement, except as otherwise provided in the 2017 Restatement. Amounts deferred and vested between November 1, 2017 and December 31, 2018 are subject to the provisions of the 2017 Restatement, except as otherwise provided in the 2019 Restatement.

ARTICLE II ELIGIBILITY

- 2.1 <u>Eligible Employee</u>. An "Eligible Employee" means, for any Plan Year, any employee of the Company whose base salary in the Company's Human Resources Information System is \$180,000 or more. Subject to the provisions of the Plan, all Eligible Employees will be eligible to defer compensation and receive benefits at the time and in the manner provided hereunder.
- 2.2 Entry Date. An Eligible Employee shall be eligible to participate in the Plan as follows:
- (a) Eligible Employees who are first hired by the Company during a plan year shall be eligible to participate in the Plan on March 1, June 1 or September 1 following the date he or she first becomes an Eligible Employee.

- (b) All other Eligible Employees shall be eligible to participate in the Plan on January 1 of the year following the year in which he or she became an Eligible Employee.
- 2.3 <u>When Participation Begins</u>. An Eligible Employee becomes a "Participant" in the Plan for the Plan Year when he or she elects to defer a portion of Eligible Compensation (defined in 3.1(b)) during the applicable Election Period pursuant to the terms of the Plan and Article III. The "Election Period" is either the Annual Election Period or, for newly hired and eligible Employees, the Initial Election Period, determined as follows:
- (a) Annual Election Period. "Annual Election Period" means the period designated each year during which Eligible Employees submit their elections to defer compensation. Leadership Benefits has discretion to establish the Annual Election Period and may establish different Annual Election Periods for different types of compensation, provided that annual elections must become irrevocable not later than the time specified under Code Section 409A. A Participant's deferral election with respect to Base Compensation and Bonus Compensation at an Annual Election Period must become irrevocable not later than December 31 of the year preceding the year in which the Participant performs services generating the Base Compensation and the Bonus Compensation.
- (b) <u>Initial Election Period</u>. The Initial Election Period for any employee who first becomes an Employee and Eligible Employee during the Plan Year is the period of thirty (30) days that begins on his or her Entry Date under 2.2. An Eligible Employee's election relates only to Compensation paid for services to be performed subsequent to the election and applies only to Base Compensation. Deferral of Bonus Compensation, Performance Share Units and Restricted Stock Units can be elected only during an Annual Election Period and, for Performance Share Units and Restricted Stock Units, can be elected only if the award agreement underlying the Performance Share Units or Restricted Stock Units specifically includes deferral provisions.
- 2.4 <u>Suspension of Participation</u>. If a Participant receives an unscheduled in-service distribution (with penalty) under the 2003 Restatement of this Plan, the Participant's eligibility to defer under this Plan shall continue for the remainder of the Plan Year in which the unscheduled in-service distribution is received, but shall be suspended for the next two Plan Years.
- 2.5 <u>When Participation Ends</u>. An individual remains a Participant as long as he or she has an Account balance that has not yet been entirely distributed. If, prior to a Participant's Termination Date, a Participant has ceased to be a member of a select group of management or highly compensated employees of the Company within the meaning of Sections 201(2), 301(a)(3) and 401(a)(4) of ERISA, such Participant's deferral elections shall continue for the remainder of the Plan Year to which the deferral elections relate. However, the Participant shall become ineligible to defer compensation under the Plan effective with the next Plan Year, and the Participant shall not re-establish eligibility to defer compensation until such time as he or she once again becomes a member of a select group of management or highly compensated employees and meets the other eligibility requirements set forth in the Plan. The Participant's Account will be distributed at the time and in the form specified by the terms of the Plan and the

Participant's elections. Notwithstanding the forgoing, if an Eligible Employee transfers to a foreign subsidiary of the Company, his or her deferrals under the Plan shall cease upon such transfer and the Participant's Account will be distributed at the time and in the form specified by the terms of the Plan and the Participant's elections.

ARTICLE III DEFERRAL OF COMPENSATION

- 3.1 <u>Deferral Elections</u>. Upon becoming eligible to be a Participant under Section 2.2, and for any Plan Year thereafter (subject to Sections 2.4 and 2.5), an Eligible Employee who wishes to defer compensation under this Plan must properly execute a Deferral Agreement on or before the last day of the applicable Election Period.
- (a) <u>Deferral Agreement</u>. As used in this Plan, the term "Deferral Agreement" means the form prescribed by Leadership Benefits, by which the Participant:
 - (1) indicates and agrees to defer a portion of the Participant's Eligible Compensation for any Plan Year; and
 - (2) specifies the time and form of payment for amounts deferred for the Plan Year.

For this purpose, an Eligible Employee will be considered to have properly executed a Deferral Agreement when he or she has enrolled via an online system, or completed, signed and returned the appropriate form of Deferral Agreement to Leadership Benefits, each in a manner approved by Leadership Benefits.

- (b) <u>Eligible Compensation</u>. For purposes of this Plan, the following items of a Participant's remuneration shall be considered "Eligible Compensation":
 - (1) <u>Base Compensation</u>. A Participant's Base Compensation, which means a Participant's base salary scheduled to be paid in the normal course through the Company's regular payroll cycles (including amounts characterized by the Company as International Premium Pay). Deferrals to this Plan are calculated and deducted before any deferrals under the 401(k) Plan, the Company's cafeteria plan under Code Section 125, and the Company's transportation fringe benefits plan under Code Section 132(f).
 - (2) <u>Bonus Compensation</u>. A Participant's Bonus Compensation, scheduled to be paid to the Participant either in cash or stock. Bonus Compensation means the amount, determined annually based on the Participant's job performance and other factors, that is paid to the Participant in excess of the Participant's Base Compensation.
 - (3) <u>Performance Share Units</u>. A Participant's Performance Share Units as defined in and governed by the Equity Incentive Plan.

(4) <u>Restricted Stock Units</u>. A Participant's Restricted Stock Units as defined in and governed by the Equity Incentive Plan.

Not all forms of Eligible Compensation may be subject to a deferral opportunity. For example, the existence of deferral opportunities for awards of Performance Share Units and Restricted Stock Units depends on whether deferral provisions are included in the agreements underlying such awards.

- 3.2 <u>Amount of Deferral</u>. A Participant may, for any Plan Year, irrevocably elect to have the following amounts of Eligible Compensation deferred and credited to the Participant's Account in accordance with the terms and conditions of the Plan:
- (a) <u>Base Compensation</u>. Effective for deferral elections made after January 1, 2015, all or a portion of the Participant's Base Compensation expressed as either a percentage or a flat dollar amount, provided that, the deferral cannot exceed eighty percent (80%) of the Eligible Employee's Base Compensation. For deferral elections made on or before January 1, 2015, the terms of the Plan in effect prior to this Amendment 2015-1 shall apply. The deferral percentage applied to a Participant's Base Compensation each pay period shall be based on a Participant's annualized Base Compensation over the number of scheduled pay periods during the Plan Year from which deferrals can be taken; this means that the actual deferral made by a Participant in any given Plan Year will not necessarily equal his or her annual Base Compensation multiplied by his or her deferral rate. For example, assume Participant's annualized Base Compensation is \$200,000 and that the election is effective as of the first day of the Plan Year; the maximum annual deferral for such Participant would be \$160,000 (= 80% x \$200,000). Assume further that Participant's deferral election is 40% of Base Compensation and that his or her annualized Base Compensation is scheduled to be paid over 23 pay periods; that Participant's total annualized deferral is \$80,000 and is deferred at a rate of \$3,478.26 (= \$80,000 / 23) per pay period.
- (b) <u>Bonus Compensation</u>. For Participants electing deferrals during an Annual Election Period under 2.3(a), all or a portion of the Participant's Bonus Compensation that is attributable to services to be performed beginning in the Plan Year immediately following the Annual Election Period. Employees who become newly Eligible Employees and who elect to enroll during an Initial Election Period under 2.3(b) may not defer Bonus Compensation payable for the Plan Year during which their enrollment occurs.
- (c) <u>Performance Share Units</u>. All or a portion of a Participant's unvested Performance Share Units awarded by the Company, provided that:
 - (1) The Company makes a deferral opportunity available by including deferral provisions within the "Performance Share Unit Agreement" underlying the award of Performance Share Units;

- (2) The Performance Share Units are scheduled to vest based on the Participant's achievement of individual or organizational performance criteria that are established within the first 90 days of a performance cycle that will last at least 12 months;
 - (3) The deferral election is made at a time when at least six (6) months remain in the performance cycle;
- (4) The Participant provides services continuously for the period from the first day of the performance cycle (or if later, the date the performance criteria are established) through the date that the deferral election is made; and
- (5) The deferral election is made before the amount of the Performance Share Units that will vest is readily ascertainable.
- (d) <u>Restricted Stock Units</u>. All or a portion of a Participant's unvested Restricted Stock Units awarded by the Company, provided that:
 - (1) With respect to an award of Restricted Stock Units that is scheduled to vest based on the Participant's achievement of individual or organizational performance criteria:
 - (A) The Company makes a deferral opportunity available by including deferral provisions within the "Restricted Stock Unit Agreement" underlying the award of Restricted Stock Units;
 - (B) The applicable individual or organizational performance criteria are established within the first 90 days of a performance cycle that will last at least 12 months;
 - (C) The deferral election is made at a time when at least six (6) months remain in the applicable award's performance cycle;
 - (D) The Participant provides services continuously for the period from the first day of the performance cycle (or if later, the date the performance criteria are established) through the date that the deferral election is made: and
 - (E) The deferral election is made before the amount of the Restricted Stock Units that will vest under the applicable award is readily ascertainable.
 - (2) With respect to an award of Restricted Stock Units that are scheduled to vest based solely on the lapse of time:

- (A) The Company makes a deferral opportunity available by including deferral provisions within the "Restricted Stock Unit Agreement" underlying the award of Restricted Stock Units; and
- (B) The deferral election must be made by the end of the Plan Year immediately preceding the Plan Year in which the award of Restricted Stock Units is initially granted.
- 3.3 <u>Minimum Deferral</u>. Effective for Plan Years beginning on and after January 1, 2014, there is no minimum deferral limitation under this Plan.
- 3.4 <u>Company Contribution Allocations</u>. The following Company contributions are permitted under the Plan:
- (a) <u>Make-up Contribution</u>. Each Plan Year, the Company shall allocate to each Participant's Account an amount corresponding to the Participant's lost share of Company contributions to its 401(k) Plan, determined as follows:
 - (1) an amount, if any, equal to such Participant's lost share of non-elective contributions under the 401(k) Plan; and
 - (2) an amount, if any, equal to such Participant's lost share of matching contributions under the 401(k) Plan.

For purposes of this allocation, a Participant's "lost share" of non-elective and matching contributions is the amount of contributions not allocated to Participant's 401(k) Plan account because of:

- (A) The reduction in the Participant's compensation (as defined under the Participant's 401(k) Plan) by reason of deferrals under this Plan, or
- (B) The Participant's exclusion from receiving a Company non-elective contribution under the Participant's 401(k) Plan on account of being considered "otherwise excludible" under Code section 410(b)(4).

The time and form of payment of Make-up Contributions shall be determined by the Participant's elections with respect to time and form applicable for the Plan Year preceding the Plan Year in which the Make-up Contribution is credited to the Participant's Account. For example, the time and form of payment of Make-up Contributions credited in early 2018 with respect to Participant's compensation deferred in the 2017 Plan Year shall be determined by the Participant's elections with respect to time and form applicable for the 2017 Plan Year. If no such election exists, then the time and form of payment of the Participant's Make-up Contribution for such Plan Year shall be as a single lump sum payment made at Participant's Separation.

Effective January 1, 2019, all Make-up Contributions shall become immediately one-hundred percent (100%) vested.

For the avoidance of doubt, to receive a Make-up Contribution with respect to a given Plan Year, the Participant must have made a deferral under this Plan for such Plan Year.

- (b) <u>Company Discretionary Contributions</u>. In addition to any Company contributions made in accordance with 3.4(a), the Company may, in its sole discretion, make discretionary contributions to the Accounts of one or more Participants at such times, in such amounts, and vested in such manner, as the Board or the Compensation Committee may determine. Such discretionary contributions shall be credited to the applicable Participant's Deemed Investment Sub-Account. The Company must designate the time and form of distribution at the time that the discretionary contributions are allocated to the Participant's Account.
- (c) <u>Restoration Contributions</u>. Beginning with Plan Years commencing January 1, 2014, the Company shall allocate to certain Participants' Accounts a Restoration Contribution, which shall be based on each Participant's Excess Compensation (defined below).

A Participant's "Excess Compensation" for Restoration Contribution allocation purposes means the excess of a Participant's Unlimited 401(k) Plan Compensation (defined below) over the Participant's actual 401(k) Plan Compensation for that Plan Year. Moreover, "Excess Compensation" shall exclude performance-based or other incentive compensation received by a Participant that both (i) relates to the economic performance of an entity other than Nordstrom, Inc. and (ii) was adopted as part of, in recognition of, or in concert with, the merger, acquisition or change in control of such entity.

A Participant's "Unlimited 401(k) Plan Compensation" for Restoration Contribution allocation purposes means Participant's 401(k) Plan Compensation for a Plan Year determined without regard to the 401(a)(17) Limit (defined below) **plus** the amount deferred by Participant into this Plan during that Plan Year. The 401(a)(17) Limit for a Plan Year means the compensation limitation under Code section 401(a)(17) (or the limit under Section 1081.01(a)(12) of the Puerto Rico Internal Revenue Code (the "PR Code"), whichever applies) in effect for such Plan Year. For the Plan Year beginning January 1, 2018, the 401(a)(17) Limit is \$275,000 and is thereafter indexed for inflation.

Example 1: Assume that for the 2018 Plan Year, Participant A is also a participant in the Nordstrom 401(k) Plan & Profit Sharing (the "Qualified Plan" for purposes of this Section 3.4(c)). During the 2018 Plan Year, Participant A's 401(k) Plan Compensation was \$275,000 and Participant A deferred \$10,000 into this Plan. The 401(a)(17) Limitation in effect for the 2018 Plan Year was \$275,000. Participant A's 2018 401(k) Plan Compensation determined without regard to the 401(a)(17) Limit was \$305,000. Consequently, Participant A's Unlimited 401(k)

Plan Compensation for the 2018 Plan Year was \$315,000 (\$305,000 plus \$10,000). Participant A's Excess Compensation was \$40,000 (\$315,000 less Participant's \$275,000 401(k) Plan Compensation).

The Restoration Contribution allocable with respect to a Participant's Excess Compensation shall be the lesser of:

- (1) the maximum matching contribution amount that could be generated by applying the matching contribution formula in effect under the Participant's 401(k) Plan for such Plan Year to the Participant's Excess Compensation, if any; and
 - (2) the amount actually deferred by Participant into this Plan for such Plan Year, if any.

Example 2: Same facts as in Example 1. Assume further that the matching formula under the Qualified Plan was 100% of Participant's elective deferrals under the Qualified Plan, up to 4% of Participant's 401(k) Plan Compensation. From Example 1, Participant A's Excess Compensation for the 2018 Plan Year was \$40,000. Applying the Qualified Plan's matching contribution to Participant A's Excess Compensation, the maximum match generated by the Excess Compensation would be \$1,600 (i.e., dollar for dollar, up to 4% of Participant's Excess Compensation). Accordingly, the Restoration Contribution allocable to Participant A under this Plan with respect to the 2018 Plan Year would be \$1,600 (the lesser of (i) the maximum matching contribution generated by Participant A's Excess Compensation and (ii) Participant A's \$10,000 Plan deferral.)

In the event that the Participant is eligible to receive matching contributions under Participant's 401(k) Plan under more than one formula during a given Plan Year, then the Restoration Contribution above shall be calculated through application of each applicable 401(k) Plan matching formula under (1) and (2) above, with the resulting amounts added together to arrive at the total Restoration Contribution for that Plan Year.

Example 3: Same facts as in Example 2. Assume further that a second matching contribution is declared under the Qualified Plan for the 2018 Plan Year. The formula for this second matching contribution was 50% of Participant's elective deferrals under the Qualified Plan, up to 4% of Participant's 401(k) Plan Compensation. Applying the Qualified Plan's second matching contribution to Participant A's Excess Compensation, the maximum match generated by the Excess Compensation would be \$800 (i.e., fifty percent (50%) of the lesser of (i) the amount deferred into this Plan or (ii) 4% of Participant's Excess Compensation). Accordingly, the Restoration Contribution allocable to Participant A under this Plan with

respect to the 2018 Plan Year would be \$800 (the lesser of (i) the maximum matching contribution generated by Participant A's Excess Compensation and (ii) Participant A's \$10,000 Plan deferral.)

Participant A's total Restoration Contribution for the 2018 Plan Year would be \$2,400 (\$1,600 under the first matching contribution formula plus \$800 under the second matching contribution formula).

The time and form of payment of Restoration Contributions shall be determined by the Participant's elections with respect to time and form of payment applicable for the Plan Year preceding the Plan Year in which the Restoration Contribution is credited to the Participant's Account. For example, the time and form of payment of Restoration Contributions credited in early 2018 with respect to Participant's Excess Compensation earned in the 2017 Plan Year shall be determined on the Participant's elections with respect to time and form of payment applicable for the 2017 Plan Year. If no such election exists, then the time and form of payment of the Participant's Restoration Contribution for such Plan Year shall be as a single lump sum payment made at Participant's Separation. Effective January 1, 2019, Restoration Contributions will be immediately one-hundred percent (100%) vested.

A Participant is ineligible to receive a Restoration Contribution for any Plan Year in which such Participant either (i) is ineligible to receive a Company matching contribution allocation under the 401(k) Plan due to application of the 401(k) Plan's employment and/or hours of service requirements to receive such matching contribution allocation or (ii) is a participant in the SERP, unless the Compensation Committee determines otherwise.

For the avoidance of doubt, (x) to receive a Restoration Contribution with respect to a given Plan Year, the Participant must have made a deferral under this Plan for such Plan Year and (y) a Participant can receive a Restoration Contribution under this Plan with respect to a given Plan Year whether or not the Participant made a deferral election under the 401(k) Plan for such Plan Year.

3.5 <u>Deferral of Signing Bonus, Retention Bonus or Separation Payments Prohibited</u>. A Participant may not defer any amounts paid to the Participant that are designated by the Company as a signing bonus, a retention bonus, or separation payments. A "signing bonus" is any amount paid to a newly hired Employee specifically as an incentive to accept a position with the Company. A "retention bonus" is any amount paid to an existing Employee specifically in exchange for an agreement to remain an Employee of the Company for a specified period. A "separation payment" is any amount paid to an Employee as a result of termination of employment with the Company; provided, however, that nothing in this Section 3.5 shall prevent the Company from negotiating a separation agreement, the provisions of which include a Company Discretionary Contribution under Section 3.4(b).

3.6 <u>Requirement for Deferral Agreement</u>. A Participant who has not timely submitted a valid Deferral Agreement may not defer any Eligible Compensation (or receive the corresponding Company Make-up Contribution or Restoration Contribution allocation under 3.4) for the applicable Plan Year under the Plan.

3.7 <u>Applicability of Deferral Agreement</u>.

(a) <u>General Rule</u>. Except as provided in this Section 3.7, a Deferral Agreement shall be irrevocable and remains in effect for the entire Plan Year to which it applies. A Participant must file a new Deferral Agreement to continue deferrals in any subsequent Plan Year. The terms of any Deferral Agreement may, but need not be, similar to the terms of any prior Deferral Agreement.

(b) Exceptions to Irrevocability.

- (1) <u>Financial Hardship</u>. A Participant's Deferral Agreement shall be automatically canceled and deferrals shall cease for the remainder of the Plan Year if the Participant receives a distribution due to an unforeseeable financial emergency, as described in Section 6.2(a)(1).
- (2) <u>Disability</u>. A Deferral Agreement shall be canceled if a Participant becomes Disabled. For purposes of this section, "Disabled" means that a Participant is determined to be eligible for benefits under the Company's long-term disability plan then in effect or is determined to be disabled by the Social Security Administration.
- (c) <u>Resuming Participation</u>. A Participant may elect to resume deferrals under this Plan at any subsequent Annual Election Period, provided that the Participant satisfies the Plan's eligibility requirements in effect at that time.

ARTICLE IV DEFERRAL ACCOUNT AND CREDITING

- 4.1 <u>Account</u>. A Participant's "Account" is the account established on the books of the Company as a record of each Participant's Plan balance. An Account may, at the discretion of the Administrative Committee, include one or more subaccounts to reflect amounts credited to a Participant under the various terms of the Plan. As of the effective date of this Restatement, the Administrative Committee has established the following sub-accounts:
- (a) <u>Deemed Investment Sub-Account</u>. A Deemed Investment Sub-Account, reflecting the Participant's account balance resulting from the deferral of Eligible Compensation (other than Performance Share Units, Restricted Stock Units or other stockbased compensation), Company Contribution allocations under Section 3.4, and the Participant's deemed investment of

such amounts under Section 4.3. The balance in such sub-account shall be expressed as a dollar amount.

- (b) <u>Common Stock Unit Sub-Account</u>. A Common Stock Unit Sub-Account reflecting the number of Performance Share Units, Restricted Stock Units, or other stock-based compensation in which the Participant is vested and which the Participant has deferred under the Plan. The balance in such sub-account shall be expressed in units, with each unit representing the value of one share of the Company's Common Stock.
- 4.2 <u>Time of Crediting Accounts</u>. Amounts deferred by a Participant under the Plan and any Company Contribution allocations made on behalf of that Participant shall be credited to the Participant's Account as soon as administratively practicable after the date deferred amounts would otherwise have been received (or beneficially received in the case of Company contributions) by the Participant. Subject to 4.4(c)(2) regarding the underwriting of the Plan's investment vehicles, Earnings shall be credited to a Participant's Account on the date determined by the Administrative Committee, but no later than the month following the month in which deferrals and Company contributions were credited to the Account in accordance with the preceding sentence. Earnings are based on the performance of the investment options selected by Participants in accordance with Section 4.3.
- 4.3 <u>Participant Deemed Investments</u>. Subject to Section 4.3(b), each Participant may, from time to time, select from the various indices provided by the Administrative Committee (under Section 4.4(b)) in which his or her Account will be deemed invested; provided, however, that the Administrative Committee is under no obligation to acquire or provide any of the investments designated by the Participant.
- (a) <u>Deemed Investment Sub-Account Valuation</u>. A Participant's Deemed Investment Sub-Account shall be credited or debited from time to time, as determined by the Administrative Committee, with additional amounts equal to the appreciation (or loss) such accounts would have experienced had they actually been invested in the specified fund indices at the relevant times. This crediting and debiting will take into account the date that a Participant's Account transactions (such as deferrals, contributions, distributions and transfers among funds) are actually reflected by the Plan's record-keeping system.
- (b) <u>Common Stock Unit Sub-Account Valuation</u>. The number of units in a Participant's Common Stock Unit Sub-Account shall be appropriately adjusted periodically to reflect any dividend, split, split-up or any combination or exchange, however accomplished, with respect to the shares of the Company's Common Stock represented by such units.
- 4.4 <u>Investments by the Company</u>. In order to provide funds to satisfy its obligations under the Plan, the Company may, but shall not be required to, keep cash or invest and reinvest in mutual funds, stocks, bonds, securities or any other assets as may be reasonably selected by the Administrative Committee in its discretion. Such investments may, but need not, follow the investment indices chosen by the Participants.

- (a) <u>Investment Advice</u>. In the exercise of the foregoing investment powers, the Administrative Committee may engage investment consultants and, if the Administrative Committee so desires, may delegate to such consultants full or limited authority to select the assets in which the funds are to be selected. Investment consultants may be officers or employees of the Company or outside consultants.
- (b) <u>Choice of Investment Indices</u>. The Administrative Committee, or its investment consultants, may specify one or more investment funds to serve as indices for the investment performance of amounts credited under the Accounts. The Administrative Committee has the authority to expand or limit the type or number of fund indices and to prescribe, in conjunction with Leadership Benefits, the frequency with which Participants may change their deemed investment elections.
- (c) <u>Insurance</u>. If the Administrative Committee elects to purchase an insurance policy or policies insuring the life of the Participant to allow the Company to recover the cost of providing the benefits hereunder:
 - (1) The Participant shall, as a condition to continued participation in the Plan, sign any papers and undergo any medical examinations or tests that may be necessary or required for such purpose;
 - (2) Notwithstanding the Participant's election or direction or any provision in the Plan to the contrary, the Participant's Account may be deemed invested in a money market fund or instrument or other liquid asset selected by the Administrative Committee or its delegate, pending the underwriting and delivery of such policy or annuity; and
 - (3) The Participant, Participant's Beneficiary, and any other person claiming through the Participant shall not have or acquire any rights whatsoever in such policy or policies or in the proceeds of the policies.
- 4.5 <u>Limited Effect of Allocation</u>. The fact that any allocation shall be made and credited to an Account shall not vest in a Participant any right, title or interest in or to any assets of the Company, or in any right to payment, except at the time(s) and upon the conditions elsewhere set forth in the Plan.
- 4.6 Report of Account. A Participant shall be provided information regarding Participant's Bookkeeping Account balance within a reasonable time after requesting such information from Leadership Benefits. Leadership Benefits shall furnish each Participant statements on a periodic basis, no less frequently than annually, as soon as administratively practicable after the allocations for the end of the Plan Year have been completed. Leadership Benefits may, in its discretion, provide Participants with account balance statements more frequently than provided in the preceding sentence.

ARTICLE V RIGHTS OF PARTICIPANT IN PLAN

- 5.1 <u>Ownership Rights in Account</u>. Subject to the restrictions provided in this Article and in Section 3.2(c), each Participant shall at all times have a vested right to the value of such Participant's Account.
- 5.2 <u>Rights in Plan are Unfunded and Unsecured</u>. The Company's obligation under the Plan shall in every case be an unfunded and unsecured promise to pay. A Participant's right to Plan distributions shall be no greater than the rights to payment of general, unsecured creditors of the Company. The Company may establish one or more grantor trusts (as defined in Code Section 671 et seq.) to facilitate the payment of benefits hereunder; however, the Company shall not be obligated under any circumstances to fund its financial obligations under the Plan. Any assets which the Company may acquire or set aside to defray its financial liabilities shall be subject to the claims of its general creditors in the event of the Company's insolvency.
- 5.3 <u>No Transfer of Interest in Plan Allowed</u>. Except as permitted by applicable law, no sale, transfer, alienation, assignment, pledge, collateralization or attachment of any benefits under the Plan shall be valid or recognized by the Company. The Participant, the Participant's spouse and a designated Beneficiary shall not have any power to hypothecate, mortgage, commute, modify or otherwise encumber in advance any of the benefits payable under the Plan. Said benefits shall not be subject to seizure for the payment of any debts, judgments, alimony, maintenance owed by the Participant or a Beneficiary, or be transferable by operation of law in the event of bankruptcy, insolvency, or otherwise. Notwithstanding the foregoing, the Company may, if the Administrative Committee so determines in its sole discretion, follow the terms of any court order issued in connection with any domestic relations proceeding including but not limited to marital dissolution or child support.
- 5.4 <u>Plan Binding Upon Parties</u>. The Plan shall be binding upon the Company, its assigns, and any successor company that acquires substantially all of its assets and business through merger, acquisition or consolidation; and upon all Participants and any Participant's Beneficiaries, assigns, heirs, executors and administrators.
- 5.5 <u>Application of Clawback Policy</u>. This section applies if the Board elects to apply the compensation recovery policy contained in the Nordstrom Corporate Governance Guidelines (the "Clawback Policy") to a Participant. To the extent that any amount in a Participant's Account is attributable to contributions based on compensation that is subject to recovery under the Clawback Policy, such amount (adjusted for investment gains and losses) shall be removed from the Participant's Account and shall be permanently and irrevocably forfeited. The provisions of this section for removal of amounts from a Participant's Account shall also apply to the Beneficiary of a Participant after the Participant's death.

ARTICLE VI DISTRIBUTIONS

6.1 <u>Separation Distributions</u>.

- (a) <u>Separation Events</u>. A Participant may elect in a Deferral Agreement to receive a distribution of his or her Account at Separation. A Participant's "Separation" shall mean the Participant's Termination Date.
- (b) <u>Separation Distribution Forms</u>. Distribution of a Participant's Account balance shall be made according to the distribution options specified in the Participant's Deferral Agreement(s). Portions of Accounts subject to installment payment shall continue to be valued as provided in Section 4.3 until distributed. The distribution options available to a Participant are:
 - (1) single lump sum payment; or
 - (2) installment payments for a period of five (5), ten (10) or fifteen (15) years. The amount of lump sum payments under this subsection (b) shall be determined as of the last day of the month in which the Participant's Termination Date occurs.
- (c) <u>Lump Sum in Lieu of Installments</u>. If the Participant's Account balance as of his or her Separation is equal to or less than \$10,000, Leadership Benefits may order the distribution of the Participant's entire Account in a single lump sum rather than in installments, provided that the lump sum payment results in the termination and liquidation of the Participant's entire interest under this Plan and all other plans or arrangements that must be aggregated with this Plan under the rules set forth under Code Section 409A. The Participant may not exercise any discretion to convert an installment election into a lump sum under this provision.
- (d) Amount and Timing of Installment Payments. The first installment shall be paid on the Payment Commencement Date as defined in 6.4. Subsequent installments shall be paid annually in January of each succeeding year. The amount of each installment shall be determined by multiplying the Participant's account balance as of the end of the month in which the scheduled distribution date occurs (as determined under Section 6.2(b) for In-Service Distributions or upon Separation for all other distributions) by a fraction, the numerator of which is one (1) and the denominator of which is (N minus P), where N is the total number of annual installments and P is the number of annual installments previously paid to the Participant. For example, if the form of payment is five annual installments, the first annual distribution is the account balance divided by 5 (5 minus 0), the second annual distribution is the account balance divided by 4 (5 minus 1), the third annual distribution is the account balance divided by 2 (5 minus 3), and the fifth annual distribution is the entire remaining account balance (5 minus 4).

- 6.2 <u>In-Service Distributions</u>. While a Participant is employed by the Company, a subsidiary or affiliate, the Participant may receive in-service Plan distributions as provided in this Section 6.2.
- (a) <u>Unforeseeable Financial Emergency</u>. At the request of a Participant, the Administrative Committee may, in its sole discretion, pay all or part of the value of the Participant's Account in the event of an unforeseeable financial emergency.
 - (1) <u>Financial Emergency</u>. In this context, an "unforeseeable financial emergency" is defined as a severe financial hardship resulting from one of the following:
 - (A) illness or accident of the Participant, the Participant's spouse or dependent (as defined in Code Section 152(a)), or the Participant's designated Beneficiary;
 - (B) loss of the Participant's property due to casualty; or
 - (C) other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant.
 - (2) <u>Amount</u>. The amount of an accelerated distribution shall be limited to an amount necessary to relieve such emergency, which may include amounts necessary to pay any federal, state, and local taxes or penalties reasonably anticipated to result from the distribution. Amounts available to the Participant due to the cancellation of the Participant's deferral election for the remainder of the Plan Year must be taken into account in determining the amount necessary to satisfy the emergency need. If the Participant's entire Account balance is distributed pursuant to this Section 6.2(a), the amount of the distribution shall be determined as of the end of the month preceding the distribution date.
 - (3) <u>Effect of Other Financial Resources</u>. A distribution on account of unforeseeable emergency may not be made to the extent that such emergency is or may be alleviated by reimbursement or compensation from insurance, liquidation of other assets (provided that the liquidation would not itself cause a severe hardship), or the cancellation of deferrals for the remainder of the Plan Year under the Plan.
 - (4) <u>Distribution Hierarchy</u>. If a Participant qualifies for a distribution due to unforeseeable financial emergency, the Participant must first exhaust amounts available from his or her paid-time off bank under the Company's Sabbatical Program before receiving a distribution from this Plan.
- (b) <u>Scheduled Distributions</u>. During any Election Period, a Participant may, in connection with his or her election to defer compensation, specify a withdrawal date for all or part of his or her compensation deferred pursuant to the election made during the Election Period. A Participant's scheduled distribution election must specify a distribution date that

occurs after the Participant's deferrals that are subject to the election have been in the Plan for at least two complete Plan Years (for example, deferrals elected during the 2007 Annual Election Period can first be scheduled for distribution in 2011). The Participant must elect the calendar year and the month (either January or June) of the scheduled distribution. The amount payable to a Participant in connection with a scheduled distribution shall in all cases be a specified dollar amount or a specified percentage of the Participant's Account balance for the Plan Year to which the Deferral Agreement applies. If a distribution event occurs with respect to a Participant before the scheduled distribution date, the Plan provisions applicable to the distribution event will take precedence over the Participant's scheduled distribution election. The amount of the distribution under this subsection (b) shall be determined as of the last day of the month before the scheduled distribution.

- 6.3 <u>Pre-Retirement Separation</u>. For Plan Years commencing prior to January 1, 2014, and for the portion of a Participant's Account that is attributable to elective deferrals and Company contributions credited to the Account for Plan Years ending through December 31, 2013, if a Participant's Termination Date occurs prior to his or her Early Retirement Date, Normal Retirement Date, or Deferred Retirement Date (a "Pre-Retirement Separation"), the time and form of payment elections in the Participant's Deferral Agreements shall be disregarded and, in lieu of those elections, the Participant shall receive the value of his or her Account in a single lump sum payment on the Payment Commencement Date set forth in Section 6.4(b) and the amount of the distribution shall be determined as of the last day of the month in which the Participant's Termination Date occurs. Commencing on and after January 1, 2014, the provisions of this Section 6.3 shall not apply to the portion of a Participant's Account that is attributable to deferrals and Company contributions made with respect to Plan Years commencing January 1, 2014 and thereafter, including Earnings thereon; instead, this portion of a Participant's Account shall at all times be distributable as provided in Section 6.1(b) (subject to 6.1(c)).
- 6.4 <u>Payment Commencement Date</u>. Distributions will begin to be paid on the following dates, subject to the delay for Specified Employees set forth in 6.5.
- (a) <u>Scheduled Distribution</u>. During the calendar month (January or June) and year specified by the Participant in his or her deferral election.
- (b) <u>Separation Distributions</u>. Within 90 days after Leadership Benefits confirms the Separation, provided that the Participant does not have the right to designate the taxable year of payment.
- (c) <u>Unforeseeable Financial Emergency</u>. Within 90 days after Leadership Benefits receives confirmation of the amount of distribution approved by the Administrative Committee, provided that the Participant does not have the right to designate the taxable year of payment.
- 6.5 <u>Delayed Payment Date</u>. If a distribution is made to a Specified Employee following his or her Separation, the first payment may not be made earlier than six months after the Specified Employee's Payment Commencement Date. If the form of distributions is installments, any

installments that would have been paid in the absence of this six-month delay will be accrued and paid at the end of the six-month period. Any installments that are due after the six-month period expires will be paid as if they were not subject to this provision. A Specified Employee means an individual who meets the requirements to be a "key employee" as defined in Code Section 416(i) (without regard to Section 416(i)(5)). If the individual is a key employee as of September 30 of a given year, the individual is treated as a Specified Employee for the entire next calendar year. This delayed payment date rule does not apply to scheduled in-service distributions, financial emergency distributions, or distributions due to the Participant's death.

- 6.6 <u>Changing the Time or Form of Distribution</u>. The time and form of payment elected in a Participant's Deferral Agreements cannot be changed by the Participant after the last day of an Election Period except as provided in this section. A Participant may change his or her form of Retirement distribution under 6.1(b) or the timing of a scheduled in-service distribution under 6.2(b), provided that:
- (a) For a scheduled in-service or Separation distribution, his or her change is filed with Leadership Benefits no later than the last day of the Plan Year that ends at least 12 months before the Payment Commencement Date;
 - (b) His or her change cannot take effect earlier than twelve months after the change is requested; and
- (c) the first payment under the newly elected form of payment cannot be made sooner than five years after the Payment Commencement Date for the form of payment that the Participant has elected to change.

The Payment Commencement Date for a series of installment payments is treated as the date on which the first of such installment payments would be made under the terms of this Plan. Where the Payment Commencement Date is stated as a period of time (e.g., a 90-day period following a distribution event), the Payment Commencement Date for purposes of this section is the first day of such period.

- 6.7 <u>Cash and Stock Distributions</u>. Distributions of a Participant's Deemed Investment Sub-Account Account shall be made in cash only. Distributions of a Participant's Common Stock Unit Sub-Account shall be made in Common Stock of the Company.
- 6.8 Postponement of Non-Deductible Distributions.
- (a) When Applicable. If the Administrative Committee determines in good faith prior to a Change in Control that there is a reasonable likelihood that any Compensation paid to a Participant for a taxable year of the Company would not be deductible by the Company solely by reason of the limitation under Code section 162(m), then to the extent deemed necessary by the Administrative Committee to ensure that the entire amount of any distribution to the Participant pursuant to this Plan prior to the Change in Control is deductible, the Administrative Committee may defer all or any portion of the distribution. After a Change in Control, the Administrative

Committee shall not have discretion to postpone payments under this provision, and all payments will be made on the dates provided in the Plan.

- (b) <u>Administration of Deferred Distributions</u>. Any distributions deferred pursuant to this limitation shall continue to be credited with interest or earnings pursuant to the terms hereof. Where a payment to a Participant is delayed under this provision, all other payments to that same Participant that could be delayed under this provision must also be delayed. The amounts so deferred and interest thereon shall be distributed to the Participant or his or her Beneficiary (in the event of a death benefit required hereunder) at the earliest possible date, as determined by the Administrative Committee in good faith, on which the deductibility of compensation paid or payable to the Participant for the taxable year of the Company during which the distribution is made will not be limited by Code section 162(m), or if earlier, the effective date of a Change in Control.
- (c) <u>"Change in Control" Defined.</u> For purposes of this Plan, Change in Control means the first of the following (1), (2), or (3) to occur.
 - (1) <u>Change in Ownership of Stock</u>. Any person, entity or group of persons purchases or acquires, within the meaning of section 13(d) or 14(d) of the Securities Exchange Act of 1934 ("Act"), or any comparable successor provisions, beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Act) of Company stock that, together with stock already held by such person, entity, or group, constitutes more than 50% of the total fair market value or total voting power of the stock of the Company.
 - (2) <u>Change in Effective Control</u>. Either of the following occurs, representing a change in effective control of the Company:
 - (A) <u>Voting Power</u>. Any one person, or more than one person acting as a group, acquires (or has acquired during the 12-month period ending on the date of the most recent acquisition by the person or group) ownership of Company stock constituting 30% or more of the total voting power of Company stock; or
 - (B) <u>Board Composition</u>. A majority of the members of the Company's Board of Directors is replaced during any period of 12 consecutive months by directors whose appointment or election is not endorsed by a majority of the members of the Company's Board of Directors prior to the date of the appointment or election.
 - (3) <u>Change in Ownership of Assets</u>. Any one person, or more than one person acting as a group, acquires (or has acquired during the 12-month period ending on the date of the most recent acquisition by such person or persons) Company assets that have a total gross fair market value equal to or greater than 40% of the total gross fair market value of all of the assets of the Company immediately prior to such acquisition or acquisitions. Gross value means the value of the assets determined without regard to any

liabilities associated with such assets. However, a Change in Control does not occur to the extent that ownership of assets is transferred to:

- (A) a Company shareholder (immediately before the asset transfer) in exchange for or with respect to his or her Company stock;
- (B) an entity, 50% or more of the total value or voting power of which is owned directly or indirectly by the Company;
- (C) a person, or more than one person acting as a group, that owns directly or indirectly 50% or more of the total value or voting power of the Company;
- (D) an entity, at least 50% of the total value or voting power of which is owned directly or indirectly by a person described in (C).
- (4) <u>Interpretation</u>. These provisions shall be interpreted and applied in a manner that is consistent with Department of Treasury regulations under Section 409A of the Code.
- 6.9 <u>Acceleration of Payment</u>. Generally, neither the Company nor any Participant may accelerate the timing of any payment under the Plan, except as specifically set forth in this Plan document. However, the Administrative Committee retains the discretion to accelerate distribution of any payment to the extent such acceleration is specifically permitted under the final regulations under Code Section 409A. Such accelerations include, but are not limited to, a distribution to permit a Participant to pay taxes on amounts deferred under this Plan, including any taxes that may be imposed under Code Section 409A, and distribution pursuant to a domestic relations order (as defined in Section 414(p)(1)(B) of the Code).
- 6.10 <u>Post-Distribution Allocations</u>. If a Participant's Account is credited after the Participant has received a full distribution of his or her Account, the remaining balance in the Account shall be paid to the Participant in a lump sum as soon as administratively practical.

ARTICLE VII DEATH BENEFITS

7.1 <u>Designation of Beneficiary</u>. A Participant shall designate a Beneficiary to receive death benefits under the Plan by completing the beneficiary designation form specified by the Administrative Committee. A Participant shall have the right to change the Beneficiary by submitting a form designating the Participant's change of Beneficiary in accordance with procedures established by the Administrative Committee. No beneficiary designation or change of beneficiary shall be effective until approved by Leadership Benefits.

- 7.2 <u>Married Participants</u>. If a Participant is married, his or her legal spouse shall be the designated Beneficiary, unless the spouse consents in writing to designation of a different Beneficiary on a form acceptable to the Administrative Committee.
- 7.3 <u>Deemed Beneficiary</u>. For Participants who die on or after January 1, 2011, if a valid beneficiary designation has not been made, or if the designated beneficiary has predeceased the Participant, then the Participant will be deemed to have designated the following as his or her surviving beneficiaries and contingent beneficiaries with priority in the order named below:
- (a) first, to the Participant's surviving spouse, as defined under federal law, or the Participant's registered life partner, as defined under the Nordstrom Welfare Benefit Plan; or
 - (b) if the Participant does not have a surviving spouse or registered life partner, to his or her estate.

Notwithstanding the foregoing, the 2007 Restatement of this Plan, and not this Restatement, shall apply to determine beneficiaries of Participants who died prior to January 1, 2011.

- 7.4 <u>Surviving Beneficiary</u>. For purposes of determining the appropriate named or deemed beneficiary or contingent beneficiary, an individual is considered to survive the Participant if that individual is alive seven (7) days after the date of the Participant's death.
- 7.5 <u>Determination of Account Balance at Death</u>. The value of a Participant's Account shall be determined as of the later of: (a) the date of the Participant's death; or (b) the date the Administrative Committee approves the distribution under Section 7.6. The amounts in such Account shall be maintained in the deemed investment Sub-Accounts under Section 4.3 after the Participant's death and until the time of distribution, unless the Participant's Beneficiary elects in writing to transfer such amounts from the deemed investment accounts into a separate interest-bearing account designated by the Administrative Committee for this purpose. Upon transfer to the interest-bearing account, the Account shall no longer be deemed invested under Section 4.3(a) and will not be adjusted for deemed investment gains and losses.
- Distribution of Account Balance at Death. Upon a Participant's death, the portion of a Participant's Account that is attributable to deferrals and Company contributions made with respect to Plan Years commencing January 1, 2014 and later, including Earnings thereon, shall at all times be distributable as provided in Section 7.6. Distributions of the portion of a Participant's Account that is attributable to deferrals and Company contributions made with respect to Plan Years ending prior to January 1, 2014 shall be governed by the Plan provisions in effect in the previous version of this Plan document. By way of reference, and as provided in Section 7.6 of the previous version of this Plan document, if the Participant dies prior to "Retirement" (as that term is defined in the previous version of this Plan document) while an employee of the Company and such Participant's death is not attributable to suicide committed within two years after becoming a Participant, such Beneficiary shall receive an amount equal to twice the Participant's actual deferrals under Section 3.2 that have been credited to the Participant's Account as of December 31, 2007 (exclusive of any earnings thereon).

Compensation deferred after December 31, 2007 shall not be taken into account in calculating this pre-retirement death benefit.

- 7.7 <u>Determination of Beneficiary</u>. If the Administrative Committee has any doubt as to the proper Beneficiary to receive payments hereunder, the Administrative Committee shall have the right to direct the Company to withhold such payments until the matter is finally adjudicated. However, as provided in Section 11.8, any payment made by the Company, in good faith and in accordance with the Plan and the directions of the Administrative Committee shall fully discharge the Company, the Board and the Administrative Committee from all further obligations with respect to that payment.
- 7.8 Payments to Minor or Incapacitated Beneficiaries. In making distributions from the Plan to or for the benefit of any minor or incapacitated Beneficiary, the Administrative Committee, in its sole and absolute discretion, may direct the Company to make such distribution to a legal or natural guardian of such Beneficiary, or to any adult with whom the minor or incompetent temporarily or permanently resides. The receipt by such guardian or other adult shall be a complete discharge of liability to the Company, the Board, and the Administrative Committee. Neither the Board, the Administrative Committee, nor the Company shall have any responsibility to see to the proper application of any payments so made.
- 7.9 <u>Effect of Divorce</u>. If a Participant and his or her Designated Beneficiary are or become married and thereafter their marriage is dissolved by entry of a decree of dissolution or other court order having the effect of dissolving the marriage, then any such pre-divorce Beneficiary designation shall be deemed automatically revoked as to such Beneficiary spouse as of the date of such dissolution unless the death benefit rights of such former spouse are subsequently reaffirmed by a qualified domestic relations order or the Participant's subsequent written designation.

ARTICLE VIII ADMINISTRATION OF THE PLAN

- 8.1 <u>Plan Sponsor and Administrator</u>. The Company is the "Plan Sponsor," and its address is: Nordstrom, Inc., 1700 Seventh Avenue, Seattle, Washington 98101-4407. The Administrative Committee is the "Plan Administrator."
- 8.2 <u>Powers and Authority of the Company.</u> The Company, acting through the Compensation Committee of its Board of Directors, has the following absolute powers and authority under the Plan:
 - (a) To amend or terminate the Plan, at any time and for any reason;
 - (b) To determine the amount, timing, vesting, and other conditions applicable to Plan contributions and benefits;

- (c) To set aside funds to assist the Company to meet its obligations under this Plan, provided that the funds are set aside in a manner that does not result in immediate taxation to Participants;
 - (d) To establish investment policy guidelines applicable to funds (if any) set aside under (c);
- (e) To establish one or more grantor trusts (as defined in Code Section 671 et seq.) to facilitate the payment of benefits under the Plan;
 - (f) To take any such other actions as it deems advisable to carry out the purposes of the Plan; and
- (g) To delegate its authority to any officer, employee, committee or agent of the Company, as it deems advisable for the effective administration of the Plan.

8.3 Administrative Committee.

- (a) Role of Administrative Committee. The Company has appointed the Administrative Committee to act as Plan Administrator. All actions taken by the Administrative Committee, or by its delegate, as Plan Administrator will be conclusive and binding on all persons having any interest under the Plan, subject only to the provisions of Article IX. All findings, decisions and determinations of any kind made by the Administrative Committee or its delegate shall not be disturbed unless the Administrative Committee has acted in an arbitrary and capricious manner.
 - (b) <u>Powers and Authority</u>. The Administrative Committee has the following powers and authority under the Plan:
 - (1) In the exercise of its sole, absolute, and exclusive discretion, to construe and interpret the terms and provisions of the Plan, to remedy and resolve ambiguities, to grant or deny any and all non-routine claims for benefits and to determine all issues relating to eligibility for benefits;
 - (2) To authorize withdrawals due to unforeseeable financial emergency;
 - (3) To amend the Plan for legal, technical, administrative, or compliance purposes, as recommended by legal counsel;
 - (4) To retain and pay service providers whose services the Administrative Committee deems necessary to effective administration of the Plan;
 - (5) To implement, in the manner it deems appropriate, the investment policy guidelines established by the Compensation Committee; and

- (6) To delegate its authority to any officer, employee, committee or agent of the Company, as it deems advisable for the effective administration of the Plan, any such delegation to carry with it the full discretion and authority vested in the Administrative Committee.
- (7) To adopt such administrative policies, procedures and protocols as it deems advisable for the effective administration of the Plan.
- (c) <u>Exercise of Authority</u>. All resolutions or other actions taken by the Administrative Committee shall be either: by vote of a majority of those present at a meeting at which a majority of the members are present; or in writing by a majority of all the members at the time in office if they act without a meeting.
- 8.4 <u>Powers and Authority of Leadership Benefits</u>. Leadership Benefits, or any other person(s), committee(s), department or group appointed by the Company's Chief Human Resources Officer, has the following powers and authority under the Plan:
- (a) To carry out day-to-day administration of the Plan, including notifying Eligible Employees of the provisions of the Plan, approving and processing Deferral Agreements, providing Participants with statements of Account, approving and processing changes in the time and/or form of distributions, and forwarding non-routine distribution requests to the Administrative Committee;
- (b) To prepare forms necessary for the administration of the Plan, including Deferral Agreements, beneficiary designation forms, investment designation forms, and any other form or document deemed necessary to the effective administration of the Plan;
- (c) To approve and adopt communications to be furnished to Eligible Employees explaining the material provisions, terms, and conditions of the Plan;
- (d) To process routine distributions and to process non-routine distributions that have been approved by the Administrative Committee;
- (e) To negotiate and document agreements with Plan service providers, subject to final approval by the Administrative Committee:
 - (f) To implement any policies or procedures approved by the Company or the Administrative Committee;
 - (g) To recommend amendments to the Plan for adoption by the Company or the Administrative Committee;
 - (h) To work with Plan service providers to ensure the effective administration of the Plan; and

- (i) To perform any and all tasks, duties, and responsibilities delegated by the Company or the Administrative Committee.
- 8.5 <u>Reliance on Opinions</u>. Each person or entity authorized to act under this Plan shall be entitled to rely on all certificates and reports made by any duly appointed accountants, and on all opinions given by any duly appointed legal counsel, including legal counsel for the Company.
- 8.6 <u>Information</u>. The Company shall supply full and timely information on all matters relating to the compensation of Participants, the date and circumstances of the termination of employment or death of a Participant and such other pertinent information as may be necessary for the effective administration of the Plan.
- 8.7 <u>Indemnification</u>. The Company shall indemnify and hold harmless each Administrative Committee or Board member, and each Company employee performing services or acting in any capacity with respect to the Plan, from and against any and all expenses and liabilities arising in connection with services performed in regard to this Plan. Expenses against which such individual shall be indemnified hereunder shall include, without limitation, the amount of any settlement or judgment, costs, counsel fees and related charges reasonably incurred in connection with a claim asserted, or a proceeding brought or settlement thereof. The foregoing right of indemnification shall be in addition to any other rights to which any such individual may be entitled as a matter of law or other agreement. However, the right to indemnification does not apply where an expense or liability is incurred due to an individual's fraudulent or intentionally dishonest acts.

ARTICLE IX CLAIMS PROCEDURE

- 9.1 <u>Submission of Claim</u>. Benefits shall be paid in accordance with the provisions of this Plan. The Participant, or any person claiming through the Participant ("Claiming Party"), shall make a written request for benefits under this Plan, mailed or delivered to Leadership Benefits. If the claim cannot be processed as a routine payment of benefits, Leadership Benefits will forward the claim to the Administrative Committee for review.
- 9.2 <u>Denial of Claim</u>. If a claim for payment of benefits is denied in full or in part, the Administrative Committee or its delegate shall provide a written notice to the Claiming Party within ninety days after receipt of the claim setting forth: (a) the specific reasons for denial; (b) any additional material or information necessary to perfect the claim; (c) an explanation of why such material or information is necessary; and (d) an explanation of the steps to be taken for a review of the denial. A claim shall be deemed denied if the Administrative Committee or its delegate does not take any action within the aforesaid ninety day period.
- 9.3 <u>Review of Denied Claim</u>. If the Claiming Party desires Administrative Committee review of a denied claim, the Claiming Party shall notify the Administrative Committee or its delegate in writing within sixty days after receipt of the written notice of denial. As part of such

written request, the Claiming Party may request a review of the Plan document or other pertinent documents, may submit any written issues and comments, and may request an extension of time for such written submission of issues and comments.

9.4 <u>Decision upon Review of Denied Claim</u>. The decision on the review of the denied claim shall be rendered by the Administrative Committee within sixty days after receipt of the request for review. The Administrative Committee may extend this period for up to sixty additional days with advance notice to the Claiming Party, an explanation of why the extension is necessary, and an estimated date of decision. The decision shall be in writing and shall state the specific reasons for the decision, including reference to specific provisions of the Plan on which the decision is based.

ARTICLE X AMENDMENT AND TERMINATION

The Plan may be amended or terminated at any time for any reason. Such amendment or termination may modify or eliminate any benefit hereunder, provided that no such amendment or termination shall in any way reduce the vested portion of the affected Participants' or Beneficiaries' Accounts. To be effective, an amendment must be in writing and must be signed by a person who has amendment authority under the terms of the Plan. Oral amendments or modifications to the Plan, and any written amendments that are not signed by an authorized person, are not valid or binding on the Company or any other person. Upon termination of the Plan, the Board of Directors may elect to accelerate distribution of Participant Accounts, but only if the accelerated distribution would not result in additional tax to the Participant under Code Section 409A.

ARTICLE XI MISCELLANEOUS

- 11.1 <u>No Employment Contract</u>. The terms and conditions of the Plan shall not be deemed to constitute a contract of employment between the Company and an employee. Nothing in this Plan shall be deemed to give an Eligible Employee the right to be retained in the service of the Company, its subsidiaries or affiliates or to interfere with any right of the Company, its subsidiaries or affiliates to discipline or discharge the Eligible Employee at any time.
- 11.2 <u>Employee Cooperation</u>. As a condition to participation in the Plan, an Eligible Employee must cooperate with the Company by furnishing any and all information reasonably requested by any of the Company, its subsidiaries or affiliates, and take such other actions as may be requested to facilitate Plan administration and the payment of benefits hereunder.
- 11.3 <u>Illegality and Invalidity</u>. If any provision of this Plan is found illegal or invalid, said illegality or invalidity shall not affect the remaining parts hereof, but the Plan shall be construed and enforced as if such illegal and invalid provision had not been included herein.

- 11.4 Required Notice. Any notice which shall be or may be given under the Plan or a Deferral Agreement shall be in writing and shall be mailed by United States mail, postage prepaid, or in such other manner as the Company determines is appropriate. If notice is to be given to the Company, such notice shall be addressed to the Company c/o Leadership Benefits Department, at 1700 Seventh Avenue, Suite 1000, Seattle Washington 98101-4407. The appeal from a denied claim must be in writing and sent physically by mail or courier to Leadership Benefits. If notice is to be given to a Participant, such notice shall be addressed to the last known address, either geographic or electronic, in the Company's Human Resources records. Any notice or filing required or permitted to be given to a Participant under this Plan shall be sufficient if in writing and handdelivered, or sent by mail (either physical or electronic), to the last known address of the Participant. Any party may, from time to time, change the address to which notices shall be mailed by giving written notice of such new address.
- 11.5 <u>Interest of Participant's Spouse</u>. The interest in the benefits hereunder of a spouse of a Participant who has predeceased the Participant shall automatically pass to the Participant and shall not be transferable by such spouse in any manner, including but not limited to such spouse's will, nor shall such interest pass under the laws of intestate succession.
- 11.6 <u>Tax Liabilities from Plan</u>. If all or any portion of a Participant's benefit under this Plan generates a tax liability to the Participant, including a liability under Code Section 409A, prior to the time that the Participant is entitled to a distribution from the Plan, the Administrative Committee may, in its discretion, instruct the Company to distribute immediately available funds to the Participant in an amount necessary to satisfy such tax liability.
- 11.7 <u>Benefits Nonexclusive</u>. The benefits provided for a Participant and Participant's Beneficiary under the Plan are in addition to any other benefits available to such Participant under any other plan or program for employees of the Company. The Plan shall supplement and shall not supersede, modify or amend any other such plan or program except as may otherwise be expressly provided.
- 11.8 <u>Discharge of Company Obligation</u>. The payment of benefits under the Plan to a Participant or Beneficiary shall fully and completely discharge the Company, the Board, and the Administrative Committee from all further obligations under this Plan with respect to a Participant, and that Participant's Deferral Agreement shall terminate upon such full payment of benefits.
- 11.9 <u>Costs of Enforcement</u>. If any action at law or in equity is necessary by the Administrative Committee or the Company to enforce the terms of the Plan, the Administrative Committee or the Company shall be entitled to recover reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which that party may be entitled.
- 11.10 <u>Gender and Case</u>. Unless the context clearly indicates otherwise, masculine pronouns shall include the feminine and singular words shall include the plural and vice versa.

- 11.11 <u>Titles and Headings</u>. Titles and headings of the Articles and Sections of the Plan are included for ease of reference only and are not to be used for the purpose of construing any portion or provision of the Plan document.
- 11.12 <u>Applicable Law</u>. To the extent not preempted by federal law, the Plan shall be governed by the laws of the State of Washington.
- 11.13 <u>Counterparts</u>. This instrument and any Deferral Agreement may be executed in one or more counterparts, each of which is legally binding and enforceable.

11.14 Additional Definitions:

- (a) "Board" means the Board of Directors of Nordstrom, Inc.
- (b) "Code" means the Internal Revenue Code of 1986, as amended.
- (c) "Deferred Retirement Date" means a Termination Date that occurs after a Participant's Normal Retirement Date.
- (d) "Early Retirement Date" means the Participant's Termination Date on or after the date the Participant has both attained age 53 and has completed at least ten (l 0) Years of Service with the Company. Notwithstanding the foregoing, the 2007 Restatement of this Plan, and not this Restatement, shall apply to determine Early Retirement for those Participants designated as a 1999 Plan Executive under the Company's Supplemental Executive Retirement Plan ("SERP") and for those Participants who, as of August 19, 2003, had attained at least age fifty and had attained at least 10 Years of Service.
- (e) "Equity Incentive Plan" means the separately stated Nordstrom, Inc. 2010 Equity Incentive Plan, as amended through February 27, 2013 and as it may be thereafter amended from time to time, or any successor to the Equity Incentive Plan that provides for performance-based equity compensation.
- (f) "401(k) Plan" means, with respect to a Participant, any Company-sponsored, tax-qualified individual account retirement plan in which the Participant is eligible and which is subject to the requirements of ERISA, whether or not that plan provides for elective deferrals under Code section 401(k). As of January 1, 2019, the definition of 401(k) Plan includes the following Company-sponsored plans: Nordstrom 401(k) Plan, and Nordstrom Puerto Rico Retirement & Savings Plan..
 - (g) "401(k) Plan Compensation" means "Compensation" as defined under the 401(k) Plan.
- (h) "Normal Retirement Date" means a Participant's 58th birthday; provided, however, that the Normal Retirement Date for a Participant who was designated in 2003 as a Transition Plan Executive under the SERP shall be age 55.

(i)	"Participating	Subsidiaries	and A	ffiliates"	means	those	subsidia	ries an	ıd affili	iates c	f the	Compar	ıy that,	subject to
approval by	the Administra	ative Commi	ittee, l	nave spec	cifically	acted	l to adop	ot this	Plan	throug	h exe	cution (of a Pa	rticipation
Agreement. A	A list of Particip	oating Subsic	liaries	and Emp	loyers a	s of Ja	nuary 1,	2017 a	appears	s as Ex	hibit	A to this	s Amen	dment.

- (j) "Participation Agreement" means the written agreement evidencing the terms and conditions under which a particular Participating Subsidiary or Affiliate participates in this Plan.
 - (k) "Plan Year" means the calendar year.
- (l) "Termination Date" means the termination of a Participant's employment with the Company, and each of its subsidiaries and affiliates, whether or not the subsidiary or affiliate participates in this Plan. A termination of employment is deemed to have occurred for purposes of this Plan on the date when the Participant and the Company reasonably anticipate that the level of bona fide services to be provided by the Participant will be permanently reduced to 49 percent or less of the average level of bona fide services provided in the immediately preceding period of 36 consecutive months. If the Participant is on a paid leave of absence, the Participant is treated as providing services at a level equal to the level of services that the Participant would have been required to perform to earn the amount of compensation paid during the paid leave of absence. If the Participant is on an unpaid leave of absence, the employment relationship is presumed to terminate on the earlier of (A) the date the Participant loses his or her statutory or contractual right to re-employment (but not sooner than six months after the unpaid leave of absence began) or (B) the date that there is no longer a reasonable expectation that the Participant will return to perform services for the Company.
- (m) "Years of Service" means consecutive full years (i.e., 12 months), based on service from the Participant's most recent date of hire.

IN WITNESS WHEREOF, this instrument setting forth the terms and conditions of this amendment and restatement to the NORDSTROM DEFERRED COMPENSATION PLAN is executed this ______ day of _____, 2018, effective for compensation deferred and vested on and after January 1, 2019, except as otherwise provided herein.

NORDSTROM, INC.

By:___

Title: Chief Human Resources Officer

EXHIBIT A

(PARTICIPATING SUBSIDIARIES)

Subsidiary	Effective Date
Nordstrom Puerto Rico LLC	January 1, 2015

AMENDMENT 2021-1 NORDSTROM DEFERRED COMPENSATION PLAN

(2019 Restatement)

The Nordstrom Deferred Compensation Plan (the "Plan") is hereby amended as follows, pursuant to Article X, to change references from "Leadership Benefits" to "Nordstrom Benefits" throughout the document, clarify provisions related to deferral of bonus compensation and base compensation, and to delete the exhibit for participating subsidiaries. Unless otherwise specified below, these amendments will be effective as of January 1, 2022:

1. Effective as of the date of this Amendment 2021-1, the term "Leadership Benefits" is hereby replaced in the following sections with "Nordstrom Benefits:"

Sections 2.3(a), 3.1(a), 4.4(b), 4.6, 6.1(c), 6.4(b), 6.4(c), 6.6(a), 7.1, 8.4, 9.1, 11.4 and the Index of Defined Terms.

- 2. Section 3.1(b)(2) "Bonus Compensation" is replaced in its entirety with the following:
- 3.1(b)(2) <u>Bonus Compensation.</u> A Participant's Bonus Compensation paid under the Company's broad-based bonus plan (the "Company Bonus Plan" and any successor plan), whether paid in cash or stock and whether paid annually, quarterly or on any other basis. A Participant's Bonus Compensation shall not include any other bonuses paid by the Company, including, but not limited to, individual or segment bonuses.
- 3. Section 3.2(a) is replaced in its entirety with the following:
- (a) <u>Base Compensation</u>. All or a portion of the Participant's Base Compensation expressed as a percentage of up to eighty percent (80%) of the Eligible Employee's Base Compensation. The deferral percentage applied to a Participant's Base Compensation each pay period shall be based on a Participant's annualized Base Compensation and shall be determined in accordance with the policies and procedures adopted by the Plan Administrator.
- 4. Effective as of the date of this Amendment 2021-1, Exhibit A (Participating Subsidiaries) is hereby deleted in its entirety.

* * * * *

IN WITNESS WHEREOF, pursuant to proper authority, this Amendment 2021-1 has been executed on behalf of the Company this 29th day of April, 2021.

NORE	OSTROM, INC.	
By:		
Chr	istine Deputy	
Title:	Chief Human Resources Officer	

Certification required by Section 302(a) of the Sarbanes-Oxley Act of 2002

- I, Erik B. Nordstrom, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Nordstrom, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f)) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles:
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 4, 2021

/s/ Erik B. Nordstrom
Erik B. Nordstrom
Chief Executive Officer of Nordstrom, Inc.

Certification required by Section 302(a) of the Sarbanes-Oxley Act of 2002

- I, Anne L. Bramman, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Nordstrom, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles:
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 4, 2021

Isl Anne L. Bramman
Anne L. Bramman
Chief Financial Officer of Nordstrom, Inc.

NORDSTROM, INC.

1617 SIXTH AVENUE

SEATTLE, WASHINGTON 98101

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Nordstrom, Inc. (the "Company") on Form 10-Q for the period ended May 1, 2021, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), we, Erik B. Nordstrom, Chief Executive Officer (Principal Executive Officer), and Anne L. Bramman, Chief Financial Officer (Principal Financial Officer), of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: June 4, 2021

/s/ Erik B. Nordstrom
Erik B. Nordstrom
Chief Executive Officer of Nordstrom, Inc.

<u>/s/ Anne L. Bramman</u> Anne L. Bramman Chief Financial Officer of Nordstrom, Inc.

A signed original of this written statement required by Section 906 has been provided to Nordstrom, Inc. and will be retained by Nordstrom, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.